

Economy and Finance in Smart-Cities

Isam Shahrour, Fateh Belaid, Véronique Flambard, Pierluigi Siano, Laura-Diana Radu, Lidia Stermieri, Chiara Delmastro, Cristina Becchio, Stefano Paolo Corgnati, Adrian Tantau, et al.

► To cite this version:

Isam Shahrour, Fateh Belaid, Véronique Flambard, Pierluigi Siano, Laura-Diana Radu, et al.. Economy and Finance in Smart-Cities. Smart Cities, 4, pp.1022 - 1038, 2021, 10.3390/smartcities3030051 . hal-04581952

HAL Id: hal-04581952 https://univ-catholille.hal.science/hal-04581952v1

Submitted on 23 May 2024 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.





an Open Access Journal by MDPI

Economy and Finance in Smart-Cities

Guest Editors:

Prof. Dr. Isam Shahrour

Civil and Geo-Environmental Laboratory, Lille University, 59650 Villeneuve d'Ascq, France

Prof. Dr. Fateh BELAID

Lille Catholic University, Faculty of Management, Economics & Sciences. UMR 9221-LEM-Lille Économie Management, Lille -59016– France

Prof. Dr. Véronique Flambard

Faculty of Management, Economics and Sciences, Université Catholique de Lille, 59000 Lille, France

Deadline for manuscript submissions: closed (15 February 2022)



Message from the Guest Editors

Dear Colleagues,

Smart city is spreading very quickly around the world. Important researches have already been conducted in the use of digital technology for mart city transformation. However, a successful smart city transformation requires a deep investigation of issues related to the economy and finance of smart cities, such as the economic model, finance mechanisms, economic return of the investment, integration of environmental and social benefits in the economic return and the private and public partnership. In addition, the smart city concept opens large perspectives for innovation in the field of smart finance and smart economy.

The objective of this Special Issue is to publish original papers concerning economics and financing in smart cities with a special interest for feed-back from smart city projects or from related issues such as smart buildings, smart water, smart energy and smart transport. This issue is also interested by innovative concepts and frameworks in the field of Smart Finance and Smart Economy.

Prof. Dr. Isam SHAHROUR Prof. Dr. Fateh BELAID Prof. Dr. Véronique FLAMBARD Guest Editors







an Open Access Journal by MDPI

Editor-in-Chief

Prof. Dr. Pierluigi Siano

Department of Management and Innovation Systems, University of Salerno, 84084 Salerno, Italy

Message from the Editor-in-Chief

Smart Cities provides an advanced forum for the dissemination of information on the science and technology of smart cities. It publishes reviews, regular research papers (articles) and communications in all areas of research concerning smart cities. Our aim is to encourage scientists to publish their experimental and theoretical results in as much detail as possible. There is no restriction on the length of the papers so that the full experimental results can be reproduced. Manuscripts regarding research proposals and research ideas are particularly welcome.

Author Benefits

Open Access: free for readers, with article processing charges (APC) paid by authors or their institutions. **High Visibility:** indexed within Scopus, ESCI (Web of Science), Inspec, AGRIS, and other databases. **Journal Rank:** CiteScore - Q1 (*Urban Studies*)

Contact Us

Smart Cities Editorial Office MDPI, St. Alban-Anlage 66 4052 Basel, Switzerland Tel: +41 61 683 77 34 www.mdpi.com mdpi.com/journal/smartcities cities@mdpi.com





Perspective Disruptive Technologies in Smart Cities: A Survey on Current Trends and Challenges

Laura-Diana Radu

Department of Accounting, Business Information Systems and Statistics, Faculty of Economics and Business Administration, "Alexandru Ioan Cuza" University, 700506 Iași, Romania; glaura@uaic.ro

Received: 16 August 2020; Accepted: 9 September 2020; Published: 13 September 2020



Abstract: This paper aims to explore the most important disruptive technologies in the development of the smart city. Every smart city is a dynamic and complex system that attracts an increasing number of people in search of the benefits of urbanisation. According to the United Nations, 68% of the world population will be living in cities by 2050. This creates challenges related to limited resources and infrastructure (energy, water, transportation system, etc.). To solve these problems, new and emerging technologies are created. Internet of Things, big data, blockchain, artificial intelligence, data analytics, and machine and cognitive learning are just a few examples. They generate changes in key sectors such as health, energy, transportation, education, public safety, etc. Based on a comprehensive literature review, we identified the main disruptive technologies in smart cities. Applications that integrate these technologies help cities to be smarter and offer better living conditions and easier access to products and services for residents. Disruptive technologies are generally considered key drivers in smart city progress. This paper presents these disruptive technologies, their applications in smart cities, the most important challenges and critics.

Keywords: smart city; disruptive technologies; Internet of Things; artificial intelligence; blockchain

1. Introduction

The continuous growth of the population in urban areas has led to new challenges. The availability of resources (physical infrastructure), the quality of knowledge communication, and human and social capital (social infrastructure) are considered some of the most important [1]. Access to education, healthcare, transport, justice, social services, utilities, etc., must be available for a population that has grown from 33.6% in 1960 to 55.7% in 2019 [2] and which—according to the estimation of the World Health Organisation—is expected to increase by 1.63% per year between 2020 and 2025, and by 1.44% per year between 2025 and 2030 [3]. To ensure the sustainability of future megacities, an important role is played by how data are gathered, stored, and accessed. Both city authorities and information and communication technology (ICT) companies are responsible for finding the best methods and technologies. Currently, data management systems that integrate the Internet of Things (IoT), big data, and cloud computing are improving certain operations, such as traffic control, sustainable resource management, quality of life, and infrastructure [4]. The smart cities concept includes a wide variety of applications from different fields. Lim et al. [5] identify four groups of technological factors in smart cities, namely, the "4Cs": Connection between things and people, Collection of data for context awareness, Computation in the cloud, and Communication by wireless means. These factors require a variety of technologies characterised by an accelerated and continuous dynamic. The massive amount of information, the speed of updating, and the increasing number of users are some of the elements that determine the emergence of disruptive innovations, mainly in the ICT area. They are rapidly assimilated by consumers. The disruptive technologies are defined as "innovative solutions that require fewer resources and can grow exponentially, very often, shaking up the economy and

structure of the related businesses" [6]. They also bring some challenges. The controlled transition of the labour market due to automation, winning the war on talent between metropolitan areas, social cohesion, inclusiveness, solidarity, secure digital environment, privacy, and resilience are some of the most significant [7]. Ullah et al. [8] identify Big9 disruptive technologies clustered in three domains: data mining—big data and artificial intelligence (AI); networking—cloud, software as a service, IoT and drones; and data collection technologies—3D scanning, weareble tech, virtual and augmented realities. Investment in these technologies has the potential to accelerate the evolution of the smart city [9]. This is conditioned by the coordination between the three pillars of a city: the industry (production), citizen (consumption), and government (circulation) [10].

The technological evolution depends on the economic context—smart economy in smart cities. This concept is used in various contexts regarding smart urban design and development, economic development, strategic planning, advertisement of the cities, and branding [11]. Without a universally accepted definition, all proposals mix the following terms in various variants: 'innovation and creativity', 'technology', 'renewable resources', 'environmental protection', 'social responsibility', 'new jobs and new business', 'labour market flexibility', 'competitiveness', 'entrepreneurship', and 'integration in local and international markets'. All these concepts, integrated into the definition of the smart economy, prove both its complex role in the development of the smart city and the influence of the evolution of all other fields in the development of the smart economy. They are both drivers and consequences. First, the adoption of the disruptive tehnologies requires considerable financial and human investment. This can be achieved by attracting talent, by infusing public and private investment, and by citizens' involvement as consumers and suppliers. Citizens are important sources of information for co-creating Internet-based applications in all sectors of the economy [12] and are creators and beneficiaries of technological evolution and economic progress.

This research further explores the opportunities and challenges of four disruptive technologies: IoT, big data, AI, and blockchain for the development of smart cities. Previous research presents the role of disruptive technologies in specific fields of the cities in general or in the particular case of smart cities, such as real estate [8], transportation and mobility [13–15], health [16], smart homes [17], etc. Other research address only one disruptive technology such as AI [18,19], big data [20], blockchain [21,22], or IoT [23]. This paper offers an overview of all these disruptive technologies together with their contribution to the development of smart cities. It is also exemplified how they are used for each of the six specific components of the smart city (smart economy, smart mobility, smart governance, smart people, smart living, and smart environment). The interdependence between these technologies and the critical aspects that must be had in their implementation and use in real life is highlighted.

This paper has the following structure: Section 2 presents a background of previous research, Section 3 introduces the materials and methods employed to analyse the disruptive technologies in smart cities, then Section 4 displays the results of our search, while Section 5 discusses their relevance. Finally, Section 6 concludes the paper and suggests potential future research.

2. Background

There are various definitions of smart cities, all related in various ways to the evolution of the digital society. However, smart cities are not just about automating routine functions, optimising the use of space, buildings, or traffic management systems. They should provide support for monitoring activity in cities, understanding, and the analysis of economic and social life to increase the efficiency and quality of life and to ensure equity. The smart city also means providing premises that could generate smarter and more accessible ways to interact, especially between citizens and local government or between suppliers of products and services and consumers. According to Batty et al. [24], in smart cities, ICT is "merged with traditional infrastructures, coordinated and integrated using new digital technologies". Everything becomes part of a system of ultra-connected subsystems that allows real-time access to information, products, and services. The level of integration of these subsystems

is an indicator of the intelligence of cities and allows for the optimisation of the flow of information between subsystems and citizens.

According to Muvuna et al. [25], smart cities have to be redesigned to become more sustainable, integrated, and collaborative. Many projects address specific issues facing the city administration and citizens. They have been developed to solve specific problems related to parking, air pollution, traffic congestion, waste management, tax management, health information management, etc. Chourabi et al. [26] identified eight groups of critical factors in creating a framework for smart cities' development: management and organization, technology, governance, policy context, people and communities, economy, built infrastructure, and natural environment. They are currently the basis for setting priorities on local government agendas. Neirotti et al. [27] divided concerns about smart cities into two major areas (Figure 1).



Figure 1. 'Hard' and 'soft' domains of smart cities (author' elaboration after [27]).

Tangible assets or 'hard' domain refer to initiatives on buildings (residential and office), natural resources management, energy management, energy grids, waste management, environment, transport, and logistics, and public security. Intangible assets or 'soft' domain include investment and innovation in education and culture, welfare and social inclusion, public administration and (e-) government, and the economy. Health and public safety are special categories. They are at the interaction between hard and soft domains because it involves both the creation of devices and software, as well as campaigns and practices created to generate social values [27]. All these categories of assets transform the city into a living one that feels and acts due to the connection of billions of sensors, wireless technologies, and software solutions capable of managing a huge volume of various data (big data). This is possible due to the evolution of Big9 disruptive technologies [8]. They underlie the development of applications for entertainment, building management, traffic management, water quality, public safety and services, smart lighting, forest fire detection, wine quality, autonomous vehicles, smart home, and office, etc. The next sections present more information about these technologies and specific areas of use.

According to Batty et al. [24], smart cities are instruments that support the improvement of competitiveness and contribute to a significant increase in the quality of life. Open data, infrastructures, mobile applications, public participation tools, IoT platforms, etc., are designed to ensure that citizens have access to the resources needed. According to Deloitte Consulting Group [7]:

"A city is smart when investments in (i) human and social capital, (ii) traditional infrastructure and (iii) disruptive technologies fuel sustainable economic growth and a high quality of life, with a wise management of natural resources, through participatory governance."

According to a report published by Statista [28], the worldwide spending on the IoT was around USD 646 billion in 2018 and USD 1100 billion in 2023. The most significant increases are estimated in the fields of discrete manufacturing, utilities, transport, and logistics. AI software revenue is also forecast to grow from USD 10.1 billion in 2018 to USD 126.0 billion by 2025 according to a report published by Tractica [29] across 28 industry sectors. Smart projects require specialised staff and create new jobs and opportunities for start-ups. Blockchain has the potential to improve transaction speeds, enhancing security, and reducing costs in the banking sector [30], which would enable real-time accounting [31], and would improve the trustworthiness of financial statements [32]. According to a report published by Statista [33], the spending on blockchain solutions was USD 1.5 billion in 2018 from USD 0.95 in 2017 and is expected to grow to USD 15.9 billion by 2023. They increase the dynamics of transactions and, implicitly, of economic activity as a whole. AI and IoT change the professional and social life of citizens. They have access to a wide volume of information and services and have more free time, as a significant part of their tasks are taken over by these technologies. Regarding the evolution of big data by 2022, the worldwide big data analytics revenue will be USD 274.3 billion according to a report published by International Data Corporation [34]. The evolution of these disruptive technologies will have a favorable economic impact. Figure 2 shows this relationship.



Figure 2. Relationship between disruptive technologies and economic progress.

Huang et al. [35] use the concept of a feeling economy to describe the changes generated by AI. According to them, "AI performs many of the analytical and thinking tasks, and human workers gravitate more toward interpersonal and empathetic tasks". The number of jobs that involve feeling tasks, such as communicating with people inside and outside the organisation, establishing and maintaining interpersonal relationships, and selling to or influencing others, resolving conflicts and negotiating with others, working directly with the public, training teams [35], etc., exceeds those requiring thinking or mechanical tasks. Unlike the feeling economy, the current stage—the thinking economy—requires a significant amount of time allocated for processing, analysis, data interpretation, decision making, work planning and prioritisation, monitoring and controlling resources, updating and using relevant knowledge, and interpreting the meaning of information. The evolution is allowed by the innovations in the field of IoT, big data, and AI technologies that automate the previous tasks.

3. Material and Methods

We have performed a rapid review of available literature on disruptive technologies in smart cities to answer the following research questions:

- (1) What are the disruptive technologies that significantly influence the evolution of smart cities?
- (2) How can these contribute to the development of smarter cities?

According to Cartaxo et al. [36], rapid review offers "reliable content", fosters "the learning of new concepts", is "problem-oriented" and provides a "flexible knowledge transfer medium". We searched in the following electronic libraries: IEEE Xplore, Web of Science, and Scopus. The combination of

phrases was ("disruptive technolog*" AND "smart cit*"). We used the truncation technique for the words that can have various endings to include all their forms into the search. The truncation symbol is an asterisk in this case. The publication date or subject area were not restricted to embrace all relevant publications. The results are presented in Table 1.

Database	Query	Number of Papers
IEEE Xplore	"All Metadata":"disruptive technolog*" AND "smart cit*" OR "Abstract":"disruptive technolog*" AND "smart cit*" OR "Document Title":"disruptive technolog*" AND "smart cit*"	21
Web of Science	("disruptive technolog*" AND "smart cit*") Timespan: All years. Indexes: SCI-EXPANDED, SSCI, A&HCI, CPCI-S, CPCI-SSH, BKCI-S, BKCI-SSH, ESCI, CCR-EXPANDED, IC.	26
Scopus	TITLE-ABS-KEY ("disruptive technolog*" AND "smart cit*") AND (LIMIT-TO (DOCTYPE, "cp") OR LIMIT-TO (DOCTYPE, "ar") OR LIMIT-TO (DOCTYPE, "re") OR LIMIT-TO (DOCTYPE, "ch"))	42

	Table 1.	Databases.	, key	r terms ι	ised	and	results.
--	----------	------------	-------	-----------	------	-----	----------

We filtered publications with the Mendeley Desktop application. The inclusion criteria were being original research or review articles, written in English and covered certain search words. First, duplicates, conference reviews, and announcements of conferences or other events were removed. After this selection, the resulting group comprises 59 papers. Second, we exclude irrelevant papers for research aim. This exclusion reduced the results to 51 articles relevant and reasonable for our research. After the screening process, the full text of studies fulfilling the inclusion criteria was downloaded and the content of these texts was examined to establish the extent to which they are appropriate for this research. From these papers, we identified the most important disruptive technologies that support the evolution of smart cities. The word cloud presented in the next section highlights these technologies, based on the analysis of the title and abstract of the selected articles. To faithfully present their contribution to the development of smarter cities, we completed the research with other references, specific to each disruptive technology. They are described in the next sections together with the fields of application and their positive and negative impact on the city's development. The qualitative evaluation of these publications aimed to form an image on the contribution of the disruptive technologies that significantly contribute to the development of smart cities, as well as the key challenges that arise in their use.

4. Results

The first step in the analysis of the selected articles was to identify the most important disruptive technologies. Based on the fact that Ullah et al.'s. [8] classification corroborated with the results obtained from the analysis of the selected articles, we considered the following four disruptive technologies with major implications for the development of smart cities: IoT, AI, big data, and blockchain. To identify these technologies, we generated the word cloud presented in Figure 3 based on the titles and abstracts of the selected articles. The cloud highlights both the technologies considered disruptive for smart cities and their main areas of application such as 'vehicle, 'transport', 'life', 'vehicle', 'economic', 'mobility', 'sustainable', etc. However, their potential is much greater, as shown by the results presented in the following sections. The cloud was generated from the 100 most used words and the common ones were removed. The cloud shows also the main challenges and concerns in the development and implementation of these technologies: architecture, environment integration, citizen privacy, infrastructure, collaboration, and sustainability.

Next sections present, with examples, each disruptive technology, IoT, AI, big data and blockchain, and their specific applications in smart cities.



Figure 3. Disruptive technologies in smart cities—a word cloud generated by the Wordle App.

4.1. Internet of Things

Smart city infrastructure is an extension of the digital city and integrates a wide range of technologies developed around IoT. According to Hashem et al. [20], this "provides a platform for sensors and actuator devices to communicate seamlessly within the smart city environment and enables an increasingly convenient information sharing across platforms".

IoT creates new opportunities for the use of electronic devices by interconnecting them. The devices perceive and collect environmental information and transmit this information to other devices. Later it can be accessed by interested users. According to a study published by Statista Research Department [37], the number of connected devices is currently 30.73 billion and is estimated to reach 75.44 billion by 2025. Furthermore, the new technologies 5G will become an enabler for IoT by allowing a wide number of simultaneous connections and ubiquity of network in smart cities characterized by densely populated areas with high mobility [38]. IoT has a wide variety of applications, including retail and logistics, smart homes, smart grid, smart farming, smart finance, and insurance, transportation, smart manufacturing, smart supply chain, healthcare, environmental monitoring [39], etc. For example, Baloyi and Telukdarie [40] analysed the benefits of IoT for Disaster Risk Reduction. Sensors monitor activity in the natural environment (water level, wind speed, volcanic gas fluctuations, etc.) and transmit information to an integrated monitoring and control system that informs the vulnerable population and automatically initiates actions to reduce the possible damage, such as opening dams, activating fire suppression systems, etc. The data are also used for predictions and preventions. Another example is the Collaborative Road Mobility System (CRMS), developed for L'Aquila city, Italy, that allows vehicles and transport infrastructure to interconnect, share information and use it to coordinate their actions [41]. The system also integrates a mobile application for traffic coordination that assists drivers in an eco-friendly manner. Other applications of IoT are intelligent waste collection, smart parking [42], air quality monitoring, health monitoring, including mental health, traffic control [20], etc. Table 2 presents a selection of IoT applications in smart city components.

Smart Economy	Smart Mobility	Smart Governance	Smart People	Smart Living	Smart Environment
Industry monitoring [10]	Traffic management [11,41]	E-government [43]	Education [44]	Entertainment [23]	Air pollution [45]
Infrastructure monitoring		E-procurement [46]	Healthc	are [16]	Noise monitoring [23]
Efficient transportation planning [41,47]		Utility management [23]	Entrepreneurial opportunities [48]	Entrepreneurial opportunities Emergency [48] services [49]	
Food and agriculture [51]	Smart parking [52]	Defense [18]			Building management [53]
Radiation and electromagnetic level [54]	Quality of shipment conditions	Public safety [18,23]		Surveillance [23]	Radiation and electromagnetic level [54]
Smart li	ighting [55]	Public services [43]			
]	Efficient energy cor	sumption [4,53]		
Wine quality	Vehicle auto-diagnosis [56]	Cultural facilities [4]	Smart home an	d office [17,23]	Forest fire detection [40]

Table 2. Applications of Internet of Things (IoT) in smart cities.

The development of hyperconnected, ubiquitous networks of devices involves taking into account all their components and characteristics. Intel [43] identified six dimensions of policy principles for the development and use of IoT technologies: connectivity and interoperability, privacy and security, intelligent analytics and big data, open standards, data, and device discoverability, and public and private partnership. However, IoT generates a large volume of data, many of them partially structured or unstructured. As a result, there is a need for more advanced technologies to store, process, and analyse these data, to extract and synthesise relevant information and to make decisions and initiate actions based on the results. Big data and AI are these technologies.

4.2. Big Data and Artificial Intelligence

The evolution of AI is closely linked to the development of big data technologies in the opinion of researchers [57–59] since they complement each other. AI becomes better as the volume of data acquired increases, and big data is useful only together with software to analyse it. These technologies offer the possibility to collect and make useful the information provided by devices connected by the Internet. Since we agree with this approach, we present them together in this section of the paper. Sensor networks generate a huge volume of data. Big data technologies allow for their management. According to Thakuriah et al. [60], big data "refers to structured and unstructured data generated naturally as a part of transactional, operational, planning and social activities, or the linkage of such data to purposefully designed data". Some authors consider big data a strategic economic resource similar in importance to gold and oil [45].

The sources of big data are directed data generated by traditional forms of surveillance and, in this case, the technology is focused on a person or place by a human operator, automated data generated by devices or systems, and volunteered data generated by users [61]. After data collection or receipt, big data analytics and algorithms are used to process and analyse the data to identify and extract useful information for different users, people, or machines [40]. As a result, big data technologies are indispensable for IoT operation in smart cities. They are not, however, without problems and challenges. According Colding et al. [62], big data analytics have "a huge potential to enhance smart city services; however, the combination of the IoT and big data is still an immature, unexplored research avenue, with new challenges for achieving the goal of the smart city". They facilitate, for example, the involvement of citizens who can become providers of information. According to Neirotti et al. [27], citizens' knowledge is a rich source of up-to-date information and could improve the quality of urban

space analysis, compared to traditional forms of spatial planning. This option can be considered an extension of democracy and increases the chances of accepting local development plans.

Big data analytics is possible due to the evolution of AI. In smart cities, their use can provide a variety of economic and social benefits, from traffic decongestion and parking, optimizing energy consumption, healthcare and security, to changing the way citizens live every day, interact and work [18]. The accuracy of the results is conditioned both by the performance of the applications used to interpret the data and by the volume of available data.

The beginning of AI in the early 1950s was followed by periods of growth, stagnation, and decline. Recent studies show a close relationship between big data technologies and AI evolution [57,59]. Figure 4 presents an analysis from Google Trends [63] on the popularity of big data, AI, machine learning, and IoT between 2004 and 2020. Between 2004 to 2008, a decline in AI is observed, followed by a period of moderate evolution until 2011 and another decline between 2011 and 2014. After 2014, AI's popularity rises again. These changes overlap with a significant increase in big data popularity.



Figure 4. Popularity of the concepts of big data, artificial intelligence (AI), IoT and machine learning based on their searches between 2004 and 2020 [63].

AI has a variety of subfields, such as robotics, NLP (content subtraction, question answering, classification, machine translation, and text generation), expert systems, speech recognition (speech to text and text to speech), vision (image recognition, machine vision), etc. Big data technologies allow or the collection of a very large volume of data from all networks. A wide variety of devices and sensors are integrated into the infrastructure. The data obtained are processed using AI-based software by applying advanced analyses. The results are used, finally, to improve the citizens 'quality of life. They will enable better results to be provided for long-term strategic planning, better public services, and better use of resources [64]. Machine learning, for example, needs a wide volume of data to ensure correct results. Big data can provide access to this data. Together with IoT and AI, big data have the potential to make smart cities more efficient and responsive [65] provided they are integrated into a physical infrastructure which, by itself, both through the production and use process, optimises the consumption of resources and reduces the negative impact on the environment. Some examples of AI and big data-based applications are presented below. In Glasgow, a network of sensors uses AI and big data to connect streetlight and traffic lights to support monitor traffic flows and increase connectivity to reduce travel time, including making decisions to plan the commutes [66]. The Seattle Police Department launched a platform for police officers' activities management to track actions such as use-of-force incidents, the number of arrests, self-initiated trips, response to calls, the number of stops, and civilian complaints [67]. London Metropolitan Police use facial recognition technology to identify suspects [68]. The Las Vegas Health Department uses machine learning to collect and examine tweets to select restaurants where inspections are required to prevent foodborne illness [69]. Big data and IoT help to improve the sustainability of smart cities through visualisation techniques. They provide tools for collecting, processing, and presenting the necessary evidence to make strategic and tactical decisions taking into account the expertise of users and facilitating their understanding [70].

Applications based on AI and big data technologies are used to improve knowledge, teaching and learning tools, as well as quickly updating the curriculum by accessing a wide variety of scientific information according to economic and social requirements; to identify patterns of traffic to anticipate future traffic and control traffic lights [5,71]; community energy management, building energy management, and electric vehicles [5]; waste management (trash collection locations and times scheduled) [72]; healthcare to predict contagious diseases and epidemics and ubiquitous availability of citizen's health data [16]; understanding, monitoring, regulating and planning for better city management [61]; real-time video image processing to identify lawbreakers [73] or video-based real-time intrusion detection system [74] to increase the safety of citizen. AI and machine learning are very important to precisely monitor and estimate the real-time traffic flow data in urban environments [19]. In healthcare, AI and big data analytics help with the evaluation of the general health conditions of patients, hospital disinfection, creation, understanding and classification of clinical documentation, cancer prognosis and prediction, rapid intervention in emergencies, surgeries, assistance for seniors, treatment, perform tasks for administrative purposes, etc. Table 3 presents a selection of big data and AI applications in smart city components.

Smart Economy	Smart Mobility	Smart Governance	Smart People	Smart Living	Smart Environment
Support business decisions [58]	Traffic management [14,19]	E-government [9,61]	Education [75]	Entertainment [76]	Water quality [77,78]
Smart lighting [5,79]		Planning [70]	Healthcar	re [16,80]	Disaster prevention [70]
Automating the data management process [81]	Transport congestion [9]	E-procurement [46]	Entrepreneurial opportunities [48]	Crowd monitoring [73,74]	Air quality prediction [82]
Complex statistical analyses [81]	Smart parking [83]	Public safety [73,74]	Social con	nfort [57]	Building management [84]
Efficient energy [5,84,85]					
Perform repetitive tasks [9]	Vehicle auto-diagnosis [9]		Smart ho	ome [86]	Waste management [50,72]

Table 3. Applications of AI and big data in smart cities.

According to Guevara et al. [38], big data and IoT due to 5G technologies will gather more useful information for citizens. The automotive industry is an example of an industry with major benefits brought by the evolution of AI and IoT. All big companies such as Audi, BMW, Daimler, Mercedes-Bosch alliance, Delphi, Didi Chuxing, Ford, General Motors, Honda, Huawei, Hyundai, Jaguar Land Rover, Lyft, Microsoft, nuTonomy, PSA, Renault-Nissan alliance, Samsung, Tesla, Toyota, Uber, Volkswagen Group, Volvo, Waymo, ZF and Zoox work on the development of connected and autonomous vehicles [56]. They will have numerous benefits for users from creating more free time to increasing traffic safety and accident prevention and will create safety smart cities.

4.3. Blockchain

Along with big data and AI, blockchain has developed as a technology that contributes to increasing the intelligence of cities. This is described as a distributed, replicated, and secure digital register that allows contracting parties to see a system of records that are immutable [57]. Blockchain has evolved rapidly. Now it has reached version 4.0, which aims to integrate this technology into the industry to meet all the business needs of users, even if they use different platforms. It encompasses all applications of blockchain technology that is not limited to cryptocurrencies, such as electronic voting, healthcare records management, identity management, decentralised notary, etc. [87]. Given the diversity of financial and contractual transactions that occur in the case of urban governance, as well as the large number of parties involved, the use of blockchain technologies can bring considerable improvements. Furthermore, blockchain technology can enhance trust in data since technology is encrypted and secure [57]. Smart contracts can be used by smart city administrations to overcome bureaucratic difficulties and political issues. In this way, they reduce the amount of time required in developing projects. According to Marsal-Llacuna [88], "blockchains' virtual and physical networking symbiosis through smart contracts empowers citizens for a bottom-up delivery of codes, owned and implemented by them and not by a central authority. In 2018, 60% of the market value of blockchain belonged to the financial sector, but it spread rapidly in other industries, such as health and agriculture [33]. For example, Hîrțan et al. [14] designed a reputation system based on blockchain technology for intelligent transport systems. Users contribute with data to the system and receive real traffic information. The system evaluates the behavior of enrolled users before integrating the received data based on data received from every vehicle. It was tested on a traffic simulator that can simulate thousands of cars in real city environments (for Rome, Beijing, and San Francisco, in this case).

A PWC report [89] on the application of blockchain technology in smart cities estimates that, together with IoT, AI and machine learning, will lead to a significant improvement in urban governance. According to the same report, cities will be characterised by prosperity, stronger and interconnected communities, healthier people, positive growth, and increased social mobility at an unprecedented level [89]. Blockchain can be used for the decentralisation of voting platforms for local elections and legislative issues; providing transaction security in decentralised smart grid energy trading and real-time and flexible pricing [22,90]; in intelligent transport systems to create better usage of the legacy infrastructure and resources [15], to provide a secure communication platform for smart cities; to control and configure devices in IoT [91]; security and energy efficiency in smart homes [17], storing the medical data in an immutable and secure manner and flexible access of the patients and doctors to their medical records; in supply chain management for data sharing tracking the detailed product information as a counterfeit product detection system; and providing on-demand, personalised and integrated services [22]. Table 4 presents a selection of blockchain applications in smart city components.

Table	4. Apr	olications	of b	lockc	hain	in	smart	cities.
-------	--------	------------	------	-------	------	----	-------	---------

Smart Economy	Smart Mobility	Smart Governance	Smart People	Smart Living	Smart Environment
Access control system [87]	Vehicles communication management [92]	E-procurement/ Smart contracts	Personal data m	anagement [92]	Air quality monitoring [93]
Shipment	tracking [92]	[46,89]	Healthca	re [22,92]	
Transportation	planning [14,47]				
Agriculture [94] Transaction	Traffic management [14]	E-voting [89]	Smart hor	me [17,87]	
history		Public safety			
management [92]		[91]			
Smart grid energy [22,90]					

Blockchain technologies are just beginning but offer many opportunities for the development of smart cities due to their characteristics, such as decentralisation, audibility, immutability, and transparency. They are complementary to big data and AI technologies that alone cannot guarantee data quality due to threats such as hacking and human mistakes [88]. By integrating them with blockchain, users' trust in data will increase.

5. Discussion

This study conducts a review of disruptive technologies in smart cities. The research investigates the available literature under the following constraints: English language, three database sources (IEEE Xplore, Web of Science, and Scopus), research and literature review papers only, focus on disruptive technologies and smart cities with limited variants on search keywords. The results facilitated the discussion on current status and progress on the topic of the paper.

Smart cities, through their citizen, integrate IoT, AI, big data, and blockchain into real daily life. The following main points and critical issues are disclosed by our research:

- The disruptive technologies have the potential to generate positive changes in the smart city by promoting sustainable economic and social activities.
- IoT, AI, and big data are useful in automating decision making and problem solving and support the development of smarter cities.
- Blockchain increases the trust in data proving a secure communication platform and better usage of the legacy infrastructure and resources.
- The literature on disruptive technologies in smart cities associates these technologies, especially with AI, big data, IoT, and blockchain.
- AI, big data, IoT, and blockchain are interdependent and complementary and provide support for all other emerging technologies used in smart cities (wearable tech, social robotics, virtual and augmented reality, 3D printing, digital twins, etc.).
- Disruptive technologies in the actual context of smart cities mainly concentrate on mobility and transport, environmental sustainability, health, security, business efficiency, energy efficiency, and education.

The results of the review reveal that disruptive technologies have a profound impact on smart cities, both on physical setup and operational level. They must adapt to the diversity of citizens' problems and needs, to the variety of environmental conditions, and possess a degree of transparency. They also must have the technical performances required by well-trained and more selective users. The level of acceptance of technological innovation will be influenced by the level of trust individuals and companies in these technologies. In this context, it is very important for the final users' implication in their development. They may be allowed to customise applications and devices according to their own needs and lifestyles. Thus, they become not only users but also providers of innovation and increase the chances of assimilating it in everyday life. An important feature of the technologies analysed refers to their contribution to the data security and citizens' safety. These issues are increasingly critical and blockchain is coming with important improvements. The European Commission [95] underlined seven key requirements for the successful use of AI, (1) human agency and oversight, (2) technical robustness and safety, (3) privacy and data governance, (4) transparency, (5) diversity, non-discrimination and fairness, (6) societal and environmental wellbeing, and (7) accountability. They are also applicable to the other disruptive technologies presented in this paper considering the strong relationship between them. The success of disruptive technologies deployment to make cities smarter depends on the citizens' knowledge and interest to keep public and personal values in their adoption [81]. As can be seen from the previous section, there is a wide variety of areas that use applications based on IoT, AI, big data, or blockchain. For each disruptive technology, we present a selection of applications to create the premises for identifying other potential uses.

However, the ubiquity of ICT also has some disadvantages. There are opinions against this trend, argued by the negative effects on the environment, the limitation of human contacts, deculturalisation, etc. According to Allam and Newman [96], smart cities are often described as the "magic bullet" to all urbanization issues.

Each of the three disruptive technologies presented above has its critics. For example, in the case of big data, Mattern [97] estimates that it suffers from "data-fication, the presumption that all meaningful flows and activity can be sensed and measured". It is necessary to maintain a balance between technology and the real needs of the city translated by citizens' needs and environmental protection. Big data allows for the more efficient and dynamic collection and management of various aspects of the city, but a series of safe and sustainable tools, policies, and practices to regulate urban life must complement and validate the innovation in this field.

Confidentiality is another important issue to study when considering the issue of collecting and accessing all data in the smart city. According to Hough [98], privacy has four components: solitude, intimacy, anonymity, and reserve. Regulatory organisations must provide the legal framework to protect the population for each component, but it is quite difficult to predict all possible violations [57]. The technologies used to implement, smart office, smart house and smart health allow access to a wide variety of private information [99]. It is very important not to limit or influence citizens' free will. Moreover, educating citizens to be aware of the boundary between public and confidential information is very important. Otherwise, data can be used to invade their privacy or to manipulate them [100]. Another important aspect is the impact on the environment. According to Beck et al. [21], it is necessary to analyse the environmental implications of powering an expanding network to support the computing power required for blockchain operation. For example, mining bitcoins requires very high and growing energy consumption [101]. Other aspects that should be considered are the distribution of wealth, predominant social arrangements, and geopolitical balance [21]. Moreover, millions of interconnected devices involve the consumption of non-renewable resources for their development. The energy consumption is difficult to estimate globally, and e-waste is difficult to recycle. A study to assess the relationship between the production process in this area and the real usefulness of products is very difficult to achieve but is very useful.

6. Conclusions

This study offers a novel contribution to the literature by mapping out the landscape of disruptive technologies and smart cities. According to the United Nations [102], by 2030, the world is projected to have 43 megacities with more than 10 million inhabitants. In this context, more and more cities have to become smart or smarter. This study aimed to reveal how disruptive technologies, IoT, big data, AI, and blockchain, in this case, contribute to the progress of smart cities. The raising of the technological level of cities is not a goal. It is the driver. The goal of financial and human investment in ICT must be the optimization of the city's functions. The smart city must be smart through the long-term well-being of citizens. This depends on the level of digitisation, but the use and usefulness of ICT in everyday life are overestimated. Over time, this can lead to a loss of privacy, a lack of focus on important goals, an irreversible negative impact on the environment, and deficiencies in the capacity to integrate into society. As a result, the existence of technological limitations is very important. However, it is obvious that, beyond these potential dangers, disruptive technologies bring invaluable benefits to smart cities as presented in this paper. The broad finding of our review findings reveals that: (1) the potential of IoT, big data, AI, and blockchain to provide positive changes in smart cities is huge if they are used with responsibility; (2) these technologies are interdependent and complementary, and will evolve together. The specific findings are: (1) IoT, big data, AI, and blockchain are the most significant disruptive technologies for the evolution of smart cities; (2) they are emerging fields of research and practice in the context of smart cities; (3) the central focus of the literature in this field is on their application in mobility and transport, environmental sustainability, health, security, business efficiency, energy efficiency, and education and less on their risks.

This article is not lacking in limitations. One limitation is the strictly theoretical character and another limitation is associated with the features of the rapid review. To address this, we searched the most representative databases for papers to score a reasonable pool of relevant results.

Successful implementation of these technologies involves collaboration between the public and private sectors, between government organisations and citizens, as well as between universities and companies to ensure a natural evolution, informing citizens about the benefits and risks of using ICT, studying the psychological impact, and establishing responsibilities both ICT developers and users. IoT, big data, artificial intelligence, and blockchain are just some of the disruptive technologies that underlie the evolution of the smart city. Gamification, wearable tech, social robotics, virtual and augmented reality, 3D printing, digital twins, etc., are other disruptive technologies developed based on technologies presented in this paper and will be the subject of future research.

Funding: This research received no external funding.

Conflicts of Interest: The author declares no conflict of interest.

References

- 1. Caragliu, A.; Del Bo, C.; Nijkamp, P. Smart cities in Europe. J. Urban Technol. 2011, 18, 65–82. [CrossRef]
- The World Bank. United Nations Population Division. World Urbanization Prospects: 2018 Revision; The World Bank: Washington, DC, USA, 2020. Available online: https://data.worldbank.org/indicator/SP.URB.TOTL.IN. ZS?end=2019&start=1960&view=chart (accessed on 15 July 2020).
- 3. Knorr, D.; Khoo, C.S.; Augustin, M.A. Food for an urban planet: Challenges and research opportunities. *Front. Nutr.* **2017**, *4*, 73. [CrossRef]
- 4. Kirimtat, A.; Krejcar, O.; Kertesz, A.; Tasgetiren, M.F. Future Trends and Current State of Smart City Concepts: A Survey. *IEEE Access* 2020, *8*, 86448–86467. [CrossRef]
- Lim, C.; Kim, K.; Maglio, P. Smart cities with big data: Reference models, challenges, and considerations. *Cities* 2018, 82, 86–99. [CrossRef]
- Herrera-Quintero, L.F.; Chavarriaga, J.; Banse, K.; Bermudez, D.; Proeller, G. Disruptive Technologies in Intelligent Transportation Systems. In Proceedings of the 2nd Latin American Conference on Intelligent Transportation Systems (ITS LATAM), Bogota, Colombia, 19–20 March 2019; pp. 1–6.
- Van Dijk, A.; Teuben, H. Smart Cities: How Rapid Advances in Technology Are Reshaping Our Economy and Society. Available online: https://www2.deloitte.com/content/dam/Deloitte/tr/Documents/public-sector/ deloitte-nl-ps-smart-cities-report.pdf (accessed on 8 July 2020).
- 8. Ullah, F.; Sepasgozar, S.; Wang, C. A systematic review of smart real estate technology: Drivers of, and barriers to, the use of digital disruptive technologies and online platforms. *Sustainability* **2018**, *10*, 3142. [CrossRef]
- 9. Soomro, K.; Bhutta, M.N.M.; Khan, Z.; Tahir, M.A. Smart city big data analytics: An advanced review. *Wiley Interdiscip. Rev. Data Min. Knowl. Discov.* **2019**, *9*, 1–25. [CrossRef]
- Luque-Vega, L.; Carlos-Mancilla, M.; Payán-Quiñónez, V.; Lopez-Neri, E. Smart Cities Oriented Project Planning and Evaluation Methodology Driven by Citizen Perception—IoT Smart Mobility Case. *Sustainability* 2020, 12, 7088. [CrossRef]
- 11. Bruneckienė, J. The concept of smart economy under the context of creation the economic value in the city. *Public Policy Adm.* **2014**, *13*, 469–482. [CrossRef]
- 12. Komninos, N.; Pallot, M.; Schaffers, H. Special Issue on Smart Cities and the Future Internet in Europe. *J. Knowl. Econ.* **2013**, *4*, 119–134. [CrossRef]
- 13. Torre-Bastida, A.; Del Ser, J.; Laña, I.; Ilardia, M.; Bilbao, M.; Campos-Cordobés, S. Big Data for transportation and mobility: Recent advances, trends and challenges. *IET Intell. Transp. Syst.* **2018**, *12*, 742–755. [CrossRef]
- 14. Hîrţan, L.; Dobre, C.; González-Vélez, H. Blockchain-based reputation for intelligent transportation systems. *Sensors* 2020, 20, 791. [CrossRef] [PubMed]
- Yuan, Y.; Wang, F. Towards blockchain-based intelligent transportation systems. In Proceedings of the International Conference on Intelligent Transportation Systems, Rio de Janeiro, Brazil, 1–4 November 2016; pp. 2663–2668.

- Manogaran, G.; Varatharajan, R.; Lopez, D.; Kumar, P.; Sundarasekar, R.; Thota, C. A new architecture of Internet of Things and big data ecosystem for secured smart healthcare monitoring and alerting system. *Future Gener. Comput. Syst.* 2018, *82*, 375–387. [CrossRef]
- 17. Lin, C.; He, D.; Kumar, N.; Huang, X.; Vijayakumar, P.; Choo, K. Homechain: A blockchain-based secure mutual authentication system for smart homes. *IEEE Internet Things J.* **2019**, *7*, 818–829. [CrossRef]
- Voda, I.A.; Radu, L.D. How can artificial intelligence respond to smart cities challenges? In Smart Cities: Issues and Challenges: Mapping Political, Social and Economic Risks and Threats Serves; Visvizi, A., Lytras, M., Eds.; Elsevier: Amsterdam, The Netherlands, 2019; pp. 199–216.
- 19. Ullah, Z.; Al-Turjman, F.; Mostarda, L.; Gagliardi, R. Applications of Artificial Intelligence and Machine learning in smart cities. *Comput. Commun.* **2020**, *154*, 313–323. [CrossRef]
- 20. Hashem, I.; Chang, V.; Anuar, N.; Adewole, K.; Yaqoob, I.; Gani, A.; Ahmed, E.; Chiroma, H. The role of big data in smart city. *J. Inf. Manag.* 2016, *36*, 748–758. [CrossRef]
- 21. Beck, R.; Avital, M.; Rossi, M.; Thatcher, J. Blockchain Technology in Business and Information Systems Research. *Bus. Inf. Syst. Eng.* **2017**, *59*, 381–384. [CrossRef]
- 22. Bhushan, B.; Khamparia, A.; Sagayam, K.; Sharma, S.; Ahad, M.; Debnath, N. Blockchain for smart cities: A review of architectures, integration trends and future research directions. *Sustain. Cities Soc.* **2020**, *61*, 102360. [CrossRef]
- 23. Talari, S.; Shafie-Khah, M.; Siano, P.; Loia, V.; Tommasetti, A.; Catalão, J. A Review of Smart Cities Based on the Internet of Things Concept. *Energies* 2017, *10*, 421. [CrossRef]
- 24. Batty, M.; Axhausen, K.; Giannotti, F.; Pozdnoukhov, A.; Bazzani, A.; Wachowicz, M.; Ouzounis, G.; Portugali, Y. Smart cities of the future. *Eur. Phys. J. Special Top.* **2012**, 214, 481–518. [CrossRef]
- 25. Muvuna, J.; Boutaleb, T.; Baker, K.; Mickovski, S. A methodology to model integrated smart city system from the information perspective. *Smart Cities* **2014**, *2*, 496–511. [CrossRef]
- Chourabi, H.; Nam, T.; Walker, S.; Gil-Garcia, J.R.; Mellouli, S.; Nahon, K.; Pardo, T.; Scholl, H. Understanding smart cities: An integrative framework. In Proceedings of the 2012 45th International Conference on System Sciences, Maui, HI, USA, 4–7 January 2012; pp. 2289–2297.
- 27. Neirotti, P.; De Marco, A.; Cagliano, A.; Mangano, G.; Scorrano, F. Current trends in Smart City initiatives: Some stylised facts. *Cities* **2014**, *38*, 25–36. [CrossRef]
- Statista. Internet of Things (IoT) Spending Worldwide 2023. Statista Research Department. Statista—The Statistics Portal. 2020. Available online: https://www.statista.com/statistics/668996/worldwide-expendituresfor-the-internet-of-things/ (accessed on 6 August 2020).
- 29. Tractica. Artificial Intelligence Market Forecasts. Tacticat 2019. Available online: https://tractica.omdia.com/ research/artificial-intelligence-market-forecasts (accessed on 6 August 2020).
- 30. Hassani, H.; Huang, X.; Silva, E. Banking with blockchain-ed big data. *J. Manag. Anal.* **2018**, *5*, 256–275. [CrossRef]
- 31. Brennan, N.; Subramaniam, N.; van Staden, C. Corporate governance implications of disruptive technology: An overview. *Br. Account. Rev.* **2019**, *51*, 100860. [CrossRef]
- 32. Byström, H. Blockchains, real-time accounting, and the future of credit risk modeling. *Ledger* **2019**, *4*, 40–47. [CrossRef]
- 33. Liu, S. Blockchain—Statistics & Facts. Statista—The Statistics Portal. Available online: https://www.statista. com/topics/5122/blockchain/ (accessed on 3 September 2020).
- 34. IDC. IDC Forecasts Revenues for Big Data and Business Analytics Solutions Will Reach \$189.1 Billion This Year with Double-Digit Annual Growth Through 2022, International Data Corporation. Available online: https://www.idc.com/getdoc.jsp?containerId=prUS44998419 (accessed on 5 September 2020).
- 35. Huang, M.; Rust, R.; Maksimovic, V. The feeling economy: Managing in the next generation of artificial intelligence (AI). *Calif. Manage Rev.* **2019**, *61*, 43–65. [CrossRef]
- 36. Cartaxo, B.; Pinto, G.; Soares, S. The role of rapid reviews in supporting decision-making in software engineering practice. In Proceedings of the 22nd International Conference on Evaluation and Assessment in Software Engineering, Christchurch, New Zealand, 28–29 June 2018; pp. 24–34.
- Statista. Internet of Things—Number of Connected Devices Worldwide 2015–2025. Available online: https:// www.statista.com/statistics/471264/iot-number-of-connected-devices-worldwide (accessed on 31 July 2020).
- 38. Guevara, L.; Auat Cheein, F. The Role of 5G Technologies: Challenges in Smart Cities and Intelligent Transportation Systems. *Sustainability* **2020**, *12*, 6469. [CrossRef]

- Ervural, B.; Ervural, B. Overview of cyber security in the industry 4.0 era. In *Industry 4.0: Managing the Digital Transformation*; Ustundag, A., Cevikcan, E., Eds.; Springer International Publishing: Cham, Switzerland, 2018; pp. 267–284.
- Baloyi, V.T.; Telukdarie, A. Internet of Things: Opportunity for Disaster Risk Reduction. In Proceedings of the International Conference on Industrial Engineering and Operations Management, Johannesburg, South Africa, 29 October–1 November 2018; pp. 2087–2096.
- 41. Autili, M.; Di Salle, A.; Gallo, F.; Pompilio, C.; Tivoli, M. A choreography-based and collaborative road mobility system for l'Aquila city. *Future Internet* **2019**, *11*, 132. [CrossRef]
- 42. Abbasi, K.; Khan, T.; Haq, I. Hierarchical Modeling of Complex Internet of Things Systems Using Conceptual Modeling Approaches. *IEEE Access* 2019, 7, 102772–102791. [CrossRef]
- 43. Intel. Policy Framework for the Internet of Things (IoT). 2014. Available online: https://www.intel.com/ content/dam/www/public/us/en/documents/corporate-information/policy-iot-framework.pdf (accessed on 18 July 2020).
- 44. Aldowah, H.; Rehman, S.; Ghazal, S.; Umar, I. Internet of Things in higher education: A study on future learning. *J. Phys. Conf. Ser.* 2017, *892*, 012017. [CrossRef]
- 45. Alharthi, A.; Krotov, V.; Bowman, M. Addressing barriers to big data. *Bus. Horiz.* 2017, 60, 285–292. [CrossRef]
- 46. Nicoletti, B. Platforms for Procurement 4.0. In *Procurement 4.0 and the Fourth Industrial Revolution;* Palgrave Macmillan: Cham, Switzerland, 2020.
- Herraiz Faixó, F.; Arroyo-Cañada, F.J.; López-Jurado, M.P.; Lauroba Pérez, A.M. Digital assets Horizon in smart cities: Urban congestion management by IoT, Blockchain/DLT and human reinforcement. In *Modelling* and Simulation in Management Sciences. MS-18 2018. Advances in Intelligent Systems and Computing, 1st ed.; Ferrer-Comalat, J.C., Linares-Mustaros, S., Merigó, J.M., Kacprzyk, J., Eds.; Springer International Publishing: Cham, Switzerland, 2020; Volume 894, pp. 63–82.
- 48. Krotov, V. The Internet of Things and new business opportunities. Bus. Horiz. 2017, 60, 831-841. [CrossRef]
- 49. Gope, P.; Hwang, T. Untraceable Sensor Movement in Distributed IoT Infrastructure. *IEEE Sens. J.* 2015, 15, 5340–5348. [CrossRef]
- Anagnostopoulos, T.; Zaslavsky, A.; Kolomvatsos, K.; Medvedev, A.; Amirian, P.; Morley, J.; Hadjieftymiades, S. Challenges and Opportunities of Waste Management in IoT-Enabled Smart Cities: A Survey. *IEEE Trans. Sustain. Energy* 2017, 2, 275–289. [CrossRef]
- 51. Islam, N.; Marinakis, Y.; Majadillas, M.; Fink, M.; Walsh, S. Here there be dragons, a pre-roadmap construct for IoT service infrastructure. *Technol. Forecast. Soc. Chang.* **2020**, *155*, 119073. [CrossRef]
- Herrera-Quintero, L.; Vega-Alfonso, J.; Bermúdez, D.; Marentes, L.; Banse, K. TS for Smart Parking Systems, towards the creation of smart city services using IoT and cloud approaches. In Proceedings of the Smart City Symposium, Prague, Czech Republic, 23–24 May 2019; pp. 1–7.
- Minoli, D.; Sohraby, K.; Occhiogrosso, B. IoT Considerations, Requirements, and Architectures for Smart Buildings—Energy Optimization and Next-Generation Building Management Systems. *IEEE Internet Things J.* 2017, 4, 269–283. [CrossRef]
- 54. Wang, J.; Ni, M.; Wu, F.; Liu, S.; Qin, J.; Zhu, R. Electromagnetic radiation based continuous authentication in edge computing enabled internet of things. *J. Syst. Archit.* **2019**, *96*, 53–61. [CrossRef]
- Sikder, A.K.; Acar, A.; Aksu, H.; Uluagac, A.S.A.K.; Conti, M. IoT-enabled smart lighting systems for smart cities. In Proceedings of the 8th Annual Computing and Communication Workshop and Conference, Las Vegas, NV, USA, 8–10 January 2018; pp. 639–645.
- 56. Nikitas, A.; Michalakopoulou, K.; Njoya, E.; Karampatzakis, D. Artificial Intelligence, Transport and the Smart City: Definitions and Dimensions of a New Mobility Era. *Sustainability* **2020**, *12*, 2789. [CrossRef]
- 57. Allam, Z.; Dhunny, Z. On big data, artificial intelligence and smart cities. Cities 2019, 89, 80–91. [CrossRef]
- 58. Brady, H. The challenge of big data and data science. Annu. Rev. Political Sci. 2019, 22, 297–323. [CrossRef]
- 59. Lovis, C. Unlocking the Power of Artificial Intelligence and Big Data in Medicine. *J. Med. Internet Res.* 2019, 21, e16607. [CrossRef]
- Thakuriah, P.; Tilahun, N.; Zeller, M. Big data and urban informatics: Innovations and challenges to urban planning and knowledge discovery. In *Seeing Cities through Big Data: Research, Methods and Applications in Urban Informatics*; Thakuriah, P.T., Ed.; Springer International Publishing: Cham, Switzerland, 2017; pp. 11–45.

- 61. Kitchin, R. The real-time city? Big data and smart urbanism. GeoJournal 2014, 79, 1–14. [CrossRef]
- 62. Colding, J.; Barthel, S.; Sörqvist, P. Wicked problems of smart cities. Smart Cities 2019, 2, 512–521. [CrossRef]
- 63. Google Trends. Popularity of Big Data, Machine Learning, IoT and AI. 2020. Available online: https://trends.google.com/trends/explore?date=all&q=big%20data,artificial%20intelligence,IoT, machine%20learning (accessed on 6 August 2020).
- 64. Serrano, W. Digital systems in smart city and infrastructure: Digital as a service. *Smart Cities* **2018**, *1*, 134–154. [CrossRef]
- 65. Mohanty, S.; Choppali, U.; Kougianos, E. Everything you wanted to know about smart cities: The Internet of Things is the backbone. *IEEE Consum. Electron. Mag.* **2016**, *5*, 60–70. [CrossRef]
- 66. Thakuriah, P.; Tilahun, N.; Zellner, M. Data Introduction to Seeing Cities Through Big Data: Research, Methods and Applications in Urban Informatics. In *Seeing Cities Through Big Data: Research, Methods and Applications in Urban Informatics*, 1st ed.; Thakuriah, P., Tilahun, N., Zellner, M., Eds.; Springer International Publishing: New York, NY, USA, 2017; pp. 1–9.
- 67. Accenture. Accenture Helps Seattle Police Department Implement Data Analytics Platform. Available online: https://newsroom.accenture.com/news/accenture-helps-seattle-police-department-implement-data-analytics-platform.htm (accessed on 5 September 2020).
- 68. The Guardian. Met Police to Begin Using Live Facial Recognition Cameras in London. Available online: https://www.theguardian.com/technology/2020/jan/24/met-police-begin-using-live-facial-recognition-cameras (accessed on 5 August 2020).
- 69. Sadilek, A.; Kautz, H.; DiPrete, L.; Labus, B.; Portman, E.; Teitel, J.; Silenzio, V. Deploying nEmesis: Preventing Foodborne Illness by Data Mining Social Media. *AI Mag.* **2017**, *38*, 37–48. [CrossRef]
- 70. Lavalle, A.; Teruel, M.; Maté, A.; Trujillo, J. Improving Sustainability of Smart Cities through Visualization Techniques for Big Data from IoT Devices. *Sustainability* **2020**, *12*, 5595. [CrossRef]
- 71. Al Nuaimi, E.; Al Neyadi, H.; Mohamed, N.; Al-Jaroodi, J. Applications of big data to smart cities. *J. Internet Serv. Appl.* **2015**, *6*, 25. [CrossRef]
- 72. Purohit, S.; Bothale, V. RFID based solid waste collection process. In Proceedings of the IEEE Recent Advances in Intelligent Computational Systems, Trivandrum, Kerala, India, 22–24 September 2011; pp. 457–460.
- Aguilar, A.; Bonilla-Robles, J.; Díaz, J.; Ochoa, A. Real-time video image processing through GPUs and CUDA and its future implementation in real problems in a Smart City. *Int. J. Comb. Optim. Probl. Inform.* 2019, 10, 33–49.
- Nayak, R.; Behera, M.; Pati, U.; Das, S. Video-based Real-time Intrusion Detection System using Deep-Learning for Smart City Applications. In Proceedings of the International Conference on Advanced Networks and Telecommunications Systems, Goa, India, 16–19 December 2019; pp. 1–6.
- 75. Bajaj, R.; Sharma, V. Smart education with artificial intelligence-based determination of learning styles. *Procedia Comput. Sci.* **2018**, *132*, 834–842. [CrossRef]
- 76. Guo, X.; Shen, Z.; Zhang, Y.; Wu, T. Review on the Application of Artificial Intelligence in Smart Homes. *Smart Cities* **2019**, *2*, 402–420. [CrossRef]
- Chau, K. A review on integration of artificial intelligence into water quality modelling. *Mar. Pollut. Bull.* 2006, 52, 726–733. [CrossRef] [PubMed]
- 78. Wang, P.; Yao, J.; Wang, G.; Hao, F.; Shrestha, S.; Xue, B.; Peng, Y. Exploring the application of artificial intelligence technology for identification of water pollution characteristics and tracing the source of water quality pollutants. *Sci. Total Environ.* **2019**, *693*, 133440. [CrossRef] [PubMed]
- 79. De Paz, J.; Bajo, J.; Rodríguez, S.; Villarrubia, G.; Corchado, J. Intelligent system for lighting control in smart cities. *Inf. Sci.* **2016**, *372*, 241–255. [CrossRef]
- 80. Ajerla, D.; Mahfuz, S.; Zulkernine, F. A real-time patient monitoring framework for fall detection. *Wirel. Commun. Mob. Comput.* **2019**, 9507938. [CrossRef]
- 81. Yigitcanlar, T.; Desouza, K.; Butler, L.; Roozkhosh, F. Contributions and risks of artificial intelligence (AI) in building smarter cities: Insights from a systematic review of the literature. *Energies* **2020**, *13*, 1473. [CrossRef]
- 82. Iskandaryan, D.; Ramos, F.; Trilles, S. Air Quality Prediction in Smart Cities Using Machine Learning Technologies Based on Sensor Data: A Review. *Appl. Sci.* **2020**, *10*, 2401. [CrossRef]
- 83. Iqbal, R.; Maniak, T.; Karyotis, C. Intelligent remote monitoring of parking spaces using licensed and unlicensed wireless technologies. *IEEE Netw.* **2019**, *33*, 23–29. [CrossRef]

- 84. Wang, Z.; Srinivasan, R. A review of artificial intelligence-based building energy use prediction: Contrasting the capabilities of single and ensemble prediction models. *Renew. Sustain. Energy Rev.* **2017**, *75*, 796–808. [CrossRef]
- 85. Bose, B. Artificial intelligence techniques in smart grid and renewable energy systems: Some example applications. *Proc. IEEE* 2017, 105, 2262–2273. [CrossRef]
- Kopytko, V.; Shevchuk, L.; Yankovska, L.; Semchuk, Z.; Strilchuk, R. Smart home and artificial intelligence as environment for the implementation of new technologies. *Traektoriâ Nauk. Path Sci.* 2018, 4, 2007–2012. [CrossRef]
- 87. Maesa, D.; Mori, P. Blockchain 3.0 applications survey. J. Parallel. Distrib. Comput. 2020, 138, 99–114. [CrossRef]
- 88. Marsal-Llacuna, M. Future living framework: Is blockchain the next enabling network? *Technol. Forecast. Soc. Chang.* **2018**, *128*, 226–234. [CrossRef]
- 89. PWC. Blockchain: The Next Innovation to Make Our Cities Smarter. 2018. Available online: http://ficci.in/study-page.asp?spid=22934§orid=41 (accessed on 5 August 2020).
- Aitzhan, N.; Svetinovic, D. Security and Privacy in Decentralized Energy Trading through Multi-signatures, Blockchain and Anonymous Messaging Streams. *IEEE Trans. Dependable Secure Comput.* 2016, 15, 1–14. [CrossRef]
- Novo, O. Blockchain meets IoT: An architecture for scalable access management in IoT. *IEEE Internet Things J.* 2018, 5, 1184–1195. [CrossRef]
- 92. Xie, J.; Tang, H.; Huang, T.; Yu, F.; Xie, R.; Liu, J.; Liu, Y. A survey of blockchain technology applied to smart cities: Research issues and challenges. *IEEE Commun. Surv. Tutor.* **2019**, *21*, 2794–2830. [CrossRef]
- 93. Benedict, S.; Rumaise, P.; Kaur, J. IoT Blockchain Solution for Air Quality Monitoring in SmartCities. In Proceedings of the IEEE International Conference on Advanced Networks and Telecommunications Systems, Goa, India, 16–19 December 2019; pp. 1–6.
- 94. Lin, Y.-P.; Petway, J.; Anthony, J.; Mukhtar, H.; Liao, S.-W.; Chou, C.-F.; Ho, Y.-F. Blockchain: The Evolutionary Next Step for ICT E-Agriculture. *Environments* 2017, 4, 50. [CrossRef]
- 95. European Commission. On Artificial Intelligence: A European Approach to Excellence and Trust. Available online: https://ec.europa.eu/info/publications/white-paper-artificial-intelligence-european-approach-excellence-and-trust_en (accessed on 6 September 2020).
- 96. Allam, Z.; Newman, P. Redefining the smart city: Culture, metabolism and governance. *Smart Cities* **2018**, *1*, 4–25. [CrossRef]
- 97. Mattern, S. Methodolatry and the Art of Measure. The New Wave of Urban Data Science. *Places J.* 2013. Available online: https://placesjournal.org/article/methodolatry-and-the-art-of-measure/?cn-reloaded=1 (accessed on 10 May 2020). [CrossRef]
- 98. Hough, M. Keeping it to ourselves: Technology, privacy, and the loss of reserve. *Technol. Soc.* **2009**, *31*, 406–413. [CrossRef]
- Georgescu, M.; Popescul, D. The Importance of Internet of Things Security for Smart Cities. Smart Cities Technologies. In *Smart Cities Technologies*, 1st ed.; Da Silva, I.-N., Flauzino, R.-A., Eds.; IntechOpen: London, UK, 2016; pp. 3–18.
- 100. Necula, S.-C.; Păvăloaia, V.-D.; Strîmbei, C.; Dospinescu, O. Enhancement of e-commerce websites with semantic web technologies. *Sustainability* **2018**, *10*, 1995. [CrossRef]
- 101. Vranken, H. Sustainability of bitcoin and blockchains. Curr. Opin. Environ. Sustain. 2017, 28, 1–9. [CrossRef]
- 102. United Nations. 68% of the World Population Projected to Live in Urban Areas by 2050, Says UN; United Nations: New York, NY, USA, 2020. Available online: https://www.un.org/development/desa/en/news/population/ 2018-revision-of-world-urbanization-prospects.html (accessed on 5 June 2020).



© 2020 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).



Article



Linking Dynamic Building Simulation with Long-Term Energy System Planning to Improve Buildings Urban Energy Planning Strategies

Lidia Stermieri *^D, Chiara Delmastro^D, Cristina Becchio^D and Stefano Paolo Corgnati

Department of Energy (DENERG), Politecnico di Torino, 10129 Torino, Italy; chiara.delmastro@polito.it (C.D.); cristina.becchio@polito.it (C.B.); stefano.corgnati@polito.it (S.P.C.)

* Correspondence: lidia.stermieri@studenti.polito.it

Received: 21 September 2020; Accepted: 19 October 2020; Published: 22 October 2020



Abstract: The building sector is currently responsible of 40% of global final energy consumption, influencing the broader energy system in terms of new electricity and heat capacity additions, as well as distribution infrastructure reinforcement. Current building energy efficiency potential is largely untapped, especially at the local level where retrofit interventions are typically enforced, neglecting their potential synergies with the entire energy system. To improve the understanding of these potential interactions, this paper proposes a methodology that links dynamic building simulation and energy planning tools at the urban scale. At first, a detailed bottom-up analysis was conducted to estimate the current and post-retrofit energy demand of the building stock. The stock analysis is further linked to a broader energy system simulation model to understand the impact of building renovation on the whole urban energy system in terms of cost, greenhouse gas emission, and primary energy consumption up to 2050. The methodology is suited to analyze the relationship between building energy demand reduction potential and clean energy sources' deployment to shift buildings away from fossil fuels, the key priority for decarbonizing buildings. The methodology was applied to the case study city of Torino, Italy, highlighting the critical role of coupling proper building retrofit intervention with district-level heat generation strategies, such as modern district heating able to exploit low-grade heat. Being able to simulate both demand and supply future alternatives, the methodology provides a robust reference for municipalities and energy suppliers aiming at promoting efficient energy policies and targeted investments.

Keywords: urban energy planning; EnergyPLAN; district heating; dynamic building simulation; future scenario analysis; energy modelling tool

1. Introduction

In the last decade, several actions were deployed to contrast climate change. According to the IEA [1], addressing climate change it is necessary to limit global warming well below 2 °C beyond preindustrial levels in 2050. The European Union, in 2009, decided to plan a reduction of greenhouse gas (GHG) emissions to at least 80% below 1990 levels by 2050. To achieve this goal, the European Climatic Foundation redacted a study analyzing the feasibility of this target [2]. With the new European Green Deal [3], ambitions to reduce CO_2 emission are growing, setting the reduction target for 2030 to 55% and aiming to achieve a net zero emission in 2050. For a future decarbonized energy system, the first step will be improved energy efficiency, especially in building patrimony. Energy consumption of the building sector is 40% of total European energy use, accounting for 36% of associated energy related CO_2 emission [4]. In some areas, space heating accounts for a larger quota of building energy consumption, up to 80% in colder climates [5]. The low energy performance of existing long life

building envelopes are among the major causes of this consumption. A total of 67% of buildings in Europe were built before energy efficiency regulation, and 50% of individual heating systems are fossil fuel boilers with energy efficiency even lower than 60%. For this reason, energy efficiency measures cover a fundamental role in buildings, starting from the renovation and insulation of building envelopes to the improvement of energy systems' efficiency. To strengthen the commitment to building energy saving, in 2017 each European nation delivered a national energy action plan for the 2017–2020 period, using the Directive 2010/31/EU on the energy performance of buildings as a guideline to state new policies and targets for the building's retrofit and the energy efficiency [6].

Energy planning actions on the building sector require, first, the assessment of current energy consumption of the stock and its potential cost-effective energy savings. Many planning approaches consider the energy demand at an aggregated level, where the energy savings potential can be estimated as a percentage reduction of total consumption derived by a gradual achievement of compliance to current regulation as in [7,8].

Detailed analyses to disaggregate the residential stock into different building archetypes or down to the single building level have been developed by several approaches in the last year. The initial approaches were based on census data for statistically distributing buildings consumption according to primary energy use for heating, as proposed by Fracastoro et al. [9]. More recently, heat atlas methodology was used to quantify and locate heat demand in buildings according to climate and building footprint ([10,11]) However, as underlined by Petrović et al. in [12], this approach considered a predefined heat demand reduction and did not analyze heat saving in detail. A solution to this lack of details was proposed in [13] by applying a spatial approach, in which geometrical data coupled with statistical analysis were used to identify building archetypes to which an energy demand is associated through real energy consumption data or by using energy simulation software. The same approach was followed in [14,15], with further calibrating of the bottom-up approach with top-down data. A higher level of disaggregation was reached in the work of Monsalvete et al. [16], in which a modular physical building model was developed in INSEL8 (Integrated Simulation Environment Language) and it was connected with the CityGML urban geometry data model.

These methodologies assessed the energy demand and savings potential of the building sectors with different level of disaggregation, however their application for energy planning purposes is limited, often neglecting heat supply options. The importance of finding a balance between heat savings and heat supply was stressed by Hansen et al. [17], who showed that energy saving measures are less efficient without a related change on the energy production.

Energy interventions planned with a systematic approach can lead to enhanced synergies and flexibility throughout the energy system, in particular in urban areas, which are relevant consumption hubs. In Europe, heat is usually supplied by individual boilers, mostly fossil fuel burning, which in future are expected to progressively shift to electricity (e.g., heat pump) [2], increasing the need for coordination between heat generation and heat networks and the electricity grid. District heating (DH) provides 9% of the EU's heating [5] and presents several advantages thanks to the possibility of integrating different heating technologies and sources [4], including geothermal heat pumps [18], solar energy, and waste heat, as well as conventional sources. Connolly et al. [19] analyzed the advantage of connecting individual heating systems within a heating network in a potential 100% renewable scenario for the European Union. The study found that district heating networks are generally more energy efficient in urban areas, reducing gas emission without relevant additional cost. The penetration of DH networks in energy systems provided evidence to achieve energy savings goals thanks to the utilization of different energy sources, reducing the emphasis on deep heat-saving measures that are often cost prohibitive [20].

Integrated demand–supply analyses are often supported by energy system tools. Connolly et al. [21] provided a review of major energy system tools and their applications. As one of the more widespread, EnergyPLAN can be used to analyze an entire energy system, dividing demand and supply into sectors [19,22]. In the new digital era, with digital technologies taking over the

management of energy sectors, due to its intuitive interface and simple data requirement, this tool can be used for urban application, with it being more understandable to urban stakeholders compared to other energy planning tools (e.g., IMEAS project [23]).

A complete overview of the overall energy system allows municipalities to handle future urban energy planning policies, developing a central role as mediator between different stakeholders from different sectors, being able to lead the construction of a common pathway for achieving future decarbonization goals.

At regional level, this tool was applied in Hong Kong [24], while at the city level, it was applied by [8], underlining that the tool considers only aggregate heat demand and also aggregate supply, with it not being detailed enough to describe alone an urban energy system. To meet the need of a better demand description in urban areas for reflecting the real status of the building stock and to integrate demand–supply analyses, this paper's goals were threefold:

- (1) Develop a spatial-based methodology to represent the urban building demand;
- (2) Link the results of the urban stock analysis with an energy planning tool to analyze both demand and supply decarbonization measures and their economic impact;
- (3) Apply the integrated tool to develop scenarios to support integrated energy planning strategies of an urban municipality.

While urban building stock studies focused on building renovation scenarios are common in literature [25,26], the implications and synergies of demand measures on the supply side are often neglected in urban analyses [27]. The main innovation of the methodology described in this paper is indeed related to its integrated structure, targeting urban analyses and providing an in depth analysis of the energy consumption of entire building stocks and potential energy refurbishment measures, including an additional analysis about the implication that a change in the demand side could have on the supply sector over the long term.

The integrated analysis was performed over a medium-long term period to help urban energy planners in defining technically feasible and economically affordable future energy efficiency scenarios in terms of CO₂ emissions, primary energy supply, and total annual cost. The integrated demand-supply analysis proposed was applied to a case study, the city of Turin. Future scenarios to achieve energy savings goal include the expansion of the existing district heating network of the city and the introduction of renewable energy technologies in the supply.

This paper is structured in four sections: the first one analyzes the case study, the second one explains the methodology, and the third one presents and analyses the results. The applicability of the methodology, the necessary conditions, and its limits are further discussed in the last section.

2. Case Study

The methodology was applied to the case study city of Turin. Turin is a city situated in northern Italy, with 2706 heating degree days (HDDs) at 18 °C. According to previous literature, heating volume of residential building is almost 139 Mm³ [28]. Almost half of the heating volume is satisfied by individual natural gas boilers, while the remaining volume is connected to district heating. The district heating (DH) in the city is the largest in Italy, with 550 km of pipes line and a building volume of 60.3 Mm³ connected to it. The heat production is satisfied for 98% by three different cogenerating thermal plants administrated by IREN Energia [29], with a total of 1200 MW electrical power installed and 740 MW thermal power. The remaining heat need is satisfied by natural gas boilers, with a thermal power installed of 1000 MW. Three thermal storages are distributed in the city, with a total capacity of 430 MW. The expansion of district heating allows for a reduction in energy consumption and the integration of renewable energy sources, without a relevant impact in the total cost, as explained in the introduction. For this reason, the following study was concentrated on the heating volume connected to DH. Real data of heat consumption of district heating were provided by IREN, manager of district heating, for year 2014, with a total annual heat consumption of 1.99 TWh/year (in 2014 district heating

3. Methodology

supported by literature [27,28].

The presented methodology was composed of three key phases. The first one consisted of analyzing the energy consumption for space and water heating of identified residential reference building typologies and the potential for energy retrofit interventions, through dynamic simulation (Section 3.1). As a second step, a spatial analysis based on census data allowed researchers to quantify the distribution of reference buildings within the urban area and consequently to estimate the total energy requirement of the residential stock (Section 3.2). In the last step, the whole energy system was analyzed with an energy system simulation tool [30], allowing researchers to extend the analysis to the supply side and to develop multiple scenarios (Figure 1) (Sections 3.3–3.5).

will be connected to DH. The assumptions of current and future expansion of the district heating are



Figure 1. Scheme of the methodology.

During the first phase, the objective was to understand where buildings' energy-saving interventions are a priority (high-energy/high-density areas) and which level of the retrofit intervention can be reasonably pursued. Achieving this type of detail required the use of a dynamic simulation software to analyze the impact of different energy saving measures on heat consumptions. To analyze the built environment, the authors introduced the widely adopted concept of a "reference building" (RB). RBs are "buildings characterized by and representative of their functionality and geographic location, including in- door and outdoor climate conditions" [6]. Simulating a RB requires a lot of data, especially when referring to dynamic simulation [31], usually difficult to find at an urban scale. The international project Typology Approach for Building Stock Energy Assessment (TABULA), 2009–2012, aimed to define European building typologies [32], providing a robust dataset of reference buildings in European countries, including the ones diffused in North Italy. The data from the TABULA project represented a valid support for this analysis.

The spatial distribution of buildings within the urban area was determined thanks to available census data of the city. This allowed us both to estimate the current consumptions of the building stock, but also their variations under different energy retrofit options. The impact of thermal demand variation on district heat generation plants, as well as new individual solutions, can be further studied thanks to the energy planning tool EnergyPLAN [30]. This allows users to analyze how to coordinate the introduction of new heat supply technologies with building energy retrofit, both from a technical and economic perspective. The presented methodology was shaped to represent a resource for urban planners to analyze in depth buildings energy savings and their effects on the urban energy system.

3.1. Dynamic Building Simulation and Total Heat Consumption

Turin has 36,158 residential and occupied buildings according to census data, which are subdivided into eight construction classes and four building types (Figure 2).



Figure 2. Turin buildings distribution for construction period.

In Turin, almost 96% of buildings were built before 1980, 40% of them are apartment blocks (ABs) and 28% are multifamily houses (MF) (Figure 3). District heating is connected to buildings with a heated volume higher than 2500 m³ [33], so mostly to AB and MF. For this reason, in this study the chosen reference buildings were apartment blocks belonging to the three most frequent construction periods: ABs built from 1945 to 1960, ABs from 1961 to 1975, and ABs from 1976 to 1990.



Figure 3. Pre-1980 building distribution for building type (AB: apartment block; MF: multifamily; TH: terraced house; SF: single family).

For each census section, census data provided information about the number of buildings, in turn separated into four building typologies (AB: apartment block; MF: multi-family; TH: terraced house; SF: single family) and in three age classes (pre-1980, post-1980, post-2005). Also, information about gross heated volume and net surface area were connected to these data.

With the support of GEOPORTALE of Turin [34] it was possible to consult the map with census sections of the city; in this way it was possible to connect territorial distribution of buildings with information about building typologies and construction periods. Considering, for each construction period, a uniform distribution in the city, it was possible to further subdivide census data, and for each section, the number of buildings comparable to the three reference buildings was calculated. Volumes connected to DH were identified, and 30.7 Mm³ of them could be represented with RBs, in particular: 14.1 Mm³ belonged to AB 1945–1960, 10.5 Mm³ to AB 1961–1975, and 6.1 Mm³ to AB 1976–1990. This subdivision demonstrates the correct choice of reference buildings, because 54% of volumes connected to DH were associable with analyzed buildings.

To evaluate a possible network expansion in relation with energy savings measures, another 43 Mm³ were analyzed, considering previous literature and assuming that zones near the river were considered not connectable [33]. Of these, 19.5 Mm³ connectable were represented by RBs: 9 Mm³ to AB 1945–1960, 6.5 Mm³ to AB 1961–1975, and 4 Mm³ to 1975–1990 (Table 1). These 19.5 Mm³, considered as individual heating buildings in the base scenario, will be gradually connected to DH in future scenarios.

Table 1. Volumes connected to district heating (DH) subject to study and future network expansion.

RB Typology	Connect to DH	Possible Expansion
AB 1945–1960	14.1 Mm ³	9 Mm ³
AB 1961–1975	10.5 Mm ³	$6.5 \mathrm{Mm^3}$
AB 1975–1980	6.1 Mm ³	4 Mm^3
tot	30.7 Mm ³	19.5 Mm ³

The TABULA project supported the association of thermal characteristics to every RB. The project presented a matrix with 32 different types of building [35,36], and a schedule for each of them containing the heated volume, the floor area, the number of apartments, the number of floors, the surface to volume ratio (S/V), the construction materials with total U-values, energy systems with their energy efficiency, and heat consumption result from steady-state calculation. Other information, such as building shape or number of windows, were derived by author assumptions and set the basis for the following reference building dynamic simulations.

Stratigraphies were hypothesized for roof, floor, and walls (two types of wall were simulated for each RB). A lower insulation was performed, as requested in TABULA ($U \approx 0.8 \text{ W/m}^2\text{K}$), and material data and thermo-physical properties were taken from the Italian National Unification (UNI) standards UNI 10355 and UNI 10351 [37], while for resistances the reference norm was UNI 6946 [38]. All structures neighboring with nonheated zones, as suggested in UNI 6946, provided a superficial resistance equal to the internal superficial resistance for each side. A building layout was created starting from available data of floor area, gross volume of building, and compactness (S/V). RB simulations were performed by DesignBuilder software, a graphic interface based on EnergyPlusTM, a building energy simulation program used to model both energy consumption and water used in buildings [39]. Heat energy requirement was calculated following TABULA indications and procedures reported in UNI part one [40], in this way, simulations results could be compared to TABULA results (Table 2), where the difference in energy requirement between the two models was in the order of 1% for each RBs.

Reference Building Simulation		AB 1945–1960	AB 1961–1975	AB 1976–1990	
Geo	metric data				
Parameter	Symbol	Unit			
Gross heated volume	V	m ³	5949	9438	
Net floor area	A _n	m ²	1763	2869	4123
Shape factor	S/V	m^{-1}	0.46	0.46	0.37
N° apartments	-	-	24	40	48
N° floors	-	-	4	8	6
Const	ruction data	l			
Parameter	Symbol	Unit			
Roof	U	W/m ² K	1.8	2.20	1.85
Walls	U	W/m ² K	2.6	1.10	0.76
Lower floor	U	W/m ² K	1.65	1.65	0.97
Upper floor	U	W/m ² K	1.3	1.56	0.98
Windows	U	W/m ² K	4.9 (g _{'gl} 0.85)	4.9 (g _{,gl} 0.85)	3.70 (g _{'gl} 0.75)
Н	eat need				
Reference building standard		kWh/m ² y	159	136	66.3
Reference building results		kWh/m ² y	139	124	61.3

Table 2. Reference buildings simulation data.

These simulations calculated only the space heat requirements, the "useful energy", while information about energy system typologies and the relative energy efficiencies were further inserted as input data in the energy planning tool.

The energy analysis performed under the previous assumptions is called "standard" evaluation, but to take in care occupant behavior, the intermittency of heating and the operating schedules of electrical equipment, it was necessary to relax the assumption made following the normative in previous simulation. In literature, several studies have considered the influence of occupant behavior on heat consumption [41,42]. In this paper this argument is not stressed, since the aim was to use reference building to simulate the entire built environment of the city, therefore taking into account a representative occupant behavior pattern. Internal gains as internal light, electrical equipment, and people were aggregated with a single operating schedule, called "internal gain schedule". Hourly profile of the schedule was extrapolated from UNI 11300 part 1 [40], this was done in order to have a reliable hourly profile.

The simulation results showed a heat requirement of 160 kWh/m²/y for AB 1945–1960, 136 kWh/m²/y for AB 1961–1975, and 66 kWh/m²/y for AB 1976–1990.

The hourly simulation provided for each RB (Appendix B) has been normalized with gross heated volume of RB, as done in [43], and then multiplied for respective volumes (Table 1), creating the total heat profile 60% of the total DH heat demand.

To identify necessary retrofit interventions to comply with new energy efficiency requirements, the authors referred to the environmental energy appendix of building regulation of Turin [44]. In the Appendix A, two restriction levels to thermal insulation are described: the first level is about prescriptive limits, while the second one concerns incentive requirements (an economic incentive

is settled for the buildings that satisfy a higher thermal insulation to further minimize the energy consumption). Heat loads can be reduced using materials with a low thermal transmittance; it is important to define a global strategy for building thermal insulation choosing a correct insulation material and relative thickness (insulation materials chosen related to prospect 2 in UNI:2015 [37], for hygrometric properties the reference was UNI:2008 prospect 4 [45]). It was assumed also that thermal bridges were corrected by retrofit measures.

Three scenarios were created to analyze the energy savings potential in buildings (Table 3):

- Scenario WS: substitution of windows (U = 1.20 W/W, $g_{\text{rgl}} = 0.30$)
- Scenario SR (standard renovation): all building envelopes are insulated according to transmittance limits in Level 1 [39]
- Scenario AR (advance renovation): all building envelopes are insulated according to transmittance limits in Level 2 [39]

The WS scenario allows the possibility of reaching a 10% of energy savings with respect to base scenario, 80% with the SR, and 88% with AR, considering the application of the energy saving interventions to all the volume connected to DH represented by the RBs (30.7 Mm³).

Table 3. Heat consumption of reference buildings (RBs) in different scenarios of building retrofit.

Typology	Base (kWh/m ² y)	WS (kWh/m ² y)	SR (kWh/m ² y)	AR (kWh/m ² y)
AB 1945–1960	158.8	142	29.5	19
AB 1961–1975	135.6	123.4	27.5	16.2
AB 1976–1990	66.3	58.2	17.5	11.3

3.2. Modeling District Heat Hourly Profiles

As a second step, the estimated total heat consumption needed to be calibrated to match the urban energy balance allocated to space heating and district heat production. At first, local district heat production was characterized. By analyzing DH real operational data for the years 2012, 2013, and 2014, average heat losses in the network resulted equal to 10% of total heat production. For district heated buildings, the DH quota dedicated to domestic hot water (DHW) was derived by analyzing the summer load; in particular, the average profile of a summer day (in which the heating systems are off according to the Italian normative) was extrapolated from DH real data and chosen as reference for DHW needs and assumed equal for every day of the year. The calculated DHW demand and the losses were subtracted from the total DH demand. By doing this, the remaining DH heat consumption could be compared to the heat profile created through RB simulations (Figure 4).



Figure 4. DH total real heat profile (red line) compared to RBs total heat profile (blue line) for a day in January. The blue line includes only the heat quote of buildings related to the RBs simulated (covering almost the 54% of the total number of buildings), without the DHW contribution.

Although considering standard values comports uncertainties because both internal loads and infiltration rates are largely influenced by the occupant's behavior (they can vary from building to building, but also from floors of the same building), as analyzed in [46], using a dynamic simulation tool is the best solution for taking into consideration occupant's behavior. Figure 4 validates the results from the reference building simulation, with the simulated heat profile (blue line) being almost 60% of the total DH profile (red line).

Building volumes not connected to DH were considered heated by individual boilers, and with the same method heat profiles were created. Retrofit actions (WS, SR, AR) were gradually analyzed in the same way, and different heat load profiles were compared to DH real data (Figure 5).



Figure 5. Energy savings percentage according to building retrofits for different scenarios. * Scenario 1: DH network expansion (19 Mm3). ** Scenario 2: DH network expansion 31 Mm³.

The decrease of heat consumption comporting an increase in heat loss was determined according to Equation (1) [47]. This equation for the estimation of heat losses was previously applied to the same network as in [26,28,48].

$$y = 0.7321 - 1.221x + 0.6462x^{2} \text{ if } x < 1$$

$$y = 0.1773 - 0.014x \text{ if } x > 1$$
(1)

in which y = loss (%) and x = linear heat density in MWh/m.

Considering 500 km of pipelines, linear density and the correlated heat losses were calculated for each building's energy saving scenario. In the year 2017, the linear density was equal to 3.5 MWh/m, and it decreased in each scenario. With a line density lower than 2 MWh/m, heat losses were considered too high and the DH network became not convenient. SR and AR scenarios provided a linear density lower than 2 MWh/m, which means 15% of heat loss—to increase the linear density up to 2 MWh/m, network expansions scenarios were associated with retrofit measures for each scenario. For the WS scenario, a lower network expansion was considered, equal to 2.9 Mm³, and for SR and AR a higher expansion was supposed, equal to 19.5 Mm³ (the remaining volume of all buildings with individual heating). This kept the linear density higher than 3 MWh/m in each scenario, and maximum heat losses were in the order of 13%. The heat losses were added to the heat profile consumption of DH and individual buildings and, together with DHW consumption, the heat load profile as input for thermal plant was created (Figure 5).

3.3. Energy Savings Costs

Costs associated to energy savings measures were derived from "Prezzario Regione Piemonte 2016" [49]. For each RB, 3 different costs analyses were done, one for each type of building retrofit option (WS, SR, AR). Cost included: construction material/technology cost, installation, and manpower costs.

Analyses with EnergyPLAN also required energy prices and investment cost data as input (investment, operations, fixed and variable maintenance, and manpower costs). Fuels cost, handling costs, and variable operations and maintenance (O&M) costs were also required for each type of technology, together with the eventual CO_2 price and the discount rate (this was assumed equal to 3%). The additional costs, as energy savings measures and network expansion, were inserted in EnergyPLAN in terms of total annual cost (Appendix A).

3.4. Global Cost Analysis

The total annual cost provided by EnergyPLAN (according to Equation (2), as reported in [50]) related to a single year in which all investments were supposed to be applied. In this way, it was possible to compare different annual costs provided from different energy efficiency investments, to understand which is the most profitable solution among the considered options.

$$A_{inves} = (C_{invest} * i)/(1 - (1/(1+i)^n))$$
(2)

with C_{invest} = investment costs; i = interest; n = lifetime.

Each investment with a different lifetime could therefore be compared as an equivalent annual cost. Total annual costs (M \notin /yref) were therefore derived by summing A_{inves} with fixed and variable annual costs.

To analyze the global cost of each scenario, according to a more realistic distribution of the investment along the timeline, the methodology in [28] and in [14] was followed: a time period horizon was defined (2017–2050) and divided into 3 period steps (2017–2025, 2025–2035, 2035–2050).

The renovation rate of buildings was set equal to 3% for each year starting from 2018, while DH expansion was planned with a year rate of 4% starting in 2025 (Table 4). Investments in thermal plants were planned according to the end-life of each system based on starting year [51,52], as explained in Table 5.

Table 4. Building retrofit and	d DH expansion rate for	r time period. * Sta	arting from 2018.

	2017–2025	2025–2035	2035–2050
Building retrofit	3% renovation volume for year *	3% renovation volume for year	3% renovation volume for year
DH expansion		4% connected volume for year	4% connected volume for year

Table 5. Thermal plant investments for time period.

	2017-2025	2025–2035	2035–2050
BAU	No investment	New natural gas CHP (662 MW _{el}), new industrial boiler (1000 MW), new heat storage (430 MW)	New natural gas CHP (360 MW _{el})
MOD	No investment	New natural gas CHP (662 MW _{el}), new industrial boiler (1000 MW), new heat storage (430 MW), heat pump (35 MW _{el})	New natural gas and biomass CHP (338 MW _{el})
ADV	No investment	New natural gas CHP (662 MW _{el}), new industrial boiler (1000 MW), new heat storage (1000 MW), heat pump (85 MW _{el})	New biomass CHP (134 MW _{el})

Building renovation rate and DH expansion rate were used to create a heat distribution demand for DH and individual buildings for each time step. Time periods were simulated considering a linear CO_2 cost and natural gas cost evolution, as explained in Section 3.5.

Variable annual costs and fixed annual costs for the three periods were assumed considering a discount rate of 3%; a present value factor (Ri) was calculated for each mid-year of the rime periods (4, 13, 25.5), and the total discounted cost at 2017 was calculated according to Equation (3).

$$C_{\text{tot}} = \sum_{i=1}^{n} (n = 3) \left[(C_{\text{inv}}(RB,i) + C_{\text{inv}}(dh,i) + (C_{\text{fix}}(i) + C_{\text{var}}(i)) * t) * R(i) \right]$$
(3)

with C_{tot} = total cost discounted at 2017 level, $C_{inv}(RB,i)$ = investment cost for building retrofit on period i, $C_{inv}(dh,i)$ = investment cost on DH expansion and heat supply on period i, C_{fix} and C_{var} = fixed and variable costs (\notin /y) on period i provide from EnergyPLAN, t = years in period I, R(i)= present value factor for mid-year of each time step.

3.5. Heat Supply Scenario

The existing synergy between district heating expansion and buildings' energy retrofit interventions has been demonstrated in several studies [19,20,28]. The production of heat in the DH system is possible through several options: renewable energies (e.g., solar and geothermal), heat pumps, conventional fossil fuels, or even by exploiting waste heat from industrial processes. Taking into account the current heat generation mix of the Turin DH system (more than 90% heat production based on gas cogeneration plants), this study focused its attention on combined heat and power (CHP) technologies and on the integration of large-scale heat pumps and storage. Two scenarios were analyzed, according to [28], considering already planned power plant installation (biomass CHP) and commercial available capacities. In future scenarios, CHP plants could operate to balance the electricity needs of heat pump.

- MOD: moderate scenario, decommission of 260 MWth of natural gas CHP in 2030, replaced by 105 MWth of heat pump, 106 MWth of biomass CHP, and 150 MW of natural gas CHP.
- ADV: advanced scenario, decommission of 260 MW of natural gas CHP, replaced by 250 MWth of heat pump, 106 MWth of biomass CHP, and 530 MW of daily thermal storage.

Biomass CHP was chosen considering the huge availability of biomass, given that the city of Turin is close to the mountain communities, though an analysis on local emissions should be explored.

To compare the proposed demand–supply options with a scenario reflecting current trends, a business as usual (BAU) scenario was considered (any change in heat consumption and production). This scenario was used as a reference to be compared with the other simulations in terms of energy savings, total primary energy supply (TPES), CO₂ emissions, and total annual cost for year 2050. In the BAU scenario, costs for maintenance and replacement of current power plants after the end of their technical lifetime and power plant investment costs were considered, while change in fuel prices and CO₂ price were considered following the same assumptions as the ones made in the Stratego project [22], assumed equally suitable for an Italian context. A linear growth was considered for the carbon price in the scenario timeline, starting with the assumption of a carbon price equal to $6 \notin/tCO_2$ in 2017, as reported in [53], and an optimistic carbon price of $25 \notin/tCO_2$ for 2050, following the hypothesis made on Heat Roadmap 2050, in which a price of $20–30 \notin/tCO_2$ is expected. By combining scenarios on retrofit measures with scenarios on heat generation, different integrated scenarios were created up to 2050 (Table 6).

Building Retrofit	WS	Windows Substitution	
	SR	Standard renovation	
	AR	Advanced renovation	
Network expansion	1	2.9 Mm ³	
*	2	19.5 Mm ³	
Heat production change	MOD	105 MW HP	
		106 MW CHP biomass	
		250 MW HP	
	ADV	106 MW CHP biomass	
		560 MW heat storage	

Table 6. Scenario descriptions.

Hourly consumption profiles obtained for individual buildings in the previous section were used as input in EnergyPLAN for describing the future evolution of the heat demand, starting from the current DH real profile provided by the local utility assumed as reference for the scenario BAU. Individual heating systems were considered as supplied by natural gas boilers with an average efficiency of 0.85. Each cost was introduced in the base year simulation for 2017, and a technical optimization (optimizing capacity sizing and energy use) was performed.

The final heat demand (district heated buildings plus individual heating) was 2.65 TWh/y, 1.99 TWh/y for DH and 0.66 TWh/y for individual heating (19.5 Mm₃). DH demand was supplied by natural gas CHP (1.93 TWh/y) and natural gas boilers (0.06 TWh/y). Total primary energy supply (TPES) was equal to 6.19 TWh/y, disaggregated into 5.41 TWh/y for CHP + boilers and 0.78 TWh/y for individual heating. The primary energy supply was calculated as the fuel consumptions needed from the energy technologies to satisfy the energy demand of the system. The fuel consumption of the plants was calculated by the model considering technologies' efficiencies. CO₂ emissions were 1.250 Mt, and total annual cost (including investment cost, fixed and variable cost, fuel cost, and carbon price) was 295 M€/y.

Three different buildings retrofit scenarios with two different levels of DH expansion options were compared for three different heat supply mixes (Table 7).

Table 7. Different combinations of building retrofit, network expansion, and heat production mix. * Network expansion: 1. expansion of 2.9 Mm³; 2. expansion of 19.5 Mm³. ** HP: heat pump; CHPB: biomass combined heat and power (CHP); HS: heat storage.

		Building Retrofit (WS,SR, AR) and Network Expansion (1,2) * Scenarios				
		Base	WS + 1 (WS: Windows substitution)	SR + 2 (SR: standard renovation)	AR + 2 (AR: advanced renovation)	
Supply side scenarios	Base MOD	BAU	А	В	С	
	(105 MW HP **, 106 MW CHP _B)		D	Ε	F	
	ADV (250 MW HP, 106 MW CHP _B 560 MW HS)		G	Н	Ι	

The scenarios A, B, and C consider the same supply configuration as the BAU scenario, integrating 3 different level of building retrofit. The introduction of a biomass CHP and heat pump (HP) of the MOD heat supply scenario correlated with the building retrofits in scenarios D (windows substitution), E (standard renovation), and F (advanced renovation). The same retrofits were considered for scenarios G, H, and I, together with ADV heat supply scenarios.

As first analysis, the total annual cost of these scenarios was analyzed for the timeline 2017–2050 according to Equation (3).

4. Results

At first, the achievable CO_2 emission reduction considering buildings retrofit together with DH expansion without changing in heat supply was analyzed (A, B, C). Compared to the BAU scenario, this reduction is equivalent to 2% for A, 15%, for B and 18% for C (Figure 6). However, demand side measures alone are not enough to achieve 40% emission reductions to 2030, as requested by [54].



Figure 6. CO₂ emissions for different scenarios from EnergyPLAN simulation.

Assuming a "moderate change" in the heat generation mix (MOD scenario) can allow a higher reduction in CO₂ emissions (42%) and TPES (40%), thanks to the integration of heat pumps (HP) and the biomass CHP, relying on low carbon electricity (2050 carbon intensity equal to 0.66 Mt/y), maintaining a natural gas CHP working for 57% of the time in cogeneration mode for heat production. The annual operating time of CHP higher than 4000 h can assure a positive profit for the investment. The optimization algorithm coordinates electricity production by CHP units to heat pumps' heat generation—CHP operation follows the electricity need of HP and heat storages are used to store CHP heat production excess. This optimization is related to the decision of balancing only the heat consumption, in future studies it might be interesting to study the impact on the electricity market as well. Furthermore, CHP production is primarily satisfied by biomass CHP to reduce the CO₂ emissions. Natural gas consumption and related carbon emissions decrease further with a higher integration of heat pumps in the ADV scenario. Together with the energy savings due to building retrofit measures, the integration of heat pump in MOD scenario guarantees 21% of TPES reduction, while ADV scenario with 250 MW of heat pumps allows a TPES reduction of 49% (Figure 7).



Figure 7. Total primary energy supply (TPES) outputs from EnergyPLAN simulations.

Cost outputs were further analyzed (Figure 8) in order to understand to what extent each intervention influences the global cost. In the BAU scenario, total annual costs would be around 460 M€/y in 2050. Fuel cost is the prevalent cost item, covering 58% of total annual costs for BAU.



Figure 8. Total annual cost subdivision for each scenario from EnergyPLAN outputs.

Compared to the BAU scenario, each intervention regarding change in heat production (D, E, F, G, H, I) connected with a reduction in heat demand provides a cost decrease. In the scenarios without changes in heat production (A, B, C), fuel costs represent 51% of total costs. In MOD scenarios (D, E, F), this cost voice decreases to 39% and it reaches a minimum value of 27% in ADV scenarios (G, H, I). Lower fuel costs due to fuel shifting in heat production mitigate investment costs for buildings retrofit and for heat generation interventions.

The ADV scenario with standard renovation in buildings, together with DH expansions, provides lower costs with respect to all the other scenarios simulated. Compared to a standard renovation, an advanced renovation on buildings comports a higher reduction in heat consumption, but also higher investment costs. These investment costs are not offset by the savings, highlighting the critical role of defining correct retrofit measures for existing buildings. Meanwhile, considering the very old building stock and the presence of a district heat network, a standard retrofit scenario (H) can provide a better balancing between new investment costs and decreasing costs as a consequence of energy savings measures.

Results from energy system simulations show the synergy between buildings' retrofit interventions and changes in district heating, in terms of network expansion and heat production configuration. From a planning perspective, combining buildings' energy savings with DH expansion strategies, together with the integration of cleaner heat generation technologies as heat pumps, allows significant reductions in terms of CO₂ emission, total costs, and primary energy supply. Larger interventions compared to baseline, as the one proposed in the H scenario, could allow reaching reductions of 78% in CO₂ emission (corresponding to a reduction of 970 MtCO₂/y) and of 72% in TPES, with benefits in terms of annual costs equivalent to 150 M€/y.

To understand which is the overall marginal cost for the energy system decarbonization in the different scenarios, a "decarbonization cost" Cdec (\notin /tCO₂) was calculated, as in Equation (4).

$$Cdec (i) = (Cin (i) - Cin(BAU))/(CO_2(BAU) - CO_2 (i))$$
 (4)

With Cin(i) = investments costs for scenario i, Cin(BAU) = investment costs in the BAU scenario, $CO_2(i) = CO_2$ emission in scenario i, $CO_2(BAU) = CO_2$ emission in BAU scenario.

The decarbonization cost is the rate between the increasing in investment cost and the CO_2 reduction due to low-carbon measures. It can be used as an index to identify cost-effective decarbonization investments.

From the results, it is possible to affirm that for scenarios with change in the heat Supply (D–I) the decarbonization cost is lower than $100 \notin tCO_2$ saving, while for scenarios A, B, and C (no change in the heat production mix, but progressive demand reduction thanks to building renovation) the decarbonization cost is almost triple compared to integrated solutions involving the MOD and ADV scenarios (Figures 9 and 10). In scenarios A, B, and C, investment costs are higher than in BAU due to building retrofits, but CO_2 reduction is not enough to justify investments. This consideration is key for building stock retrofits, where the potential for deep retrofit intervention is often limited by technical feasibility or affordability, which needs to be properly defined.



Figure 9. Decarbonization costs in different scenarios.



Figure 10. Investment costs and related CO₂ emissions for each scenario.

Despite the fact that the cost of decarbonization can already represent a good economic indicator, greater insights can be derived from a more detailed economic analysis that considers the cost distribution over the whole time horizon. As shown by Figure 11, compared with the annual cost analysis performed with EnergyPLAN, a more detailed global cost analysis shows that scenarios with improved heat supply (MOD and ADV) can deliver bigger cost reductions. On the other hand, by applying this methodology, the longer term benefits of advanced building retrofit (C, F, I) can be observed more precisely [55]. This might help to deliver improved energy retrofit considerations.



Figure 11. Total costs discounted at 2017 level for different scenarios, divided into investment costs and fix + var costs.

A sensitivity analysis variating discount rates might improve the understanding of which retrofit measures may have reasonable payback times—a lower discount rate presents higher influence of future energy costs, while a higher discount rate presents lower influence on future costs and higher influence of investment costs. In Figure 12, total costs for each scenario were analyzed with different discount rates. Comparing BAU scenario with a moderate scenario (for example E), a higher discount rate comports a lower costs difference between BAU and E, this means that a BAU scenario with few energy efficiency investments can be more convenient with respect to a scenario with bigger energy investments (e.g., E,C) if discount rate increases (or equivalently, without proper policies). Targeted tax policies can represent a useful instrument to push the energy efficiency measures and unlock the necessary fiscal resources for driving a clean energy transition.



Figure 12. Discount rate impact on total costs for different scenarios.

5. Discussion

Linking energy system analyses to dynamic building simulations to analyze different urban energy patterns connects a data-driven robust building-level analysis with a technology-rich supply-side understanding, allowing researchers catch their interactions. The opportunity to study together the demand side and the supply side allows a better understanding of the decarbonization potential of a city, especially if characterized by an ancient building stock, as in many European cities. A complete
overview of the overall energy system allows municipalities to handle future urban energy planning policies, developing a central role as mediator between different stakeholders from different sectors, being able to lead the construction of a common pathway for achieving future decarbonization goals. The application of EnergyPLAN for national-level studies is well known in literature, but the proposed methodology allows us to expand its use to an urban level without renouncing the level of detail required from the urban environment. The methodology can be further improved considering the increasing data availability in the residential sector (cadaster, smart meters, energy certifications), allowing the calibration of building simulations with real data.

EnergyPLAN was chosen because it provides several advantages: it's a free download tool; on the web page is possible find documentations, guides, and tutorials to help users; input data are easily to find; and it has a user-friendly interface. On the other hand, as previously identified, EnergyPLAN shows some limits in the analyses of the supply side. The technical simulation to balance the heat demands applied by the tool relies on a predefined hierarchy of technologies (solar thermal, industrial CHP, heat production from waste fuel, heat plant CHP, heat pumps, peak load boilers), limiting the possibility of choosing other technology combinations or different parameters to optimize the energy system. Although, this limitation can be overcome by utilizing the MATLAB Toolbox for EnergyPLAN [56]. Furthermore, it requires aggregated input data for production that do not allow users to analyze each thermal plant singularly. and this means that the introduction of a new thermal plant (e.g., biomass CHP in this case study) cannot be analyzed individually. Usually power plants are projected to satisfy different cities, while an aggregated simulation requires a sort of close control volume, in which entry flows and exit flows are well known. However, these uncertainties are limited for the previous case study because the studied power plants were designed to satisfy mainly the energy demand of Turin. To extend the methodology to other municipalities, a collaboration with the utilities is suggested to increase the representation of the supply side, as done with this paper for the demand side.

As shown by this paper, a discretization of the urban demand through the identification and simulation of reference buildings is a valuable contribution to improving the understanding of building retrofit potential, the reference buildings approach being useful to identify an average behavior for the residential sector. Despite this, in the future, with growing data availability thanks to the spread of digitalization technologies, data-driven approaches might be more suitable for this purpose, overcoming barriers related to RBs applied to large-scale analyses, such as catching shading profiles of closed buildings, internal loads, and infiltration rates, as explained in [43].

Further improvements in the methodology can be related to improving the representation of the influence that the DH network itself has on the heat demand. Delay effect and heat loss, together with mixing effects due to return flows of each building, comport a heat load profile of thermal plants different from that of buildings, especially from the peak load, which is different for magnitude (lower than user peak) and for duration (peak of thermal plant lasts for hours, for buildings lasts for few minutes). For a better demand and supply relationship evaluation, a future development of the methodology could integrate a tool to simulate the network to analyze load profile of thermal plant and to apply regulation efficiency intervention in building (as the anticipation of the thermal request of buildings to reduce morning peak analyzed in [57]).

Additional improvements in future analysis can be done by integrating the effectiveness of buildings savings related to the increase of temperatures (e.g., lower heating degree days might reduce the costs benefits related to energy retrofit, reducing its profitability) [46].

6. Conclusions

The aim of the proposed approach was to investigate the relation between buildings' energy savings and low-carbon heat generation strategies at an urban scale, to help municipalities to lead the transition towards a decarbonized energy society through a targeted planning of future investment and energy policies. For the achievement of the European goals in terms of CO₂ emission reduction and

1259

primary energy consumption reduction, it is important correlate a demand reduction with a low-carbon supply strategy. The correlation of these two aspects is evident in the residential sector, especially in the historical buildings and general old buildings that represent the highest share of the residential sector in European cities. For these buildings, where the installation of new technologies, such as decentralized heat pumps, is difficult (due to either social, economic, and technical reasons), the role of the district heating, thanks to its easy integration with several low carbon technologies, is very relevant to pursue the goals of the energy transition. The proposed methodology allows an in-depth analysis on effective measures applicable for heat saving at the building level, integrated with an analysis of the energy system supply performed with the EnergyPLAN tool. The integration and expansion of DH in the city was analyzed through different hypothesis of future energy system configurations. The results underline the correlation between the demand side and the supply side, showing that a joint effect of both demand reduction at building level and change in the heat supply is needed to achieve the energy savings goals. Scenarios with changes in heat production provide a big reduction on CO₂ and TPES, due to the integration of heat pumps, and on biomass CHP. In particular ADV scenarios, with the integration of 250 MW heat pumps, 560 MW of heat storage, and 106 MW of biomass CHP comport a CO₂ reduction from a minimum of 56% (without considering buildings savings and DH expansion) to a maximum of 76% considering advanced renovation level for building retrofit together with DH expansion. The integration of large-scale heat pumps in district heating provides a strong reduction in natural gas consumption, and a lower dependence on natural gas assures a safe energy system. New investments in district heating and in building stock are supported by the strong decrease of fuel consumption and its relative costs. However, the cost reduction of these scenarios is supported by a presumed increase in the cost of fossil fuel and in the carbon taxes on 2050. The cost-effectiveness of the energy savings investment is strongly related to the trend of future energy policy decisions that can provide a strong push for embrace a low carbon energy system.

Author Contributions: Conceptualization, methodology, formal analysis, writing-original draft preparation: L.S. Writing—review and editing, supervision: C.D. Supervision, project administration: C.B. and S.P.C. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Acknowledgments: The authors would like to thank Roberta Roberto from ENEA (Italian National Agency for New Technologies, Energy, and Sustainable Economic Development) working for the project IMEAS.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

	AB 1945–1960	AB 1961–1975	AB 1976–1990	Average
WS (€/m ³)	10.68	9.96	5.12	8.59
SR (€/m ³)	34.23	34.35	25.61	31.40
AR (€/m ³)	46.27	46.01	35.09	42.46

Table A1. Average costs of building retrofit.

Table A2. Total cost of energy savings measures for volumes connected to DH.

	Average (€/m ³)	Volumes DHMm ³	Total Cost (M€)
WS	8.59	30.7	263.62
SR	31.40	30.7	963.6
AR	42.46	30.7	1303.38

	Investment Cost (M€)	Period (Years)	Fixed O&M (%)
Small CHP units	0.85	25	3.75
Heat storage CHP	3	20	0.7
Large-scale heat pump	1.7	25	0.9
Boilers (CHP)	0.12	20	3.8
Boilers (residential)/1000 units	12	25	3.8

Table A3. Investment and fixed operations and maintenance (O&M) [24].

Table A4. Fuel costs in the year 2017 [21].

	Natural Gas (€/GJ)	Biomass (€/GJ)
Fuel price	6	5.65
Fuel handling costs (distribution and refinery):	2.05	1196
To individual house holds	3146	-

Table A5. Additional costs [24].

	Period (Years)	O&M (%)	Investment (€/KW)	Total Investment (M€)
Existing DH pipes + heat exchanger	40	100	10.20	5.92
New DH pipes + heat exchanger	40	1	1018.40	WS: 30.54 SR: 292.45 AR:292.45
Heat saving retrofit	30	-	-	WS: 263.62 SR: 963.6 AR: 1303.38

|--|

2017	2050
6	11.83
5.65	8.1
6	25
	2017 6 5.65 6

Appendix B

 Table A7. Geometric and construction data, apartment building 1945–1960 [36].

		AE	8, 1945–1960	
	Geometric D	Data		
Parameter Gross heated volume Net floor area Shape factor N° apartments N° floors Gross heated surface Gross heated floor Windows area	Symbol V An S/V - S A ₁ W	Unit m^3 m^2 m^{-1} m^2 m^2 m^2 m^2	5949 1763 0.46 24 4 2746 1595 217	
Reference floor area	A _{ref}	m ²	1478	

AB, 1945–1960						
	Construction	n Data				
Roof	U	W/m ² K	1.8	Space heating profile		
Walls	U_1	W/m ² K	1.15	160		
	U_2	W/m ² K	2.6	140		
Lower floor	U	W/m ² K	1.65	120		
Upper floor	U	W/m ² K	1.3	100		
Window	U	W/m ² K	4.9	₹ 80		
	-			60		
				40		
			0.05	20		
	g, gl	-	0.85	0		
				1 3 5 7 9 111315171921		
				hour		

Table A7. Cont.

Table A8. Geometric and construction data, apartment building 1961–1975 [36].



Ce	onstruction D	ata		
Roof	U	W/m ² K	2.2	Space heating profile
Walls	U_1	W/m ² K	1.1	45
	U_2	W/m ² K	1.13	45
Lower floor	U	W/m ² K	1.65	35
Upper floor	U	W/m ² K	1.56	30
Window	U	W/m ² K	4.9	≥ 25
				± 20 15
				10
	g, gl	-	0.85	5
				0
				1 3 5 7 9 11131517192123
				Hour

		AB,	1976–1990	
Geo	metric Data	1		
Parameter Gross heated volume Net floor area Shape factor N° apartments N° floors Gross heated surface Gross heated floor Windows area Reference floor area	Symbol V A _n S/V - S A ₁ W A _{ref}	Unit m^3 m^2 m^{-1} - m^2 m^2 m^2 m^2	4123 0.37 48 6 4734 3803 361 3474	
Cons	truction Dat	ta		
Roof Walls Lower floor Upper floor Window	U U ₁ U U U U g _{'gl}	W/m ² K W/m ² K W/m ² K W/m ² K W/m ² K	1.85 0.76 0.76 0.97 0.98 3.7 0.75	space heating profile 60 50 40 20 10 0 1 3 5 7 9 11 13 15 17 19 21 23 hour
		DH total	heat prof	ile
1200 1000 800				

Table A9. Geometric and construction data, apartment building 1976–1990 [36].



Figure A1. Heat load profile: losses, domestic hot water (DHW), and heat consumption demand for district heating.

References

- 1. International Energy Agency Energy. *Climate Change and Environment 2016 Insights;* Internal Energy Agency: Paris, France, 2016; p. 133. [CrossRef]
- 2. European Commission. Roadmap 2050. Policy 2012, 1–9. [CrossRef]
- 3. European Commission. The European Green Deal. Eur. Comm. 2019, 53, 24. [CrossRef]

- Sayegh, M.A.; Danielewicz, J.; Nannou, T.; Miniewicz, M.; Jadwiszczak, P.; Piekarska, K.; Jouhara, H. Trends of European research and development in district heating technologies. *Renew. Sustain. Energy Rev.* 2017, 68, 1183–1192. [CrossRef]
- 5. EC. An EU strategy on heating and cooling 2016. J. Chem. Inf. Model. 2016, 53, 1689–1699. [CrossRef]
- 6. EU. Directive 2010/31/EU of the European Parliament and of the Council of 19 May 2010 on the energy performance of buildings (recast). *Off. J. Eur. Union* **2010**, 13–35. [CrossRef]
- 7. Lund, H.; Möller, B.; Mathiesen, B.V.; Dyrelund, A. The role of district heating in future renewable energy systems. *Energy* **2010**, *35*, 1381–1390. [CrossRef]
- 8. Brandoni, C.; Arteconi, A.; Ciriachi, G.; Polonara, F. Assessing the impact of micro-generation technologies on local sustainability. *Energy Convers. Manag.* **2014**, *87*, 1281–1290. [CrossRef]
- 9. Fracastoro, G.V.; Serraino, M. A methodology for assessing the energy performance of large scale building stocks and possible applications. *Energy Build.* **2011**, *43*, 844–852. [CrossRef]
- 10. Möller, B.; Lund, H. Conversion of individual natural gas to district heating: Geographical studies of supply costs and consequences for the Danish energy system. *Appl. Energy* **2010**, *87*, 1846–1857. [CrossRef]
- 11. Sperling, K.; Möller, B. End-use energy savings and district heating expansion in a local renewable energy system—A short-term perspective. *Appl. Energy* **2012**, *92*, 831–842. [CrossRef]
- 12. Petrović, S.; Karlsson, K. Ringkøbing-Skjern energy atlas for analysis of heat saving potentials in building stock. *Energy* **2016**, *110*, 166–177. [CrossRef]
- 13. Delmastro, C.; Mutani, G.; Corgnati, S.P. A supporting method for selecting cost-optimal energy retrofit policies for residential buildings at the urban scale. *Energy Policy* **2016**, *99*, 42–56. [CrossRef]
- 14. Mutani, G.; Delmastro, C.; Gargiulo, M.; Corgnati, S.P. Characterization of Building Thermal Energy Consumption at the Urban Scale. *Energy Procedia* **2016**, *101*, 384–391. [CrossRef]
- 15. Caputo, P.; Costa, G.; Ferrari, S. A supporting method for defining energy strategies in the building sector at urban scale. *Energy Policy* **2013**, *55*, 261–270. [CrossRef]
- 16. Monsalvete, P.; Robinson, D.; Eicker, U. Dynamic simulation methodologies for urban energy demand. *Energy Procedia* **2015**, *78*, 3360–3365. [CrossRef]
- 17. Hansen, K.; Connolly, D.; Lund, H.; Drysdale, D.; Thellufsen, J.Z. Heat Roadmap Europe: Identifying the balance between saving heat and supplying heat. *Energy* **2016**, *115*, 1663–1671. [CrossRef]
- 18. Østergaard, P.A.; Lund, H. A renewable energy system in Frederikshavn using low-temperature geothermal energy for district heating. *Appl. Energy* **2011**, *88*, 479–487. [CrossRef]
- Connolly, D.; Lund, H.; Mathiesen, B.V. Smart Energy Europe: The technical and economic impact of one potential 100% renewable energy scenario for the European Union. *Renew. Sustain. Energy Rev.* 2016, 60, 1634–1653. [CrossRef]
- Connolly, D.; Lund, H.; Mathiesen, B.V.; Werner, S.; Möller, B.; Persson, U.; Boermans, T.; Trier, D.; Østergaard, P.A.; Nielsen, S. Heat roadmap Europe: Combining district heating with heat savings to decarbonise the EU energy system. *Energy Policy* 2014, 65, 475–489. [CrossRef]
- 21. Connolly, D.; Lund, H.; Mathiesen, B.V.; Leahy, M. A review of computer tools for analysing the integration of renewable energy into various energy systems. *Appl. Energy* **2010**, *87*, 1059–1082. [CrossRef]
- 22. Advanced Heating and Cooling Plans to Determine the Effects of Greater Energy Efficiency in EU Member States, STRATEGO project, Main report. Available online: https://www.euroheat.org/our-projects/strategomulti-level-actions-enhanced-heating-cooling-plans/ (accessed on 15 April 2018).
- 23. IMEAS Project. Available online: https://www.alpine-space.eu/projects/imeas/en/home (accessed on 21 October 2020).
- 24. Ma, T.; Ostergaard, P.A.; Lund, H.; Yang, H.; Lu, L. An energy system model for Hong Kong in 2020. *Energy* **2014**, *68*, 301–310. [CrossRef]
- 25. Gupta, R.; Gregg, M. Targeting and modelling urban energy retrofits using a city-scale energy mapping approach. *J. Clean. Prod.* **2018**, *174*, 401–412. [CrossRef]
- Torabi Moghadam, S.; Lombardi, P. An interactive multi-criteria spatial decision support system for energy retrofitting of building stocks using CommunityVIZ to support urban energy planning. *Build. Environ.* 2019, 163, 106233. [CrossRef]
- 27. Delmastro, C.; Gargiulo, M. Capturing the long-term interdependencies between building thermal energy supply and demand in urban planning strategies. *Appl. Energy* **2020**, *268*, 114774. [CrossRef]

- Delmastro, C.; Martinsson, F.; Dulac, J.; Corgnati, S.P. Sustainable urban heat strategies: Perspectives from integrated district energy choices and energy conservation in buildings. Case studies in Torino and Stockholm. *Energy* 2017, 138, 1209–1220. [CrossRef]
- 29. IREN Energia. The Cogeneration and District Heating Plants in Turin. Available online: https://www.irenenergia.it/ChiSiamo/Attivita/EnergiaElettrica/ImpCogenerazione.html (accessed on 15 April 2018). (In Italian)
- 30. Østergaard, P.A. Reviewing EnergyPLAN simulations and performance indicator applications in EnergyPLAN simulations. *Appl. Energy* **2015**, *154*, 921–933. [CrossRef]
- 31. Corgnati, S.P.; Fabrizio, E.; Filippi, M.; Monetti, V. Reference buildings for cost optimal analysis: Method of definition and application. *Appl. Energy* **2013**, *102*, 983–993. [CrossRef]
- 32. Ballarini, I.; Corgnati, S.P.; Corrado, V. Use of reference buildings to assess the energy saving potentials of the residential building stock: The experience of TABULA project. *Energy Policy* **2014**, *68*, 273–284. [CrossRef]
- 33. Guelpa, E.; Mutani, G.; Todeschi, V.; Verda, V. A feasibility study on the potential expansion of the district heating network of Turin. *Energy Procedia* **2017**, *122*, 847–852. [CrossRef]
- 34. Città di Torino GEOPORTALE. 2017. Available online: http://www.comune.torino.it/geoportale/ (accessed on 4 March 2018).
- 35. Corrado, V.; Ballarini, I.; Corgnati, S.P. *Typology Approach for Building Stock National Scientific Report on the TABULA Activities in Italy*; Politecnico di Torino: Torino, Italy, 2012; Available online: https://episcope.eu/communication/download/ (accessed on 10 January 2018).
- 36. Tabula WebTool. Available online: http://webtool.building-typology.eu/#bm (accessed on 10 January 2018).
- 37. UNI 10351:2015. Materiali e Prodotti per Edilizia-Proprietà Termoigrometriche-Procedura per la Scelta dei Valori di Progetto. Available online: https://www.uni.com/ (accessed on 4 December 2017).
- 38. UNI EN ISO 6946:2008. Componenti ed Elementi per Edilizia-Resistenza Termica e Trasmittanza Termica-Metodi di Calcolo. Available online: https://www.uni.com/ (accessed on 4 December 2017).
- 39. Berkeley, L.; Ridge, O.A.K.; Ut-battelle, M.B.Y.; For, A.; Energy, S.; Or, D.; In, T.; Form, A.N.Y.; By, O.R.; Means, A.N.Y.; et al. EnergyPlus Essentials. 2019. Available online: https://energyplus.net/documentation (accessed on 14 October 2020).
- 40. UNI/TS 11300-1:2014. Prestazioni Energetiche Degli Edifici—Parte 1: Determinazione del Fabbisogno di Energia Termica Dell'edificio per la Climatizzazione Estiva ed Invernale. Available online: https://www.uni.com/ (accessed on 4 December 2017).
- Corrado, V.; Ballarini, I.; Paduos, S.; Primo, E.; Madonna, F. Application of Dynamic Numerical Simulation to Investigate the Effects of Occupant Behaviour Changes in Retrofitted Buildings. *Proc. Build. Simul.* 2017, 2017, 678–685.
- Barthelmes, V.M.; Becchio, C.; Fabi, V.; Corgnati, S.P. Occupant behaviour lifestyles and effects on building energy use: Investigation on high and low performing building features. *Energy Procedia* 2017, 140, 93–101. [CrossRef]
- 43. Kazas, G.; Fabrizio, E.; Perino, M. Energy demand profile generation with detailed time resolution at an urban district scale: A reference building approach and case study. *Appl. Energy* **2017**, *193*, 243–262. [CrossRef]
- 44. Città di Torino. Energy-environmental annex. *Regolam. Edil. Torino* 2011. Available online: http://www.comune.torino.it/ (accessed on 15 December 2017). (In Italian)
- 45. UNI EN ISO 10456:2008. Materiali e Prodotti per Edilizia-Proprietà igrometriche-Valori Tabulati di Progetto e Procedimenti per la Determinazione dei Valori Termici Dichiarati e di Progetto. Available online: https://www.uni.com/ (accessed on 4 December 2017).
- Andrić, I.; Gomes, N.; Pina, A.; Ferrão, P.; Fournier, J.; Lacarrière, B.; Le Corre, O. Modeling the long-term effect of climate change on building heat demand: Case study on a district level. *Energy Build*. 2016, 126, 77–93. [CrossRef]
- 47. Fjärrvärme, S. Kulvertkostnadskatalog. 2007. Available online: https://www.energiforetagen.se/ (accessed on 8 October 2017). (In Swedish).
- 48. IEA. Energy Technology Perspective 2016. Available online: https://www.iea.org/reports/energy-technology-perspectives-2016 (accessed on 5 October 2017).
- 49. Città di Torino. Prezzario Regione Piemonte. 2016. Available online: http://www.regione.piemonte.it/oopp/ prezzario/index16.htm (accessed on 10 November 2017).

- 50. Lund, H.; Connolly, D.; Thellufsen, J.Z.; Van Mathiesen, B.; Østergaard, P.A.; Lund, R.S.; Ridjan, I.; Hansen, K.; Drysdale, D. EnergyPLAN documentation. Available online: https://www.energyplan.eu/ (accessed on 10 September 2017).
- 51. Moncalieri Cogeneration Thermoelectric Plant—Environmental Declaration 2015. 2015. Available online: https://www.irenenergia.it/ (accessed on 15 February 2018). (In Italian)
- 52. Torino Nord Cogeneration Thermoelectric Plant—Environmental Declaration 2015. 2015. Available online: https://www.irenenergia.it/ (accessed on 15 February 2018). (In Italian)
- 53. Rapporto Sulle Aste di Quote Europee di Emissione 2017. 2017. Available online: https://www.gse.it/ (accessed on 19 January 2018).
- 54. Ministero dello Sviluppo Economico; Ministero dell'Ambiente e della Tutela del Territorio e del Mare; Ministero delle Infrastrutture e dei Trasporti. Integrated National Energy and Climate Plan. 2019. Available online: https://www.mise.gov.it/ (accessed on 12 October 2020).
- 55. Corrado, V.; Ballarini, I.; Paduos, S.; Fabrizio, E.; Filippi, M.; Monetti, V. *Development of the Cost-Optimal Comparative Methodology According to Directive 2010/31/UE*; Report RdS/2013/144; ENEA: Roma, Italy, 2013. (In Italian)
- 56. Cabrera, P.; Lund, H.; Thellufsen, J.Z.; Sorknæs, P. The MATLAB Toolbox for EnergyPLAN: A tool to extend energy planning studies. *Sci. Comput. Program.* **2020**, *191*, 102405. [CrossRef]
- 57. Guelpa, E.; Barbero, G.; Sciacovelli, A.; Verda, V. Peak-shaving in district heating systems through optimal management of the thermal request of buildings. *Energy* **2017**, *137*, 706–714. [CrossRef]

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





Article New Energy Policy Directions in the European Union Developing the Concept of Smart Cities

Adrian Tantau¹ and Ana-Maria Iulia Şanta^{2,*}

- ¹ UNESCO Department for Business Administration, The Bucharest University of Economic Studies, 010374 Bucharest, Romania; adrian.tantau@fabiz.ase.ro
- ² Department of Economics and Economic Policies, The Bucharest University of Economic Studies, 010374 Bucharest, Romania
- * Correspondence: anamaria.santa@economie.ase.ro

Abstract: In the context of the European Union promoting clean energy, sustainability and better living conditions for its citizens, the development of smarts cities is an initiative supported at the European Union level, in line with the new energy policies of the European Union promoted by the package "Clean Energy for All Europeans". The concept of smart cities gains increasing importance in the European Union, a fact that is reflected in the project "European Innovation Partnership on Smart Cities and Communities" of the European Commission. Smart cities are a practical example of how the new energy policies shape the lives of the European Union citizens, trying to improve it. As a consequence, new business models arise in big cities, involving the use of technology for better living conditions. These new, technology-based business models are important, as they improve the life quality of the inhabitants, they reduce the climate change impact, and they contribute as well to job creation in the IT-industry, promoting innovation. They have as well a social impact, as they bring experts from energy policies, business, economics, legal and IT together in order to project a new type of city—the smart city. The research hypothesis of the present article is that there is a high acceptance towards the concept of smart cities at the European Union level and that this concept could be implemented with the help of information technology and of artificial intelligence. This way, legal provisions, economic measures and IT-tools work together in order to create synergy effects for better life quality of the citizens of the European Union. The research hypothesis is analyzed by means of the questionnaire as a qualitative research method and is as well assessed by using case studies (e.g., Austria, Finland, Romania). The novelty of the case studies is that the development of smart cities is analyzed due to the new trend towards sustainability in two countries with different living conditions in the European Union.

Keywords: new energy policies in the European Union; smart cities; sustainability; clean energy; innovation; information technology; artificial intelligence

1. Introduction

Smart cities are a concept of great importance due to their dynamics in our current society, a concept aiming to enhance sustainability. Cities tend to develop, and at the same time, the standard of living of their inhabitants develops, so new solutions must be found in order to harmonize these two trends. More than 55% of our planet's population live in big cities, and the trend is ascending, according to the United Nations [1]

Europe will concentrate on a high degree of this urban development so that cities in Europe face new challenges related to the ascending trend of urbanization. Such challenges are related to modern buildings, to a modern, information technology-based public transportation system and to the increased use of clean energy in order to reduce the negative impact of urbanization on the environment. Big cities are important energy consumers, as 80% of the energy consumption in the European Union will be registered in big cities and their surroundings [2]. A consequence will be a high concentration of CO_2 emissions in



Citation: Tantau, A.; Şanta, A.-M.I. New Energy Policy Directions in the European Union Developing the Concept of Smart Cities. *Smart Cities* **2021**, *4*, 241–252. https:// doi.org/10.3390/smartcities4010015

Academic Editor: Isam Shahrour

Received: 30 November 2020 Accepted: 6 February 2021 Published: 9 February 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). this area and thus a negative impact on the environment if new innovative solutions would not be put in place. In this context, it is important to shape energy policies at the European Union level, aiming for sustainability in big cities and promoting the smooth transition from big cities to sustainable smart cities. New energy policies are drafted at the present moment with a special focus on sustainability. They need as well to be operationalized to find a concrete expression and new projects in society implementing these new policy measures. Smart cities may be a concrete expression of the new directions of energy policies at the European Union in line with the principles of sustainability, a concrete expression of sustainability that was not researched enough until now, especially with regard to its social impact.

The aim of this paper is to assess new energy policy directions promoting smart cities and sustainability in urban areas, evaluating as well the acceptance towards these new developments and new tools in order to implement the business models in smarts cities, such as artificial intelligence-based instruments. The research hypothesis of the present article is that there is a high acceptance towards the concept of smart cities at the European Union level. The evaluation of new energy policies in the European Union is important and needed, as the European Union has a very ambitious and visionary goal of transforming the European energy system into one of the most sustainable in the world.

In order to accomplish this ambitious policy goal, energy savings are needed, and energy performance must be increased. Such developments are supported by new IT&C technologies (used for smart metering and smart grid), by the Internet of things (IoT), by artificial intelligence and by raising the awareness at the level of the consumers and thus shaping new consumer behavior in this direction. Increasing the efficiency of buildings and raising awareness of consumers towards this issue is important. A document establishing such concepts and developments and thus relevant for the assessment is the Energy Performance of Buildings Directive (EPBD). It is in line with the initiative of the European Commission to increase their energy efficiency as part of its new energy policy [3]. Buildings are important energy consumers in cities of the European Union, and the building sector is responsible for 40% of the total energy consumption and for 36% of CO2 emissions [4]. Given these circumstances, the EPBD Directive expresses the need to find solutions for the construction of zero energy buildings by 2020 [4]. All buildings that are built should be zero energy with no carbon emissions in order to achieve the ambitious policy goal mentioned above.

These policy measures should be supported by innovative technologies (such as IoT and artificial intelligence), which are described in the Strategic Energy Technology Plan (SET Plan), making it possible to ensure a transition to a low carbon economy by 2050 in the European Union. This objective was mentioned in 2011 in a document of the European Commission, namely in the Roadmap for moving to a competitive low-carbon economy by 2050. The documents mention as well the concept of smarts cities, reflected in the Smart Cities and Communities Initiative of the European Commission. The aim of this initiative is to ensure cooperation between stakeholders for reaching an improved standard of living in smart cities in the European Union and thus improving the life quality of citizens in the European Union [5]. Energy-efficient cities are part of the Europe Strategy 20–20–20, and their development can be ensured by using IT&C technologies and artificial intelligence supporting new business models for these cities and their inhabitants. In the context of digitalization and of the increased use of information technology in order to improve our lives, artificial intelligence is a good tool to be used for improving smart cities if there is acceptance of this new instrument. The use of artificial intelligence may encourage the development of new product markets in smart cities, such as the market for smart homes or the market for smart vehicles, an issue which is assessed in the present research. All these new aspects related to new markets and products are a consequence of implementing the concept of smart cities in the European Union.

The methodology of the present research is based on a qualitative research approach, with an international, comparative dimension. Case studies, the questionnaire used as a

The present article has the following structure for approaching the research topic: Introduction, The current state of research (Literature review), Materials and Methods, Results, Discussions, Conclusions and References.

2. The Current State of Research (Literature Review)

2.1. Smart Cities—Using the Benefits of Technology

According to the studied literature, a smart city is a place where the traditional networks and services are made more efficient by using digital and telecommunication technologies, for the benefit of its inhabitants and businesses [6]. A smart city is thus a sustainable and efficient urban center that provides a high quality of life to its inhabitants through optimal management of its resources.

According to the literature, information technology (IT) comprises "means and methods of collecting, processing and transferring data" [7].

2.2. Smart Cities and Sustainability

Smart cities are a concrete expression of implementing sustainability. Sustainability is a concept that does not have a unitary approach in the economic literature. Despite the difficulty of finding a unitary definition, it was clear that sustainability and sustainable development are important goals for international organizations (United Nations, 2015). Even with no universal definition of sustainability [8], it is clear and commonly accepted that this concept addresses human needs from a broad perspective [9]. Sustainable development can be understood in its environmental dimension, in its economic dimension and in its social dimension [10]. Given the challenges of climate change, the environmental dimension is currently more often analyzed. Nevertheless, the social dimension of sustainability is as well of great importance due to its impact on society [9].

The present research analyses smart cities in the context of new, sustainable energy policies, building a bridge between previous literature and the newest developments at the European policy level. It considers the consumer focus in the transition towards new energy policies [11], aspects that have an impact in improving smart cities. The new sustainability-based energy policy is determined by consumer preferences [12]. Given the current trends of analysis of circular economy [13] and of its impact on our society, the present article is in line with previous research, further developing it and adding a new perspective of the European policies. The research has a multidisciplinary dimension that is appropriate to the research question, as Economics belongs to the field of econosciences [14], which allows this kind of approach.

3. Materials and Methods

The present article uses an interdisciplinary research approach, assessing aspects of business, economics, legal provisions and social impact in order to provide a comparative perspective with an international dimension on the researched topic. The research hypothesis of the present paper defines smart cities as a visionary concept supporting sustainability as a mindset of the future citizens.

The relevant literature was consulted with regard to the researched topic. The definitions of the relevant concepts and their importance were summarized in the literature review.

The questionnaire is used as a qualitative research method. The questionnaire contains five sections summing up 21 semi-structured questions referring to the topic of new energy policies in the European Union and their application in order to achieve sustainability in its economic, social and environmental dimensions. The questionnaire contains questions analyzing the acceptance of these new values within the citizens of the European Union. The research hypothesis of the present article, which is the research problem if there is a high acceptance towards the concept of smart cities at the European Union level and if that this concept could be implemented with the help of information technology and of artificial intelligence, is being assessed in the results of the questionnaire. The type of questions corresponds to the semi-structured interview. The questions offer the respondent the opportunity to add some ideas or express his/her opinion if this is not contained in the response alternative in the questionnaire. As an example of a question, the respondents are asked about how they assess the orientation policy of smart cities towards clean energy, energy efficiency, smart vehicles or consumer protection. The research is based as well on questions assessing in detail the acceptance of specific elements, which are the basic aspects of smart cities. This acceptance is detailed in separate questions in the questionnaire about the importance of the reduction of carbon emissions, the importance of energy efficiency, of energy savings and recycling (and thus promoting a circular economy), to the importance of environment protection, to the importance of Clean Energy and renewable energy.

The type of analysis used in the present research for assessing the interviews is the thematic analysis. The answers of the respondents were grouped by means of the thematic analysis in order to have a better overview of the expressed ideas and to better follow the reflection of the research hypothesis in the answers of the respondents. This research design ensures an optimal valuation of the expertise of the respondents. Ideas regarding the acceptance of smart cities in the European Union, the need to focus on energy savings and circular economy, the need to improve life quality, the importance of the use of technology and the need to improve mobility were assessed in a comparative manner. Evidence from the interviews supports the formulated research hypothesis. The research was conducted in the period February 2018–February 2019, assessing the answers of 138 respondents. The respondents to the questions are representatives from companies acting in the energy sector in the European Union (from countries like Romania, Austria, Germany, the United Kingdom, Italy, Portugal, Spain, Greece, Belgium), university professors from the European universities from Romania, Germany, Poland and Bulgaria, with a high academic qualification. The experience and qualifications of the respondents were used in a qualitative research approach. The criteria for selecting the respondents were their experience in fields that are relevant for smart cities, such as the field of energy or new developments in the energy sector. The respondents are active as university staff involved in teaching subjects related to energy, or they are representatives of companies in this field, acting at the top management level.

Using qualitative methods in economic research is an aspect of novelty and of originality, corresponding to the newest trends in academic research in the field of business and economics. Qualitative research empowers the researcher at gives a high role to the research participants in being part of the research project. This is a motivation for the research participants, e.g., the respondents, to participate in the research project, as they are aware that their qualification and expertise becomes useful for research. This is the reason why special importance was given to this research method in the present article.

Another research method used in the present article is represented by case studies from Austria and Romania presented in a comparative manner. Vienna was chosen as a case study, as it offers a very high standard of living, a very good quality of life according to international rankings. For illustrating current developments in Romania in this field, the city of Cluj-Napoca is presented as a case study. Smart cities find a great acceptance in Romania and are an opportunity to improve the life quality of inhabitants, a trend that is appreciated in Romania and which offers as well the opportunity to innovate for entrepreneurs and firms (startups) wanting to support this modern and sustainable trend.

Indicators provided by the European Commission are analyzed in a complementary manner, in a mixed-methods approach. Such as indicator is the circular economy Indicator, showing to what extent the mindset of people in the European Union changed in this field and how the awareness regarding the importance of sustainability resulted in societal behavior.

4. Results

Sustainable business models are a key concept for creating smart cities. Such cities should offer a high quality of life by using IT&C based solutions in smart cities. Based on advanced algorithm and control techniques, Computational Intelligence and machine learning will enable us to think about new optimization levels for the energy systems in a smart city. The new communication technologies and the related communication networks and connectivity must be followed by a new cyber-security design for energy systems in a smart city [15]. Supervisory control and data acquisitions (SCADA) systems will be confronted with new high-level processes for the supervisory management. New security policies for smart cities will as well include the concept of a safe city [16]). The two pillars of improving smart cities are represented in Figure 1.



Figure 1. Pillars for improving smart cities. Source: own representation based on own research, including [15].

The research hypothesis of the present paper defining smart cities as a visionary concept supporting sustainability as a mindset of the future citizens is validated by the research results. Both pillars have similar importance in the vision of the respondents, as 55% of the respondents indicated improved life quality as the main element of smart cities strategy, while 45% of the respondents referred to the importance of innovative technologies in supporting smart cities. Especially people working in the private sector with age under 40 years found the importance of technology very high. Among the respondents, IT-workers were as well interviewed (about 3% of the interviewed respondents), and they found artificial intelligence of very high importance in the development of smart cities. An important research result is that respondents in the IT-sector showed great awareness towards the importance of clean energy and of new energy policies.

Improved life quality becomes a mindset of the European Union citizens. They look for the proper instruments to implement sustainability-based energy policies, which become concrete in the concept of smart cities.

At the level of the European Union, Eurostat presented the Urban Audit for EU countries as a monitoring framework for sustainability aspects. This tool is based on Eurostat data from yearbooks, and it is compiled with the help of national statistics offices. This tool is appropriate for assessing smart cities in the European Union, as it is a Eurostat tool, and therefore, we are not confronted with the problem of the lack of comparable data or of a unitary methodology any more [9]).

Vienna, the capital of Austria, accomplishes the goals of smarts cities, as a good quality of life to its inhabitants, achieving as well to save resources and to use modern and innovative technologies. Vienna defines as goals until 2050 the efficient use of energy, good energy performance for its buildings, good mobility achieved with effective use of resources [17]. The attributes of the smart city Vienna are represented in Figure 2: Attributes of the smart city Vienna.



Figure 2. Attributes of the smart city Vienna. Source: own research based on [17].

Life quality is a main goal in the long-run in developing smart cities, as is obvious in the case study of Vienna. This goal may be supported by technology, which is a tool in implementing the new directions of sustainable energy policies. Given this fact and the importance of technologies in implementing the concept of smart cities, new business models related to this concept may be implemented by means of information technology and of artificial intelligence. 55% of the respondents to the questionnaire found life quality as the main goal to be achieved in smart cities. 30% of the respondents found technology very important, while 15% were preoccupied with energy savings in their environment. The city of Vienna was selected for this case study, as it is recognized for urban development, with experts researching this field [18].

Another example of a smart city in Austria in Graz, where energy efficiency is an important aspect of improving life quality in this city [19]. "smart city Project Graz" is a project implementing this concept [20].

In Romania, the trend of promoting smart cities finds acceptance. A city with an impressive development in this direction is Cluj-Napoca. It has built a public transportation system based on clean vehicles, such as electric buses, and is supported by technology in order to provide information to the inhabitants of the city. It tries as well to reduce noise pollution and to provide smart parking for its inhabitants. Its main features are represented in Figure 3.



Figure 3. Development of smart cities in Romania: Cluj-Napoca. Source: own research based on [21].

Another city in Romania trying to become a smart city is Oradea, which implements criteria like building a Smart Community based on smart infrastructure and smart mobility, smart utilities, smart environment, smart living and as well smart government in order to support the digitalization of the local administration [22].

The present research analyzes the opinion of experts from various countries related to the topic of smart cities. There are specific semi-structured questions in the questionnaire aiming to investigate the acceptance of the respondents towards the new values of sustainable energy policies, which are implemented and get concrete expression in the development of smart cities. Such values are related to the acceptance of clean energy, of defining life quality depending on sustainable development aspects, the willingness to implement these new concepts in daily life and the need to have new energy policies considering the new trends of sustainability. The assessment of the questionnaire results pointed out a high degree of acceptance towards promoting smarts cities in the context of new energy policies of the European Union. The acceptance towards the values promoted as goals of smart cities among the interview participants is illustrated in Figure 4.





An important difference between Austria and Romania related to the smart city strategy is given by the fact that Austria has a well-established, long-term strategy for smart cities and establishes its goal for the smart city strategy until 2050, while in Romania, there is not yet such a long vision regarding this aspect. The idea of smart cities is only at the level of pilot projects in Romania, and there is no clear framework yet, how this field would develop.

The results of the own conducted research by using the questionnaire revealed the fact that more than 50% (53%) of the respondents (74 respondents of 138) believe that clean energy is a key-direction of new energy policies.

One of the main goals of promoting new energy policies is related to consumer protection; however, as the research results indicate that 29% of respondents find this goal important.

Energy efficiency is an important energy policy goal at the EU level. Unfortunately, this goal is not very clear or not so important for the majority of the consumers. Therefore, only 10% of respondents consider that energy efficiency must be a priority for sustainable development. Another curious result is that only 7% of respondents find smart vehicles as new instruments to promote new energy policies. It must be mentioned that the results are significant influenced by the different living conditions in different countries in the EU (e.g., Austria and Romania).

An important index that might reflect societal behavior in the field of energy policies is the circular economy indicator, which is provided by the European Commission. Figure 5 is presenting a comparative view among the values of this indicator in the European Union Member States:





This indicator shows societal behavior related to the circular economy, meaning if there is a common practice related to the circular economy in order to protect the environment like we have the case in Austria, or it shows if there is a lack in such behavior like we have the situation in Romania for example. In such a situation, an awareness raining campaign is needed in order to draw attention to the importance of this topic.

A more detailed comparative perspective (at the global level) related to the electronic media, which is an important sector in the European Union, is provided by the European Commission in Figure 6.



Figure 6. Circular economy in electronic mass media. Source: [24]

Smart cities manage to gather the best ideas and innovation in order to protect both consumers, which are inhabitants of smart cities and the environment. A sustainable approach is needed in the context of providing better health to the inhabitants of big cities. Such aspects are as well a result of a shift in cultural values. People share values related to sustainability and to sustainable approaches, taking into account future generations. Improving smart cities is a topic bringing together the best experts in business, economics, in the legal field, in technology and in social research for evaluating the impact of these cities in society.

5. Discussions

Discussions related to the research topic might refer to the fact that there is no unitary definition of smart cities at the European Union level, meaning there is not only one acceptance of this concept. Being a new concept, it is still very dynamic and fluid, and it is being shaped according to the latest developments at the international level. Given the lack of a unitary, unique definition of the concept of smart cities, it is difficult to find a harmonized approach, which is valid in each and every country.

5.1. A Comparison of the Findings with Previous Research

Application programming interfaces (APIs) for smart cities can be developed in order to implement digitalization in cities for the benefit of the consumers. Such applications have already been used in smart cities in Finland. More than that, in Finland, the approach is to have harmonized applications for smart cities. This ensures comparable data support that can be used for improving the life quality of the citizens in smart cities [25]. The best practices in Finland in cities such as Helsinki, but as well in Tampere, Espoo and Turku are a good example of using technology in the best interest of the consumer, of the citizens [26]. The impact of technological development is thus revealed not only in the technology sector but as well in the whole society. Positive externalities can thus be achieved, with important social benefits and with social value.

5.2. Theoretical and Practical Implications of the Research

At the same time, a common, harmonized basis could be established if this concept of applications for smart cities find acceptance. It could be discussed if best practices showing the advantages of these technology tools might be used as a source of inspiration for some other smart cities in the European Union, needing guidance in order to implement the concept of smart cities. The City of Barcelona in Spain is another example of a smart city using applications for smart cities in order to provide data for companies and municipalities or other institutions, making better cooperation between these actors possible and promoting the use of digitalization for improving life quality [17,27]. (Projects at the European Union level are developed in order to implement the concept of application programming interfaces for smart cities. Examples of such projects developed at the European Union level are the projects FI-WARE, ICOS (Intelligent/Smart cities Open-source community) and smart objects for intelligent applications (SOFIA) [17,27]. The concept has a real success as well beyond Europe, for example, in the United States of America, with New York as a best-practice example [17]. These positive experiences proved the fact that it is a successful concept, and countries such as Romania, which intend to develop the concept of smart cities at their level, may use such positive experiences as best practice.

5.3. Limitations of the Research

Possible limitations of the research could be given by the fact that the research is performed on a qualitative basis, not a quantitative one. Another limitation is the fact that the case-studies-approach covers only specific cases; it cannot be expanded on each and every city. Each city has a different practical situation and different policies or criteria for defining smart cities.

Another possible limitation of research could be the lack of comparable data, as the field of smart cities is new, and there it is difficult to find harmonized, comparable data in this field.

5.4. Conclusions and Future Perspective of the Research

Another topic of discussion might is if it is measurable if a city is a smart city or not if it accomplishes the status of a smart city if there are any objective, measurable criteria for establishing if a city is a smart city or not and if there is general acceptance at the European Union level or at a global level regarding the assessment of smart cities and of cities aiming to become smart cities.

The discussion might be developed as well around the idea that not only it lacks a common, unitary definition of smart cities worldwide or at the European Union level, there is, in fact, no unitary definition of sustainability worldwide. Given the complexity of this issue, discussions may clarify what is, in fact, the sustainability of smart cities, how it can be evaluated to what extent a smart city is sustainable, to what extent it contributes to a sustainable environment if there are any measurable criteria or indicators to answer these questions and as well if there is acceptance towards the use of some indicators in this field.

6. Conclusions

The research findings of the present article validate the research hypothesis that smart cities are a visionary concept supporting sustainability as a mindset of the future citizens. New business models related to this concept might be implemented by means of information technology and of artificial intelligence.

These tools may contribute to a better future for all the inhabitants of smart cities and might have a social impact in improving the quality of life of people living in smart cities. The gap between technology-based approaches and human-focused approaches might this way disappear, as the two visions can find solutions towards better living conditions. The use of information technology and artificial intelligence for smart cities has a very high potential, given the development of the IT sector worldwide.

The novelty and originality of the present research are given by the interdisciplinary research method used in order to evaluate new perspectives related to smart cities in the European Union, aiming to achieve sustainability and better life quality for citizens. The methods used have a contribution to research, as they use a comparative perspective with an international dimension, using the experience of the respondents in a qualitative manner. Case studies show best practice examples for the researched concept that can be used by new Member States such as Romania, having less experience with smart cities, but wanting to implement this concept.

The results of the present research can be further developed in future research projects. The findings of the present research will have an impact on future research articles, as the acceptance towards smart cities needs to be assessed before implementing measures related to the support and financing of smart cities. The acceptance of citizens towards these new measures needs as well to be assessed. Such additional research will for sure be needed in the future, given the impact novelty of the subject and the dynamics of smart cities in the European Union, a project aiming at better living conditions for people living in these cities. Given this context, it is a good opportunity to ensure collaboration between research and practice. This way, a strong impact of research beyond academia is achieved, an impact in the society contributing to improving the values and the standards of society and creating a new mindset in accordance with new, modern visions based on sustainability.

Author Contributions: Conceptualization, A.-M.I.Ş.; Data curation, A.T. and A.-M.I.Ş.; Formal analysis, A.T. and A.-M.I.Ş.; Investigation, A.T. and A.-M.I.Ş.; Methodology, A.T. and A.-M.I.Ş.; Supervision, A.T.; Validation, A.T.; Writing—original draft, A.-M.I.Ş.; Review of original draft A.T.; Writing—review & editing A.-M.I.Ş. and A.T. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. United Nations. The 2030 Agenda for Sustainable Development. 2015. Available online: https://sustainabledevelopment.un. org/hlpf/2018 (accessed on 31 December 2018).
- EEA. Urban Environment, European Environment Agency. 2017. Available online: http://www.eea.europa.eu/themes/urban/ intro (accessed on 20 January 2017).
- 3. EC. Energy and Smart Cities. 2018. Available online: https://ec.europa.eu/energy/en/topics/technology-and-innovation/ energy-and-smart-cities (accessed on 10 December 2018).
- 4. EC. Buildings. 2018. Available online: https://ec.europa.eu/energy/en/topics/energy-efficiency/buildings (accessed on 10 December 2018).
- 5. European Commission. Towards an Integrated Strategic Energy Technology (SET) Plan: Accelerating the European Energy System Transformation. In *Communication from the Commission C;* European Commission: Brussels, Belgium, 2015; 6317 final.
- 6. Bakıcı, T.; Almirall, E.; Wareham, J. A Smart City Initiative: The Case of Barcelona. J. Knowl. Econ. 2012, 4, 135–148. [CrossRef]
- 7. Odintsova, S.A.; Kenesova, N.T.; Sarsekeyeva, Z.E. Information Technology: Definition, Essence and content of the concept. *Educ. Sci. Bord.* **2013**, *4*, 107–109.
- 8. Turcu, C. Re-thinking sustainability indicators: Local perspectives of urban sustainability. *J. Environ. Plan. Manag.* 2013, 56, 695–719. [CrossRef]
- 9. Verma, P.; Raghubanshi, A.S. Urban sustainability indicators: Challenges and opportunities. *Ecol. Indic.* 2018, 93, 282–291. [CrossRef]
- Escribano, M.; Díaz-Caro, C.; Mesias, F.J. A participative approach to develop sustainability indicators for dehesa agroforestry farms. *Sci. Total Environ.* 2018, 640, 89–97. [CrossRef] [PubMed]
- 11. Pelau, C.; Pop, N.A. Implications for the energy policy derived from the relation between the cultural dimensions of Hofstede's model and the consumption of renewable energies. *Energy Policy* **2018**, *118*, 160–168. [CrossRef]
- 12. Tanţău, A.D.; Nichifor, M.A. Consumer Behavior and Perception on the Renewable Energy Instalments. Case Study: Wind Energy in Romania and the Netherlands. In Proceedings of the 6th International Symposium on Environmental Management and Material Flow Management, Bor, Serbia, 2–4 October 2016; Available online: www.emfm.tf.bor.ac.rs (accessed on 10 January 2019).
- 13. Tantau, A.D.; Maassen, M.A.; Fratila, L. Models for Analyzing the Dependencies between Indicators for a Circular Economy in the European Union. *Sustainability* **2018**, *10*, 2141. [CrossRef]
- 14. Dinu, V.; Săvoiu, G.H.; Dabija, D.-C. *A Concepe, a Redacta și a Publica un Articol Științific: O Abordare în Contextul Cercetării Economice;* Translation: How to Write and Publish a Research Article; Editura ASE: Bucharest, Romania, 2016.
- 15. O'Dwyer, E.; Pan, I.; Acha, S.; Shah, N. Smart energy systems for sustainable smart cities: Current developments, trends and future directions. *Appl. Energy* **2019**, 237, 581–597. [CrossRef]
- 16. Lacinak, M.; Ristvej, J. Smart city, safety and security. Procedia Eng. 2017, 192, 522–527. [CrossRef]
- 17. Stadt Wien. 2019. Available online: https://smartcity.wien.gv.at/site/buergerinnen/ (accessed on 18 January 2019).
- 18. BBVA API_Market. 2021. Available online: www.bbvaapimarket.com/en/api-world/how-apis-are-powering-smart-cities/ (accessed on 28 January 2021).
- 19. Smart City Graz. 2021. Available online: www.smartcitygraz.at (accessed on 30 January 2021).
- 20. Hammerl, B. Smart City Labs als Möglichkeitsraum für technologische und soziale Innovationen zur Steigerung der Lebensqualität in Städten. In Proceedings of the REAL CORP 2014 Tagungsband, Vienna, Austria, 21–23 May 2014; Available online: http://www.corp.at (accessed on 14 January 2021).
- 21. Smart City Romania Association. Available online: http://romaniansmartcity.ro/2017/07/26/cluj-napoca-un-viitor-smart-city/ (accessed on 18 January 2019).
- 22. Oradea. O Strategie Pentru un Oraș Inteligent: Oradea Smart City. (Translation: Oradea—A Strategy for a Smart City). 2021. Available online: www.oradea.ro/stiri-oradea/o-strategie-pentru-un-oras-inteligent-oradea-smart-city (accessed on 29 December 2020).
- 23. EC. Societal Behaviours | Eco-Innovation Action Plan (europa.eu). 2020. Available online: https://ec.europa.eu/environment/ ecoap/indicators/societal-behaviours_en (accessed on 17 November 2020).
- 24. EC (2021). Available online: https://www.switchtogreen.eu/?p=1166 (accessed on 17 January 2021).

- 25. Data Business Finland. Fast-Track to Getting Started with Open Smart City APIs. 2021. Available online: www.databusiness.fi/ en/News/fast-track-to-getting-started-with-open-smart-city-apis/DataBusiness (accessed on 27 January 2021).
- 26. Data Business Finland. 2021. Available online: www.databusiness.fi/content/uploads/2017/10/20171109_HarmonisedSmartCityAPIs_WEB.pdf (accessed on 27 January 2021).
- 27. European Data Portal. Open Smart City APIs, Portalul European De Date (europeandataportal.eu). 2021. Available online: www.europeandataportal.eu/ro/news/open-smart-city-apis (accessed on 27 January 2021).





Article Economic Framework of Smart and Integrated Urban Water Systems

Neil Grigg



Citation: Grigg, N. Economic Framework of Smart and Integrated Urban Water Systems. *Smart Cities* 2022, *5*, 241–250. https://doi.org/ 10.3390/smartcities5010015

Academic Editors: Isam Shahrour, Fateh BELAID and Véronique FLAMBARD

Received: 28 January 2022 Accepted: 28 February 2022 Published: 2 March 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Department of Civil and Environmental Engineering, Colorado State University, Fort Collins, CO 80524, USA; neilg@engr.colostate.edu

Abstract: Smart and integrated urban water systems have important roles in advancing smart cities, but their contributions go much further by supplying needed public services and connecting other sectors to meet sustainability goals. Achieving integration and gaining access to financing are obstacles to implementing smart water systems and both are implicit in the economic framework of smart cities. Problems in financing the start-up of smart water systems are reported often. The local and diverse nature of water systems is another barrier because an approach that works in one place may not work in another with different conditions. The paper identifies the challenges posed by the economic framework and provides examples from four cities with diverse characteristics. It outlines pathways to advance implementation of smart water systems by improving control strategies, advancing instrumentation and control technologies, and most of all, to help transform cities by raising customer awareness and trust through reliable and useful water information.

Keywords: urban water; smart systems; automation; integrated management

1. Introduction

Smart cities have many interrelated parts, and their water systems provide essential functions by supplying needed public services and connecting other sectors with water to aid them in meeting economic, social, and ecological goals. To play these roles effectively, urban water systems should be integrated and take on smart attributes [1]. However, while technological aspects of smart urban water systems are impressive, the systems face challenges to achieve integration, employ smart technologies effectively, and sustain adequate financing sources [2].

Achieving integration and gaining access to adequate financing are principal obstacles to advancing the state of the art of smart urban water systems. While they provide multiple types of services to diverse stakeholders and are vital to cities, they confront financial challenges caused by their multiplexed economic framework. Managing these challenges is complex in cities with decentralized power structures, especially considering issues of affordability and financial limitations.

The paper describes the economic framework of smart water systems and smart cities, and it explains the resulting incentives and controls that it imposes on them. It also identifies challenges the systems face, and it outlines future directions to help advance the implementation of smart water systems in smart cities. The future directions go along three lines. One is improvement in control technologies, which is the main topic that is discussed by researchers studying smart water systems. A second is the advancement of instrumentation and controls, which is driven mainly by commercial incentives. Most significant, however, is the third line, which is about how customers use smart water information to help transform cities.

Several brief case studies of smart urban water systems and integrated approaches are summarized as examples of current situations. These were selected from a growing inventory of case studies that have been made available through recent studies of the International Water Resources Association [2]. The selected cases illustrate a leading-edge city (Singapore), a new smart city in Korea (Paju), a growing medium city in the western United States (Fort Collins), and a large city in Mexico with a very limited water supply (Juarez). While the case citations are brief, they range across the major issues along the three lines of discussion, advancing technologies, new control methods, and customer interactions in urban areas.

Interest in smart water systems is advancing rapidly, but their implementation and success will be shaped more by economic and social forces than by advances in technology. The literature about smart water systems mainly focuses on examples of new technologies, and this paper aims to focus more directly on the economic forces that must be confronted by system managers to facilitate improved urban water services.

2. Urban Water Systems

Urban water systems are conglomerates of the infrastructures and operating controls of water supply, wastewater, stormwater, and recycled water systems [3]. They serve the social, constructed, and natural subsystems of living cities and they connect elements of these subsystems through interdependences [4]. For example, water supply connects to public safety by providing standby fire protection, stormwater systems can add to open space and recreational opportunities, and urban water systems draw supplies from ecosystems and can in turn nourish them.

This general perspective explains how water interacts with other subsystems, but a city is a system of systems, and it requires more information to explain how urban water systems work and are controlled. A concept for the interdependences among water supply, wastewater, and reclaimed elements is shown in Figure 1, and stormwater systems can be shown to cut across these as they involve entire cities. Stormwater harvesting can be added to the diagram, as well as the biosolids management that results from wastewater treatment.



Figure 1. The urban water system.

The urban water system works with raw water diverted and conveyed to treatment, distribution, and to users. Wastewater is collected, treated, and discharged to the receiving stream. Biosolids (or sludge) are dewatered, digested, and disposed of. The stormwater system traverses the city to convey diffused waters, some of which end up in the collection system. Water can be recycled from the wastewater or stormwater system, and raw water can be used directly for purposes such as fire suppression or landscape irrigation.

Integration of urban water systems requires cooperation among the management structures for the separate services. This can be formal, as in placing all services in a single organizational unit, or it can be informal, with cooperation occurring on an ad hoc basis. Formal integration requires a comprehensive approach that includes diversifying sources, protecting water at its source, and integrating operations for water storage, distribution, treatment, recycling, and disposal. This requires that the systems are operated as one utility to facilitate efficiencies like the recycling of wastewater or reducing water footprints. Informal integration is much weaker and depends on incentives as well as opportunities for joint work [5].

As urban water systems are nested within the economic, social, and natural systems of cities there are several facets of integration. One integrates the services, such as when the separate services are operated as one utility. Another facet of integration is between the utilities and the external water environment of the streams and aquifers in and around the city. The urban water system impacts the interflows between ground and surface water and the heat island effects in cities, as well as flood plain health and urban ecology. A test for the effectiveness of integration is whether it enables the city to reach net-zero impact status in providing water, preventing pollution, and sustaining urban ecology.

3. Why Integration Is Needed

Integration among the separate urban water subsystems is a logical partner to smart systems. Application of controls to the separate water systems has been evolving for decades, but without integration, the efforts will fall short of addressing the major issues that cities face. This reality is widely recognized in the water industry and among urban leaders, and concepts such as One Water are emerging to develop appropriate strategies [6–8]. As an example, one of the cases cited here, the city of Singapore, faces the need to develop a total One Water approach if it is to meet its water needs in the future. Another of the cases, Juarez, which also faces a severely limited water supply, is addressing its smart water system with a different approach.

Integrated urban water resources management through the One Water approach has the potential to affect the shape and functionality of cities [9]. If the One Water approach is to achieve its goals, it must provide customers with integrated information to avoid confusion. For example, customers would not tolerate one message about water use and another about wastewater, especially when they are confronted with so many new categories of information in today's world. If integration can give people a sense of living in a total water environment, they will be more likely to accept a bill for services that pays for all systems involved. They should be informed how smart systems are improving efficiencies through combinations of services with multiple use facilities, for example, detention ponds and stormwater harvesting. Integration can improve information flow in different ways and, if handled effectively, has the potential to improve customer trust in utilities. Another important advantage of integrated approaches is to protect the urban water ecology, which can be built into the combined system to minimize water withdrawals and protect water quality and species [10,11].

4. Smart Water Systems in Smart Cities

A smart urban water system would join the constellation of other smart systems in a smart city, which use computer controls and information to collect data, use that data to improve operations, and communicate with citizens about all aspects of their lives in the cities [12]. In a smart city, a smart integrated urban water system would feature infrastructure controls, collection of system and user information, application of the information to control water operations and to inform citizens, and it would be available for emergency management.

The concept is illustrated in Figure 2, which shows a system to be controlled and actuators to implement controls based on decisions informed by models and data. The data collection function informs system operators and users as well. Data from the system can be used for operations and troubleshooting. Emergency information can be sent to the actuators.

The concept has evolved from the early supervisory control and data acquisition (SCADA) and control systems to today's smart concept [13]. More specifically, prior to the evolution of smart systems, the concept would have not included the transmission of data shown by the dotted lines stemming from the general flow of information and control



commands. So, the ability of the smart system to provide the information for operations, customers, and emergency management comprises the most promising new feature.

Figure 2. Systems control and information interfaces.

These features of smart water systems have evolved from the past when informationbased controls started with the advent of computers. This led to the development of the SCADA systems, and later sophisticated databases and geographical information systems were added to create technology-based platforms for system management. When personal digital assistants became available around the turn of the century, they were widely used in control and data acquisition functions.

Currently, these technologies have advanced with the development of smartphones, smart meters, and improved process controls [14]. Toward the future, more automation, advanced use of information for system management tasks such as leak detection, and more attention to cybersecurity are expected. Additionally, user information is expected to improve substantially.

5. Controls, Instruments, and Utilization of Smart Water Systems

The advancement of technologies for instrumentation and controls is driving the trend toward smart systems in different sectors of the economy. The concept of smart systems is being applied in cities, for example with smart buildings and smart transportation systems, among others. All have similar attributes based on the availability of sensors, actuators, and controls of subsystems for functions such as lighting, energy management, scheduling, and more.

Advancement in control methods follows the availability of technologies but also learning and experience about successful strategies. For example, in the case of smart water systems, some computer-based simulation models have proved to be useful and effective, while others have been developed but discarded due to lack of practical application. A possible example of this situation is the use of control methods that seem promising but have high levels of complexity and maintenance requirements that create a risk of malfunction. The use of inflatable dams placed inside of combined sewers to control overflows seems to fit this category [15]. These were proposed 50 years ago in San Francisco, for example, to address the expensive and perplexing problem of the aging combined sewers. However, their application has been limited. While much of the interest in smart water systems has been driven by technologies and control methods, how customers use smart water information to help transform cities is more strategic. If the past is a guide to the future, it is expected that technologies and control methods will continue to advance incrementally, new business opportunities will be created, and experts and managers will continue to seek out new methods to experiment with them. Meanwhile, the benefits of these advances to customers and how the evolving smart systems are accepted and improve quality of life in cities remain larger questions. This issue is not limited to smart water systems, of course, because people today are confronted with a rapidly changing information environment which affects their daily lives in many ways.

6. Economic Framework of Smart Water Systems

An economic framework for smart water systems will involve the supply and demand for the public and private goods provided, their direct and indirect benefits and costs, and their linkages and impacts due to interdependences with other sectors. The organization of system management to facilitate collective action is also involved, as are social issues stemming from interactions with system customers and other stakeholders [16].

The demands for water, wastewater, stormwater, and recycled water services are different. The water supply and recycled water are commodities that can be sold like toll goods [17]. Most explanations of water sector demands focus on them, but the demands for other services must also be assessed. Wastewater service comprises carrying unwanted used water away from properties and requires paying a portion of the treatment and disposal. Stormwater service involves draining properties and public areas and paying to prevent pollution from urban runoff. The supply of each of these services involves a combination of responding to demands and meeting mandates.

The direct and indirect benefits and costs of the services stem from the demands that are to be met [18]. The benefits of water supply services involve applying the water for uses that range from meeting essential drinking water requirements to satisfying social needs, like health-related hygiene in the home. For wastewater, the direct benefit to customers is the removal of the unwanted residual water, and environmental protection is an indirect benefit. Stormwater services have multiple benefits, including protecting property and facilitating traffic movement in the city. Their multi-faceted attributes have made them a key part of a sustainability assessment protocol [19]. Taken together, the benefits from urban water services range across multiple categories and cannot be quantified easily.

The indirect benefits of urban water services demonstrate the connector roles of water services to various aspects of the economy and society. They stem from linkages and interdependencies with the other sectors, like employment, housing, energy supply, urban transportation, and the environment. Social impacts from water and its interactions can be considered indirect benefits and costs, and there is a substantial literature discussing how to define them [20].

Smart water systems will operate in the context of these interdependencies during normal and emergency times. For example, quality housing depends on meeting all water demands effectively during normal times. If there is a failure, the quality of the housing deteriorates. In the case of emergencies, the interdependencies can lead to systemic failures. For example, during flooding the failure of a stormwater system might trigger failure if the flooding shuts down a part of an electric power system [21].

The economic framework also involves the organization of urban water systems to facilitate effective operation and integration. The organization will depend on the structure of government and on public–private cooperation. Each of the services can be a separate utility or various degrees of consolidation and integration can be forged.

Finance is perhaps the most important issue in the economic framework of smart water systems. The management organizations can operate as utilities, but public goods are included among their services, and financing them may require subsidies along with fees for services. Each of the water services has its own financial structure. For example, water supply is like a public utility. Wastewater involves charges that are mandatory and may be offered like a utility, even if its services do not involve the distribution of a commodity. Stormwater has a unique financial structure and has been moving toward a utility model [22]. Recycled water is difficult to finance but is positioned as a utility to facilitate it [23].

By adding smart capabilities to integrated urban water systems new possibilities for effectiveness, equity, and reliability are created. Perhaps the greatest benefits are to fulfill demands more effectively and to extend services on a cost-effective basis to more people. Smart systems with their information and computing technologies should be able to promote integration through linkages for communication and controls [24]. By sensing the status of systems and their performance, greater reliability should be fostered [25].

In the category of improving performance by reducing failures, smart systems can monitor environmental water quality and water in piped systems to prevent exposure of people to negative situations [26]. Additionally, hydrometeorological sensing networks can provide advance notice of urban flood problems and warnings can be issued [27]. Blocking cyber terrorism is another potential benefit, although the smart capability added by linking the systems may open them to new possibilities for attack [28].

Urban customers can expect more information about total benefits from water services and they can learn more about the total picture of water in cities through education programs. They can also learn about water finance and conservation with information about their bills. Subject to privacy concerns, information can be extended down to the water use of individual fixtures by using the technology of the Internet of Things (IoT).

7. Examples of Smart Integrated Water Systems

The selection of brief case studies as examples is based on innovation and illustrates a range of situations. Singapore is one of the world's major cities, and it is chosen as the primary case study because it has developed its concepts comprehensively over a period of years. Paju Smart Water City in South Korea is smaller than Singapore and its experience is explained due to its connection with an evolving smart city. Fort Collins, Colorado is still smaller and considered a medium city in the United States that is evolving its smart systems in ways that are characteristic of leading-edge cities. Juarez, Mexico is a large city in an arid zone with major issues in sustaining an adequate water supply. It has used smart systems to address major issues of sustainability.

In Singapore, the imminent loss of its source of water motivated the city to create an integrated water system, and this led to the development of a "four taps" approach to water management that includes local catchments, imported water, reuse, and desalination. The four taps provide for integration of sources, reuse provides a link between water supply and wastewater infrastructure, and a separate stormwater system is also used [29].

Singapore's water agency (PUB) has explained its approach to smart water management [30]. The integration will connect the smart drainage grid, smart plants, smart water grid, and smart sewer grid. Technologies to enable the smart features will include machine learning for decision support systems, big data to provide insights about the system, the Internet of things to connect sensors and devices, process simulations for scenarios using digital twins, and robots or unmanned vehicles to perform manual tasks. Singapore anticipates using many smart features in its individual systems. For example, the drainage grid will have extensive hydrometeorological monitoring, the smart plants will feature automation, the smart water grid will include pressure flow and water quality, and the smart sewer grid will include illegal discharge tracing. When these features are fully operational, they will aid Singapore in its quest to be self-sufficient in water and to have greater water security.

Paju is a developing city in the northern part of South Korea, near the 38th parallel. Its population is about 425,000, and it has been the focus of developing a smart water system with the sponsorship of K-Water, South Korea's national water agency. Paju has

been explained as a case study is one of the selections in the study by the International Water Resources Association about smart water management.

Korea has devoted substantial attention to the development of improved integrated water management and the use of intelligent water resources operations technology to promote water security and improved operations [31,32]. This emphasis on using technology is carried out in conjunction with multiple objectives, such as water safety and security, and including a "tastier and healthier water supply". This requires the use of smart water management to provide real-time information about tap water quality to gain public trust. K-water is pursuing objectives to manage source water, optimize water treatment, operate intelligent networks, customize industrial water supply, and optimize wastewater treatment. These goals are translated into water quality management, water quantity management, risk management, energy management, pipe network analysis, and demand forecasting.

So far, the major accomplishments of the Paju Smart Water City Pilot Project focus on improving the reliability of tap water, better customer service, and relieving distrust by citizens of tap water. Prior to the project, only about 1% of the population used water for drinking, but after the project pilot project, this percentage increased to 36.3%. Reports of satisfaction have also improved and the need for water purifiers will be significantly reduced.

Fort Collins, Colorado has an integrated utility where four services are managed together: water supply, wastewater, stormwater, and electric power. The city has recently completed a study of how to foster additional integration by linking raw irrigation water with its water supply system to utilize water sources and infrastructure better [33]. Fort Collins exhibits a small amount of water recycling, but its gravity water supply system would make it expensive to treat wastewater and pump it back uphill. Moreover, the water law system that operates in Colorado places certain barriers in place for water recycling.

The smart water features in the Fort Collins system are typical of those in US cities that are implementing new technologies. For example, the city has district metering areas with valves and remote pressure gauges, access to radar-based weather forecasting, and nondestructive sensors for main wastewater sewers. Fort Collins is customer friendly with its utility services and engages actively with water users by reporting about periodic water uses to encourage conservation. Fort Collins can access other technologies such as advanced metering infrastructure and real-time modeling for water distribution, and it implements them according to need.

The Juarez City case study addresses issues of a major metropolitan area in an arid zone. Juarez and El Paso are twin metropolitan areas coalescing as one urban region. Juarez draws its water from a binational aquifer in which management can be contentious [34]. The city has undertaken an aggressive smart water program based on measurement and control to interact with customers in proactive ways for water conservation. The features of the system are smart meters with high accuracy, pressure control valves, advanced metering infrastructure, and a web platform for customer outreach.

The project is to be self-funded and progress through 2021 has been promising. Implementation started with major users and will be extended to additional smart meter applications and system controls. The eventual goal is to eliminate most water losses, measure and manage water use carefully, and improve the capacity of the utility to continue to serve the growing city, despite water supply limitations.

More case studies are available, and a synthesis can be found in [2]. Many case studies of integrated approaches are also available [35]. The lessons learned converge on a few principal challenges. These will be summarized next, and future directions will be summarized afterward with the identification of trends that are evident from the cases.

8. Challenges to Integration

An understanding of significant barriers to implementing integrated and smart water systems has emerged from the pilot studies. Some of the challenges were summarized in the smart water management study by (IWRA, 2021), which had that as a goal. Other barriers have been identified through studies such as [36], which assessed studies about the implementation of integrated strategies. The understanding of challenges converges to a few main problems.

Most barriers to integration deal with organizational stovepipes and the difficulty in merging or even gaining cooperation among public organizations [37]. Another challenge is the complexity of integrated urban water systems, with their different levels of infrastructure and operations management and their separate regulatory structures. To move the needle will require transformations in institutional cultures to overcome resistance to change. Concessions may be needed to overcome resistance to integration. For example, participants in one case study overcame stovepipes partially by allowing the merged organizations to retain their separate SCADA systems.

In terms of implementing smart attributes, financial barriers were identified most often, with a focus on higher-than-expected costs and lower revenues than needed. Lack of financial support for demonstration and training was also mentioned. A related barrier was the large amount of time and institutional commitment required.

The institutional commitment required competes with other capacity issues, which also pose limiting factors for the implementation of smart systems. Capacity is needed among staff to plan and manage programs, as well as from support organizations to provide services and equipment. For example, in support organizations, the large-scale demand for the production of smart meters may be a constraint. On the utility side, the task of reducing staff positions that perform traditional but now outdated functions can be difficult. Additionally, other stakeholders such as allied agencies must have enough capacity to participate productively, to understand, and support the transition to smart systems.

The local nature of water systems and the diversity it creates constitutes a barrier because, even if an approach works in one place, it may not be possible to replicate it in another with different conditions. This means that pilot studies may have limited applicability, except for very specific components or practices in smart systems.

It is also difficult to assess the benefits of smart systems if incremental performance improvements and providing enhanced information to customers do not yield politically attractive results in the near term. This problem, when combined with the limited funding for smart systems, may defer programs indefinitely.

Given the problems in large cities, the greatest opportunities for innovation may be in small and medium-size cities where different approaches in governance are more feasible. A reality in the vision of integrated urban water systems is the case of struggling cities where public services fall way below minimum standards [38].

Smart water cities hold the potential to improve quality of life and urban environments anywhere in any place where effective governance is in place. When it is, it seems inevitable that urban water systems will change and their performance will respond to the felt needs of people in the cities and that smart systems can help to effect behavioral changes such as greater confidence in tap water for drinking, water conservation, greater use of recycling, and improved urban water environments.

The behavioral changes will occur as people learn more about their water systems, which can occur through the provision of more useful and effective information about the systems. People can learn about health effects, positive environmental management, and preparation for emergencies, among other water-related subjects.

The model shown in Figure 2 provides a way to forecast how these changes can occur. The primary driver will be user information for daily needs and emergency preparation. People have ongoing needs for quality drinking water service, responsible wastewater management, drainage of properties, and common areas. While they may take the services for granted in some cases, with creative approaches to water education, better cooperative arrangements between water managers and water users can take place.

Organizational and governance changes required to facilitate future improvements are likely, although in some situations progress may be slow. Resistance to change is as deep roots both on an individual and organizational basis. Possibility for improvements brought about by new technologies may drive change more rapidly than top-down political decisions intended to force changes prematurely. As new generations of water managers take the reins, they will be more familiar with technological possibilities and search for ways to change the way business is conducted.

With ongoing changes in technologies, it seems certain that unexpected possibilities will develop. The recent availability of social media, smart devices, wireless communications, machine learning, and others have ushered in management possibilities that were heretofore unknown. These have already changed many institutions and although urban water organizations may be slower and less agile than private businesses, they will change as well. The results should be positive to make systems more robust and reliable and for consumers is well, both in quality of service and in effectiveness.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The author declares no conflict of interest.

References

- IWA. The Journey to Water-Wise Cities. 2022. Available online: https://iwa-network.org/projects/water-wise-cities/ (accessed on 25 January 2022).
- International Water Resources Association. Smart Water Management Case Study Report. 2021. Available online: https: //www.iwra.org/swm-2/ (accessed on 25 January 2022).
- 3. Loucks, D.P.; van Beek, E. Urban water systems. In *Water Resource Systems Planning and Management*; Springer: Cham, Switzerland, 2017. [CrossRef]
- 4. Zimmerman, R. Critical infrastructure and interdependency revisited. In *The McGraw-Hill Homeland Security Handbook*, 2nd ed.; Kamien, D.G., Ed.; The McGraw-Hill Companies: New York, NY, USA, 2012; pp. 437–460.
- 5. Nieuwenhuis, E.; Cuppen, E.; Langeveld, J.; de Bruijn, H. Towards the integrated management of urban water systems: Conceptualizing integration and its uncertainties. *J. Clean. Prod.* **2021**, *280*, 124977. [CrossRef]
- US Water Alliance. One Water Roadmap. 2016. Available online: http://uswateralliance.org/sites/uswateralliance.org/files/ publications/Roadmap%20FINAL.pdf (accessed on 25 January 2022).
- 7. Paulson, C.; Broley, W.; Stephens, L. Blueprint for One Water; Water Research Foundation: Denver, CO, USA, 2017.
- 8. LaFrance, D. Total water solutions: Caring about every drop, everywhere. J. Am. Water Works Assoc. 2013, 105, 6. [CrossRef]
- 9. American Planning Association. Integrated Water Resource Management. 2020. Available online: https://www.planning.org/knowledgebase/watermanagement/ (accessed on 25 January 2022).
- Gambrill, M.; Naughton, M.; Kirchner, L.; Goksu, A. Mainstreaming Water Resources Management in Urban Projects: Taking An Integrated Urban Water Management Approach. 2016. Available online: https://www.worldbank.org/en/topic/water (accessed on 25 January 2022).
- 11. Grigg, N.S. Integrated Water Resources Management: An Interdisciplinary Approach; Palgrave Macmillan: London, UK, 2016.
- 12. Ministry of Housing and Urban Affairs. What Do We Mean by a Smart City? 2022. Available online: https://smartcities.gov.in/ about-scm (accessed on 25 January 2022).
- Anton, S.D.; Fraunholz, D.; Lipps, C.; Pohl, F.; Zimmermann, M.; Schotten, H.D. Two decades of SCADA exploitation: A brief history. In Proceedings of the IEEE Conference on Application, Information and Network Security, Miri, Malaysia, 13–14 November 2017. [CrossRef]
- 14. Sims, S. Evolving Control Systems are Key to Improved Performance. Control Engineering. 2021. Available online: https://www.controleng.com/articles/evolving-control-systems-are-key-to-improved-performance/ (accessed on 25 January 2022).
- 15. Ridgway, K.; Rabbaig, M. Use of in-line inflatable dams for CSO control. *Proc. Water Environ. Fed.* **2006**, 2006, 5578–5590. [CrossRef]
- 16. Noll, R. *The Economics of Urban Water Systems. Thirsting for Efficiency: The Economics and Politics of Urban Water System Reform;* Pergamon: Oxford, UK, 2002; pp. 43–63. [CrossRef]
- 17. House-Peters, L.A.; Chang, H. Urban water demand modeling: Review of concepts, methods, and organizing principles. *Water Resour. Res.* **2011**, *47*, W05401. [CrossRef]
- 18. Rietveld, L.C.; Siri, J.G.; Chakravarty, I.; Arsenio, A.M.; Biswas, R.; Chatterjee, A. Improving health in cities through systems approaches for urban water management. *Environ. Health* **2016**, *15*, S31. [CrossRef] [PubMed]
- 19. Institute for Sustainable Infrastructure. Envision Sustainable Infrastructure Framework. 2022. Available online: https://sustainableinfrastructure.org/wp-content/uploads/EnvisionV3.9.7.2018.pdf (accessed on 25 January 2022).

- Cosgrove, W.J.; Loucks, D.P. Water management: Current and future challenges and research directions. *Water Resour. Res.* 2015, 51, 4823–4839. [CrossRef]
- Narayanan, A.; Finucane, M.; Acosta, J.; Wicker, A. From awareness to action: Accounting for infrastructure interdependencies in disaster response and recovery planning. *GeoHealth* 2020, 4, e2020GH000251. [CrossRef]
- Kane, J.; Shivaram, R. As Flood Risks Intensify, Stormwater Utilities Offer a More Resilient Solution. 2017. Available online: https://www.brookings.edu/blog/the-avenue/2017/09/21/as-flood-risks-intensify-stormwater-utilities-offer-a-moreresilient-solution/ (accessed on 25 January 2022).
- De Paoli, G.; Mattheiss, V. Cost, Pricing and Financing of Water Reuse against Natural Water Resources. 2016. Available online: http://demoware.ctm.com.es/en/results/deliverables/deliverable-d4-7-cost-pricing-and-financing-of-water-reuseagainst-natural-water-resources.pdf/view (accessed on 25 January 2022).
- 24. Goldstein, P. Smart Water Technology: How IoT Helps Cities Save Money and Conserve Water. 2020. Available online: https://statetechmagazine.com/article/2020/07/smart-water-technology-how-iot-helps-cities-save-money-and-conservewater-perfcon (accessed on 25 January 2022).
- 25. Moy de Vitry, M.; Schneider, M.Y.; Wani, O.F.; Manny, L.; Leitao, J.P.; Eggimann, S. Smart urban water systems: What could possibly go wrong? *Environ. Res. Lett.* **2019**, *14*, 081001. [CrossRef]
- 26. Cantor, A. Water Distribution System Monitoring: A Practical Approach for Evaluating Drinking Water Quality; Routledge: London, UK; Taylor & Francis: Boca Raton, FL, USA, 2018.
- 27. Hénonin, J.; Russo, B.; Mark, O.; Gourbesville, P. Real-time urban flood forecasting and modelling—A state of the art. *J. Hydroinformat.* **2013**, *15*, 717–736. [CrossRef]
- Clark, R.; Panguluri, S.; Nelson, T.; Wyman, R. Protecting drinking water utilities from cyberthreats. J. Am. Water Works Assoc. 2017, 109, 50–58. [CrossRef]
- 29. Tortajada, C. Singapore: An Exemplary Case for Urban Water Management. Case Study. 2006. Available online: https://thirdworldcentre.org/wp-content/uploads/2020/07/RPP-Nov-6-06-Singapore-an-Exemplary-Case-for-Urban-Water-Management-Report.pdf (accessed on 25 January 2022).
- PUB. Smart Water Management: Singapore's Experience. 2022. Available online: https://events.development.asia/system/files/ materials/2020/11/202011-smart-water-management-singapore-experience.pdf (accessed on 25 January 2022).
- Yi, S.; Ryu, M.; Suh, J.; Kim, S.; Seo, S.; Kim, S. Smart Water Management: Application to Paju Smart Water City. 2021. Available online: https://www.iwra.org/wp-content/uploads/2018/11/3-SWM-Paju-final.pdf (accessed on 25 January 2022).
- 32. Lee, S.-H. Global competition, smart water service. J. Water Policy Econ. 2015, 24.
- Cole, J.; Sharvelle, S.; Fourness, D.; Grigg, N.; Roesner, L.; Haukaas, J. Evaluation of centralized and decentralized strategies for dual water supply: A case study. ASCE J. Water Resour. Plan. Manag. 2017, 144, 05017017. [CrossRef]
- Ibáñez, O.; Lazo, J.; Meza, R.; Miranda, A. Juarez City, Mexico: Smart network for meter reading and pressure control management. In Proceedings of the World Water Congress XVII, Daegu, Korea, 29 November–3 December 2021; International Water Resources Association: Paris, France, 2021.
- American Water Resources Association. Case Studies in Integrated Water Resources Management: From Local Stewardship to National Vision. 2013. Available online: https://inyo-monowater.org/wp-content/uploads/2012/11/AWRA-Case-Studies-IWRM.pdf (accessed on 25 January 2022).
- 36. Mukheibir, P.; Howe, C.; Gallet, D. Institutional issues for integrated 'One water' management. In *Water Environment Research Foundation, Water Research Foundation and Water Research Australia*; IWA Publishing: London, UK, 2015.
- 37. Entwistle, T.; Martin, S. From competition to collaboration in public service delivery: A new agenda for research. *Pub. Adm.* 2005, 83, 233–242. [CrossRef]
- Tucci, C.E.M. Integrated Urban Water Management in Large Cities: A Practical Tool for Assessing Key Water Management Issues in the Large Cities of the Developing World. World Bank. 2009. Available online: https://collaboration.worldbank.org/ content/sites/collaboration-for-development/en/groups/tdlc-technical-deep-dive/groups/iuwm-south-south-knowledgeexchange/documents.entry.html/2019/01/23/general_iuwm-WqPt.html (accessed on 25 January 2022).





Article Designing Geographic Information System Based Property Tax Assessment in India

Anu Singh¹, Suraj Kumar Singh², Gowhar Meraj³, Shruti Kanga^{1,*}, Majid Farooq³, Nikola Kranjčić^{4,*}, Bojan Đurin⁵ and Sudhanshu¹

- ¹ Centre for Climate Change & Water Research, Suresh Gyan Vihar University, Jaipur 302017, Rajasthan, India; anu.49647@mygyanvihar.com (A.S.); cm@mygyanvihar.com (S.)
- ² Centre for Sustainable Development, Suresh Gyan Vihar University, Jaipur 302017, Rajasthan, India; suraj.kumar@mygyanvihar.com
- ³ Department of Ecology, Environment and Remote Sensing, Government of Jammu and Kashmir, Srinagar 190018, Kashmir, India; gowharmeraj@gmail.com (G.M.); majid_rsgis@yahoo.com (M.F.)
- ⁴ Faculty of Geotechnical Engineering, University of Zagreb, Hallerova aleja 7, 42000 Varaždin, Croatia
- ⁵ Department of Civil Engineering, University North, 42000 Varaždin, Croatia; bdjurin@unin.hr
- Correspondence: shruti.kanga@mygyanvihar.com (S.K.); nikola.kranjcic@gfv.unizg.hr (N.K.)

Abstract: Property tax is the primary source of revenue for municipal bodies. In India, municipal corporations are facing issues in property tax collection, and the primary reason for it is a lack of count of assessed properties under its jurisdiction. Also, the storage of information on the properties is mainly based on manual efforts, which leads to data redundancy and failure to appropriate tax collection. Geographical Information Systems (GIS) consists of technology, personnel, and resources to create, maintain, visualize, search, and share geospatial data and services. The study has been carried out in the Hauz Khas Ward, South Delhi Municipal Corporation, Delhi. This paper aims to develop a spatial database for property tax management. It includes capturing the building footprint, road, land use such as parks, paved area, drains, and demarcation of boundaries such as locality slums, based on a regular grid net with a cell size of 250 m by 250 m. The generated geospatial database has been finally used to evaluate parameters for property tax calculation. Moreover, this spatial database can be organized as different models for any web-based application for municipal services. This study provides a working example of a GIS-based property tax collection solution for whole of India and other South-Asian countries.

Keywords: property tax; GIS; geodatabase; spatial data; decision support system

1. Introduction

Worldwide more than 130 countries collect taxes on the properties, and almost all countries consider it the prime source of revenue for the local governments to administrate effectively [1,2]. Moreover, developing countries rely on it more than developed ones and acknowledge the importance of so-called local tax [3,4]. Even in most of the developing countries, the local governments rely on property-related revenues to service out other essential services such as education and health [5]. Now that the global community has acknowledged the significance of taxation as a 'state-building' process, the practical consequences of implementing a strategy to tax reform require clarification and efficient procedure to implement.

Property tax is a levy on property that the owner is required to pay to the governing authority of the area where the property exists. For any development, property tax is one of the major sources of income state or country to cover the expenditure of development [6]. The local or municipal authorities are bound for the development of their administration, and without capital, it can become a challenge for them. Besides, taxation can be used as an urban management tool that can track land use, urban expansion, land market, and



Citation: Singh, A.; Singh, S.K.; Meraj, G.; Kanga, S.; Farooq, M.; Kranjčić, N.; Đurin, B.; Sudhanshu Designing Geographic Information System Based Property Tax Assessment in India. *Smart Cities* **2022**, *5*, 364–381. https://doi.org/ 10.3390/smartcities5010021

Academic Editors: Isam Shahrour, Fateh Belaid and Véronique Flambard

Received: 15 February 2022 Accepted: 14 March 2022 Published: 16 March 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). transactions related to properties [7]. As a matter of fact, for property taxation, location is foremost as it matters to unit value for tax calculation. The location as a spatial attribute of property is crucial for the analysis or estimation of many aspects related to property or urban management.

India also has the same scenarios of revenue collections related to property. At present, there are about 3842 urban local bodies (ULBs) in India, wherein 2108 are Nagar Panchayats, 1595 are Municipalities, and 139 are Municipal Corporations. Mainly there are three sources from where ULBs generate their revenue, i.e., revenue shared taxes, exclusive taxes, and nontax revenues. The exclusive taxes include property, entertainment vacant land, advertisement, and professional taxes. While the property tax is the most important tax levied among all the taxes at the local scale. For any local or municipal body, property tax reflects the relation between the types of services financed at the local level vs. the benefit to property values. Hence, it is considered a good tax. In India, the state government is responsible for designing the property tax system, and the ULBs control the tax rate and collection planning.

In India, many studies have been carried out to lay down the taxation system of the properties. A study carried out by the National Institute of Urban Affairs (NIUA) in 10 Indian cities shows that the trend has shifted from the annual rental value-based system to an area-based system in many cities [8]. The area-based taxation is based on factors such as location, usage, type of construction, and age of the building. Many corporations have introduced the self-assessment scheme, and a few, like in Bangalore, have introduced online tax payment. As mentioned earlier, the success of reforms has been uneven. While the revenue productivity in Bangalore has shown a sharp increase, the revenues have continued to be stagnant in many other metros, including Delhi [9].

Noncompliance with property taxes is common in south Asian countries, and it can worsen their liquidity problems [7]. Revenue elasticities and metrics of taxpayer hardship are used to characterize optimal enforcement and taxing regimes [5]. These characteristics are calculated utilizing a variety of sources of variance as well as administrative data. Rate hikes and increased enforcement generate revenue, but liquidity restrictions also influence taxpayer behavior [4]. Despite liquidity restrictions, increasing tax rates improve wellbeing. On the other hand, enforcement has larger private expenses than welfare benefits. Mostly on margin, welfare-maximizing authorities would instead raise tax rates than strengthen enforcement [3]. Therefore an efficient system is required for assisting the collection and management of the property taxes.

Property tax can be a significant source of revenue, but practically it fails to collect expected revenue efficiently. In Delhi, the method of tax calculation is unit based for which many factors are required, such as property address, colony name, colony category, no. of floors, covered area, vacant land area, type of property (residential/commercial), type of occupancy (owner/tenant), type of structure (kaccha, pacca, semi pucca), year of construction and owner type (single/joint). A door-to-door ground survey is essential to complete this database. Proper assessment of properties and efficient tax collection are vital for the ULBs. The current infrastructure has insufficient information to ease the process of tax status verification. For tax inspectors, it is imperative to identify the defaulters who are not paying appropriate tax in terms of its factors or not paying at all. The good thing about property taxation is that the primary object, i.e., property, is immoveable. But again, if you don't know its location, you can't manage it. Therefore, many municipal bodies are planning to adopt GIS-based property tax systems to boost their revenues. GIS is a modern system of tools for effective and efficient storage, retrieval, and manipulation of spatial and non-spatial data for attaining efficient management and policy-making information [10]. GIS can provide a system for inputting spatial data and intelligence for planning and monitoring project implementation [11].

GIS is used to locate assets, track information, and save an organization time and resources. Multiple analyses can be done on a database that can be readily displayed on any GIS-based platform with various color-coded parameters that the corporation's officials

can easily interpret. One can quickly identify the number of households who want to know the defaulters in their assigned area. It provides a visual result in a hard copy or digital map about where those properties are located. Moreover, it was challenging to identify the impact of location on land value as determining accurate location was very challenging [12]. Nowadays, using GIS, to analyze spatial attributes of any number of entities is relatively easy [13–16]. Also, GIS mapping is of great significance in scientific research, planning, and management [17]. GIS technologies can handle large volumes of data from multiple sources, integrating them to produce information in a spatial context through maps and models [18]. Applying GIS-based municipal information tools can lead to the solution of these data management problems. Thus, GIS can be a handy tool for municipal planning and decision-making that evaluates and assesses assets. Not just worldwide but also in India, many local and municipal governments use GIS as a decision-making tool to design and support development programs [19].

This study is a pilot study to be carried out for the whole of India initially with the four major metropolitans, Delhi, Mumbai, Kolkatta, and Chennai. The present study is the representative study of one of the wards of South Delhi, i.e., Hauz Khas. The agency responsible for levying the tax on properties in South Delhi is South Delhi Municipal Corporation (SDMC). The present work aimed to create a geospatial database of all the properties present in Hauz Khas to take them under the tax net of SDMC. In Delhi, the property tax calculation is based on the unit area method. Since 2004, the collection of property tax has been moved to the self-assessment method. Since then, the number of assessed properties has decreased [20]. Also, because it is a self-assessment basis, SDMC has information for those properties that have filed the property tax. The properties which have never paid the tax have not come under the notice of SDMC. Therefore there is an ever-increasing need for digital spatial data to be more feasible and reliable for this decision-making system [21–25].

The paper demonstrates the need as well as how a successful spatial database consisting of multiple layers in GIS environment is created to provide a solution through ground reflective analyses and could be easily accessed through an intranet or internet application [26–28]. This work is also designed as an enterprise GIS for municipalities all over India that can customize attributes and geospatial data with spatial and non-spatial models and operations to create a service-oriented architecture from Geo web services [29–32]. The study also demonstrates how planned spatial data (attribute and non-attribute) such as building footprint, roads, drains, parks, location boundaries are digitized for the jurisdiction of Hauzkhas Ward, South Delhi Municipal Corporation, in the GIS environment.

2. Materials and Methods

2.1. Study Area

Delhi is located at a latitude of 28°34′ N and a longitude of 77°07′ E, having an average elevation of 233 m (ranging from 213 to 305 m) above the mean sea level. The NCT has three local municipal corporations: Municipal Corporation of Delhi (providing civic amenities to an estimated 13.78 million people), New Delhi Municipal Council (New Delhi), and Delhi Cantonment Board. The total geographic area of Delhi is 1483 km² (Rural-689 km², urban-624 km² and forest-170 km²). Recently Delhi Municipal Corporation of Delhi has been trifurcated into three smaller Municipal corporations—North Delhi Municipal Corporation, South Delhi Municipal Corporation, East Delhi Municipal Corporation. Among all the corporations, South Delhi Municipal corporation is the largest one, having 667.38 km². The South Delhi municipal corporation is further divided into four-zone and 104 wards. Hauz Khas ward (Ward No.62S) falls under the South Zone of South Delhi Municipal corporation (Figure 1). It consists of planned, unplanned as well as commercial properties.



Figure 1. Location of the study area. (a) India, (b) Delhi Municipal Corporation, (c) Hauz Khas ward.

2.2. Methods

For this study, the primary concern is building, as it is fundamental for property tax calculation. For the level of visual interpretation required in this study, WorldView-2 imagery has been used. The resolution of the world view is 0.5 m which was suitable for the interpretation of building edges and partitions. A visible feature like building, land-cover, drains can be captured using this image, but the boundary of localities is not possible with it. For interpretation of localities, blocks, colonies, other sources such as Google or Eicher maps are useful. For this paper, a reference to Eicher map-2007 and Google map has been used. ArcGIS (version 10.1), an ESRI suite platform, was opted for geodatabase creation. Figure 2 shows an overview of the overall methodology.

The primary data, i.e., categorized property data, was collected through field surveys using the property tax form (Annexure 1, 2). The secondary, i.e., wards boundary gazette notification, slum boundary list, enumeration block extent list, SDMC property tax form, and colony categorization lists have been collected from the concerned departments. The unique key between the property survey form and spatial data of the building was proposed to be an identifier referred to as BUID. On the platform of ArcGIS, the survey data have been mapped using the *geocode tool*. The survey data led to creating three related layers, i.e., building, properties, and unit.



Figure 2. Overall methodology opted for in the present study.

Since for property taxation, "Property" is the foremost requirement, and it can be one building or part of a building or cluster of buildings. During digitalization, all buildings have been assigned a BUID with a unique number for each building so the polygons can be used to map real-world properties. All buildings were assigned with a reference number ranging from 1 to N within the Grid, where N is the maximum count of buildings in any grid having limits of 99999. This reference number was formulated as a five digit alphanumeric text. To make it 5 digits, an additional pre "0" has been added to the attribute's reference number (RN). For example, if the given Grid no. 30274, the building has been given reference no. from 1 to 114. The building Id will be "00001", 00010, 00100, etc.

The unique reference number (URN) is a 10-digit number combined with Grid number and Building ID. This was a means of creating a unique reference number for each building throughout the data. In Table 1, the design and creation of BUID have been shown.

Map Number	Building Reference Number	Building ID	BUID
30274	1	00001	3027400001
30274	2	00002	3027400002
30274	81	00081	3027400081
30274	82	00082	3027400082
30274	112	00112	3027400112
30274	113	00113	3027400113

Table 1. Creation of Building Unique ID.

For digitization of location boundary, Eicher image has been used. Eicher is an image having visualized major base layer information. The colonies and locations are generally separated by any natural or manmade boundary such as road, drain, building, or landcover pattern. A Reference Google image has also been taken for this task. Information like the name of a major road, major drains, localities, and major landmarks has been added in the data using Eicher and Google.

3. Results and Discussion

3.1. Geodatabase of the Implementing the Property Tax

Understanding the ground reality in terms of available features, structures, and economic activities is essential to creating any management and planning activities for any area. This requires digitization of all the available features on the ground, such as buildings, lands, forests, water bodies, and roads within the municipal boundaries [33–37]. This is wherein GIS-based geodatabase plays come into play. The data captured as a .shp file and stored in a .gdb format at the ESRI Platform forms the data input process. The geodatabase is a structured way to store multiple shapefiles of different feature types, whether point or line or polygon [38]. The structure of the geodatabase used in this study is presented in Figure 3. The set of shapefiles has been categorized in geodatabase based on its entity type. Before capturing the data in the shapefile, required fields with the name have been created as per the data type and field length.



Figure 3. Geodatabase structure of the Hauz Khas ward (I).

There are four major geodatabases (*.gdb*)categorized in the present work, i.e., building, road, land use, and boundary, under which various other essential information layers are stored either by digitization or creation. One of the important layers in this project is a grid of 250×250 m which is the base for creating a unique ID for building and property survey maps. Each Grid is assigned a unique number, and each building with the Grid has
also been assigned a unique number. A combination of both the numbers at the Grid and building will create a unique ID for the building, i.e., BUID (Figures 4 and 5).

On the ground, there are multiple properties. By assigning a unique no. to each property within the building, it will create a Unique ID to each property. In another way, the concatenation of Grid no., building no, and property no. will create a Unique ID for property (PID: Property ID).



Figure 4. Geodatabase structure of the Hauz Khas ward (II).

The creation of spatial data is majorly an image interpretation job. Before capturing any building, a 250×250 m grid has been created. This Grid has been created using the fishnet tool in Arc GIS [39]. Each Grid has been assigned a unique number. The same grid number will be updated in the building under the respective Grid. It will not just help in tracking of digitization of data but also will help create a unique ID for Building.



Figure 5. Reference number (RN) for each building in the Hauz Khas ward.

SDMC colony boundaries were mapped starting with the enumeration block. Enumeration Block (EB) means a specific area allotted to one particular enumerator to carry out census operations relating to population enumeration. The State Election Commission defines these EB Boundaries. SDMC uses these EB Boundaries to form SDMC colony boundaries. The municipal valuation committee decides the category of colonies. The boundary of wards and colonies is most likely to change within its Assembly constituency boundary every five years. Demarcation of EBs on the digital platform is a one-time exercise that will be a base for future modification of ward or colony boundaries.

The digital profile of the Hauz Khas ward has been created, which has visual information of building, road, use of lands, and locality. Land use and land cover information are important for several planning and management activities concerned with the surface of the earth [40,41]. The land-use types are estimates of the study area is given in Table 2. The land use distribution can define and predict human activity using agriculture, housing, or institutional. Ward area of Hauzkhas is 3.14 km² having 5620 buildings under its jurisdiction. The covered area of the building is 0.93 km² in Hauzkhas, i.e., 29% of the area, which is the tax-net for the SDMC (Figure 6). Through this approach, only horizontal expansion of built-up can be determined. Still, again for the actual covered area, it is imperative to calculate the vertical built-up, which can be best possible through the door-to-door survey of these buildings. Apart from this, approximately 20% of the area is covered by roads and paved areas. Since the Hauzkhas ward is planned, 23% of the area has been dedicated to the gardens and park. Since many of these portions are owned privately, this can be maintained the same in consecutive years. One more thing is very focal in this study for the municipal that approx. 23% of the area is open land, which is prone to have unauthorized construction leading to unauthorized colonies.

Land Use	Area (m ²)	Percentage
Water Body	0.02	0.005
Building	92.57	29.50
Open land	72.36	23.06
Others	12.11	3.86
Park	72.48	23.09

Table 2. Land use/Land cover of Hauz Khas Ward estimates.

Paved Area

Road Edge

Apart from the land-use profile, this database has also resulted in a building polygon with a unique ID. The decided range for building reference is 99999 for each Grid, which is quite enough as each Grid area is 250×250 m². The 250×250 m grid size is a requirement under the Government of India guidelines to map buildings within such grids under this framework. Very high-scale grids would complicate the GIS-based processing and shall require more computing power (storage and processing). Therefore the 250×250 m grids were selected after various hit and trial experiments to avoid computational and data storage issues and, at the same time, was fair enough to provide a rigorous comparison.

26.87

37.45

Since the data type of Building ID is alphanumeric, an alphabet can be added to the ID for any adverse condition such as partition or new building construction in the same building.

This BUID is key to smooth access of these polygons to use it to collect property information through field surveyor to link it with other digital databases or to map any other attributes. For the development and maintaining the ward, this exercise can provide a foundation where multiple analyses and predictions can be made.

8.56

11.93



Figure 6. Land use map of Hauz Khas Ward.

3.2. Property Tax Implementation

Property tax isn't used to its full potential due to low property valuation rates and ineffective revenue collection. Municipalities and local civic bodies' largest difficulty today is defining improved property valuation procedures, keeping a regular track of income receipts, and calculating losses on exemptions [42]. GIS aids in the monitoring of property tax revenue by establishing a platform that spatially interconnects all property-related data, such as the number of storeys in each building, the total structured area of each building, individual plot areas, locality details, and municipal colony boundaries, to the tax calculation principles.

The formula of property tax is

Annual value \times Tax Rate

In Delhi, the rate of tax is divided into eight categories from A to H (Table 3).

Category	Residential Property	Commercial Property	Industrial Property	
А	12%	20%	15%	
В	12%	20%	15%	
С	11%	20%	12%	
D	11%	20%	12%	
E	11%	20%	12%	
F	7%	20%	10%	
G	7%	20%	10%	
Н	7%	20%	10%	

Table 3. Rate of Colony Categories for Property Tax Calculation.

To devolve a robust database, the creation of thematic maps of the property tax formula parameters will be a major advantage. For example, the database will have visible information like use factor such as commercial, residential; Age factor means a year of construction of the building; Structure factor such as kachha, pucca; Occupancy factor such as self-occupied, tenants.

> Annual value = Unit area value per $m^2 \times Unit$ area of property \times Age factor \times Use factor \times Structure factor \times Occupancy factor

Apart from this, many other rebate factors are applicable for tax calculation. More the variable stored in the database directly proportionates to the improvement of the database. Access to these kinds of the color-coded map through the administration system of property tax will be boon for the tax inspector. It will be easy to identify and verify the values given by taxpayers. The GIS information layers are shown below as maps (Figure 7a,b).

Also, for calculating annual value, one of the factors is Unit area value per m² which is the assigned value per square of meter built-up area of the property. It also varies on colony categories.



Figure 7. Cont.



Figure 7. Information required for calculating the annual tax i.e., structure type, building type, floor distribution, and owner type. (a) Zoomed-in view (b) Overall Hauz Khas.

3.3. Mapping of Taxpayers Data

There is no regular monitoring of whether property tax has been paid or not. As with the case of property tax, there is no assessment of whether owners make proper property taxes in terms of its factors. In the current system, there are various blockers to identify such defaulters. If any tax has never been paid for any property, then the record of the property does not exist in the database. Even the tax for the property has been paid since property tax payment is a self-assessment method, so to verify the details of property provided by the owner, the only option for a tax inspector is ground verification (Figure 8).



Figure 8. Annual tax payers status in the geodatabase.

Using the spatial digital property database would be a systematic and easy process to identify these issues [43]. SDMC has two types of taxpayer data: Online tax data paid by the taxpayer using an online portal and a ledger that has information on those taxpayer data that pay tax offline. Both the data contain information about taxpayers, including addresses and arrears. Joining SDMC taxpayer data with the property database helps bring out the required quality for developing an application that includes correlation and computerization of owner details and tax-paid history on the spatial platform to visualize the property from all aspects. After spatial joining and data sorting, analysis is done on each record to identify the status based on criteria, i.e., tax Paid and Not Paid (Figure 8). The final analysis result will be colony-wise sorted spatial data with information of property details on the ground and given by the owner concerning its tax history.

The amount of money collected from property tax in India is currently low. Its revenues currently account for just 0.2 percent of India's GDP, but it accounts for 3 to 4% of GDP in wealthy countries such as Canada and the United States [44]. Even a small increase in revenue from these might help local municipalities significantly grow their infrastructure, perhaps enhancing their development operations. Therefore an efficient property tax system is essential for development at grass-roots levels in India.

In fact, every country works in the field of its economic development and pushes itself to rise with other global powers. This is a joint exercise between the government and the people [45,46]. Taxation is one such system that acts as a feedback loop [47]. The government necessarily has to establish a system that would help the people have enough money to keep up with the rising output scales of any form of taxation. With automation gaining a central stage in India, various options are available for people to participate in the economic cycle. While industrialized countries already have significant shifts in revenues from the taxations, developing countries like India are trying to cope with their pace with the systems like the one discussed in the present study. Various studies suggest that developing countries should deliberately implement automation to maximize the financial outputs [48,49]. This study attempts to address India's progress in tax automation in the form of spatial information systems, GIS. Existing property tax collection systems do not try to close the gap in the efficient assessment of the taxes [50]. Most crucially, planned GIS-based automation must be implemented through solid regulations that recognize property-specific flexibility and allow reinstatement methods to achieve effective economic returns that ultimately help in restoring ecosystem services [51,52].

4. Conclusions

The study demonstrates the working process of a geospatial database of various information parameters to calculate the property tax in the Hauz Khas ward of South Delhi. A well-organized geodatabase is created with four main types of information databases, building, road, land use, and boundary. Within these, a lot of other parameters, e.g., property address, colony name, category of the colony, no. of the floor, covered area, vacant land area, type of property (residential/commercial), type of occupancy (owner/tenant), type of structure (kaccha, pacca, semi pucca), year of construction and owner type (single/joint) were incorporated aimed at the calculation of property of the study area. For property taxation, there are mainly three steps, i.e., identification of properties to be taxed, accounting and creation of tax based on property parameter, and collection of tax, including arrears. All this information was collected through the image and a door-to-door ground survey to complete this database. The created database will make this process effortless and in time. It also provides control to secure the database and can be backed up to prevent data losses. Again, maintenance and updating of the database are much faster than the current hardcopy data format. It can overlay with any other layer, such as planning the layout of the development area and roads. This database can be used by other administrative divisions and will be easily upgraded by exchanging information of property tax department with other departments. So, the cost can be shared by multiple divisions of government. On the other hand, it will provide a channel for the

interaction of citizens and municipalities. Both parties can access the data with different rights of control. Having all information on the property will bring transparency in local taxation. Municipal applications need databases at a large scale for micro-level planning. Digital database creation is the primary and most crucial step towards development. Since the data has been captured on the GIS platform for fieldwork activity and the transfer of multiple attributes of field data, it will act as a powerful tool. Again, for the evaluation and development of colonies, other information on infrastructures such as type of road and drain and governing bodies is vital. This one-time task will be the key to multiple solutions to municipal challenges. This database can systematically store historical, current, and future databases and be utilized for analysis or development. Since there are many GIS platforms, this can be used and accessed through any IT-enabled GIS application.

Author Contributions: Conceptualization, A.S., S.K. and G.M.; methodology, S.K., S.K.S., M.F.; software, A.S., S.K., S.K.S., G.M., B.D., S. and N.K. validation, A.S., G.M. and M.F.; formal analysis, A.S.; investigation, A.S., S.K., S.K.S., G.M., B.D., S. and N.K.; resources, B.D., S.; data curation, A.S., and G.M.; writing—original draft preparation, A.S., S.K. and G.M.; writing—review and editing, A.S., G.M. and S.K.S; visualization, A.S. and M.F.; supervision, S.K.; funding acquisition, B.D. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Data available upon request from the authors.

Acknowledgments: The authors express their gratefulness to the three anonymous reviewers for their valuable comments and suggestions on the earlier version of the manuscript that greatly improved its content and structure. The author G.M. is thankful to Department of Science and Technology, Government of India for providing the Fellowship under Scheme for Young Scientists and Technology (SYST-SEED) [Grant no. SP/YO/2019/1362(G) and (C)].

Conflicts of Interest: The authors declare no conflict of interest.

References

- Eckert, J.K.; Goudemans, R.J.; Almy, R.R. Properp Appraisal and Assessment Administration; The International Association of Assessing Otlicers: Kansas City, MO, USA, 1990; p. 6.
- Dillinger, W. Urban Property Tax Reform: Guidelines and Recommendations. World Bank Urban Management Program Tool #1; World Bank: Washington, DC, USA, 1991.
- 3. Bahl, R.; Linn, J. Urban Public Finance in Developing Countries; Oxford University Press: New York, NY, USA, 1992.
- Bahl, R. Implementation Rules for Fiscal Decentralization. Paper Presented at the International Seminar on Land Policy and Economic Development, Land Reform Training Institute, Taoyuan City, Taiwan, 17 November 1998; International Studies Program, School of Policy Studies, Georgia State University: Atlanta, GA, USA, 1999.
- Corker, L.; Nieminen, J. Improving Municipal Cash Flow—Systematic Land Information Management. In Proceedings of the International Conference on Spatial Information and Sustainable Development, TS 6.1 Economic Values, Nairobi, Kenya, 2–5 October 2001; pp. 1–12.
- 6. Soeb, P.; Zahari, W.; David, M.; Fazira, S. Property Tax Management Model of Local Authorities in Malaysia Soeb Pawi. *Chinese Bus. Rev.* **2011**, *10*, 1–12.
- Kundu, D.; Ghosh, D. Innovations in property taxation systems in India. In *Innovative Land and Property Taxation*; Sietchiping, R., Ed.; UN-Habitat: Nairobi, Kenya, 2011; pp. 92–109, ISBN 978-92-1-132407-5.
- 8. National Institute of Urban Affairs (NIUA). Available online: https://www.niua.org/urban-india/urban-india-2010-jan-jun (accessed on 12 December 2021).
- 9. Gupta, R.D.; Garg, P.K.; Arora, M. A GIS Based Decision Support System for Developmental Planning in Dehradun District. J. Indian Cartogr. 2001, 21.
- 10. Bell, M.E. An Optimal Property Tax: Concepts and Practices. In Proceedings of the Intergovernmental Fiscal Relations and Local Financial Management, Almaty, Kazakhstan, 17–21 April 2000.
- 11. Mathur, O.P.; Thakur, D.; Rajadhyaksha, N.; Bahl, R. *Urban Property Tax Potential in India*; National Institute of Public Finance and Policy: New Delhi, India, 2009.
- Stylianidis, E.; Roustanis, T.; Karanikolas, N. A Geographical Information System for Real Estate (GEOVAL) BT—Location Based Services and TeleCartography II: From Sensor Fusion to Context Models; Gartner, G., Rehrl, K., Eds.; Springer: Berlin/Heidelberg, Germany, 2009; pp. 317–329, ISBN 978-3-540-87393-8.
- 13. Arbia, G. Spatial Data Configuration in Statistical Analysis of Regional Economic and Related Problems; Kluwer Academic: Dordrecht, The Netherlans; Boston, MA, USA, 1989; ISBNs 0792302842/9780792302841.

- 14. Tomlin, C.D. *Geographic Information Systems and Cartographic Modeling*, 1st ed.; Prentice-Hall, Inc.: Englewood Cliffs, NJ, USA, 1990; ISBN 0-13-350927-3.
- 15. Huxold, W.E. An Introduction to Urban Geographic Information Systems; Oxford University Press: New York, NY, USA, 1991; ISBN 9780195065343.
- 16. Star, J.; Estes, J. Geographic information systems: An introduction. Geocarto Int. 1991, 6, 46. [CrossRef]
- 17. Aronoff, S. Geographic information systems: A management perspective. *Geocarto Int.* 1989, 4, 58. [CrossRef]
- 18. Fedosin, S.; Yamashkin, S.A. Technological process of solving the problem of modeling the structure of land use based on remote sensing data. *Sci. Tech. Bull. Volga Reg.* **2014**, *6*, 356–359.
- 19. Farooqi, S.A.; Gazali, S. Application of Geographical Information System in Urban Management and Planning: A Case Study of Kulgaon-Badlapur, Dist-Thane, Maharashtra. *Int. J. Adv. Remote Sens. GIS* **2014**, *3*, 476–485.
- 20. Bandyopadhyay, S. Property Tax Reforms in India: A Comparison of Delhi and Bangalore; Georgia State University: Atlanta, GA, USA, 2013.
- Samadzadegan, F.; Alvand, A.; Abootalebi, A.; Hasanlou, M. Enterprise GIS For Municipalities—A Service Oriented Approach. In Proceedings of the The International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences; Leibniz University Hannover Institute of Photogrammetry and GeoInformation: Hannover, Germany, 2008; pp. 1133–1136.
- 22. Idrizi, B.; Sulejmani, V.; Zimeri, Z. Multi-Scale Map for Three Levels of Spatial Planning Data Sets for the Municipality of Vitia in Kosova. In Proceedings of the 7th International Conference on Cartography and GIS, Sozopol, Bulgaria, 18–23 June 2018.
- Krishna, G.; Biradar, C. Geo-Big Data in Digital Augmentation and Accelerating Sustainable Agroecosystems. In Data Science in Agriculture and Natural Resource Management; Springer: Singapore, 2022; pp. 221–242.
- 24. Sibilla, M.; Abanda, F.H. Multi-Criteria Decision Making Optimisation Framework for Positive Energy Blocks for Cities. *Sustainability* **2022**, *14*, 446. [CrossRef]
- Olawumi, T.O.; Chan, D.W. Cloud-based sustainability assessment (CSA) system for automating the sustainability decisionmaking process of built assets. *Expert Syst. Appl.* 2022, 188, 116020. [CrossRef]
- Ahansal, Y.; Bouziani, M.; Yaagoubi, R.; Sebari, I.; Sebari, K.; Kenny, L. Towards Smart Irrigation: A Literature Review on the Use of Geospatial Technologies and Machine Learning in the Management of Water Resources in Arboriculture. *Agronomy* 2022, 12, 297. [CrossRef]
- El Hoummaidi, L.A.L.A.; Larabi, A. Dubai Smart Map, A New Tool Enabling Geo-Digital Government, Driven By Innovative Geospatial Thinking For A Smart And Sustainable Dubai. Int. J. Civ. Struct. Environ. Infrastruct. Eng. Res. Dev. 2019, 9, 17–36.
- 28. Ujjwal, K.C.; Garg, S.; Hilton, J.; Aryal, J.; Forbes-Smith, N. Cloud Computing in natural hazard modeling systems: Current research trends and future directions. *Int. J. Disaster Risk Reduct.* **2019**, *38*, 101188.
- 29. Pretorius, R.W.; Carow, S.; Wilson, G.; Schmitz, P. Using real-world engagements for sustainability learning in ODeL in the Global South: Challenges and opportunities. *Int. J. Sustain. High. Educ.* **2021**, *21*, 1316–1335. [CrossRef]
- Shah, P. IGiS for Managing Cities Smartly–Urban Geoinformatics. Int. Arch. Photogramm. Remote Sens. Spat. Inf. Sci. 2021, 46, 509–513. [CrossRef]
- Rowland, A.; Folmer, E.; Beek, W. Towards Self-Service GIS—Combining the Best of the Semantic Web and Web GIS. ISPRS Int. J. Geo-Inf. 2020, 9, 753. [CrossRef]
- Miloudi, L.; Rezeg, K. Leveraging the power of integrated solutions of IoT and GIS. In Proceedings of the 2018 3rd International Conference on Pattern Analysis and Intelligent Systems (PAIS), Tebessa, Algeria, 24–25 October 2018; pp. 1–7.
- Meraj, G.; Kanga, S.; Kranjčić, N.; Đurin, B.; Singh, S.K. Role of natural capital economics for sustainable management of earth resources. *Earth* 2021, 2, 622–634. [CrossRef]
- 34. Shyam, M.; Meraj, G.; Kanga, S.; Farooq, M.; Singh, S.K.; Sahu, N.; Kumar, P. Assessing the Groundwater Reserves of the Udaipur District, Aravalli Range, India, Using Geospatial Techniques. *Water* **2022**, *14*, 648. [CrossRef]
- Kanga, S.; Singh, S.K.; Meraj, G.; Kumar, A.; Parveen, R.; Kranjčić, N.; Đurin, B. Assessment of the Impact of Urbanization on Geoenvironmental Settings Using Geospatial Techniques: A Study of Panchkula District, Haryana. *Geographies* 2022, 2, 1–10. [CrossRef]
- 36. Tomar, P.; Singh, S.K.; Kanga, S.; Meraj, G.; Kranjčić, N.; Đurin, B.; Pattanaik, A. GIS-Based Urban Flood Risk Assessment and Management—A Case Study of Delhi National Capital Territory (NCT), India. *Sustainability* **2021**, *13*, 12850. [CrossRef]
- Farooq, M.; Gazali, S.; Dada, M.; Gera, N.; Meraj, G. Forest Fire Alert System of India with a Special Reference to Fire Vulnerability Assessment of the UT of Jammu and Kashmir. In *Disaster Management in the Complex Himalayan Terrains*; Springer: Cham, Switzerland, 2022; pp. 155–167.
- 38. Bajjali, W. Geodatabase. In ArcGIS for Environmental and Water Issues; Springer: Cham, Switzerland, 2018; pp. 103–116.
- 39. Musiaka, Ł.; Nalej, M. Application of GIS Tools in the Measurement Analysis of Urban Spatial Layouts Using the Square Grid Method. *ISPRS Int. J. Geo-Inf.* 2021, *10*, 558. [CrossRef]
- Farooq, M.; Meraj, G.; Kanga, S.; Nathawat, R.; Singh, S.K.; Ranga, V. Slum Categorization for Efficient Development Plan—A Case Study of Udhampur City, Jammu and Kashmir Using Remote Sensing and GIS. In *Geospatial Technology for Landscape and Environmental Management*; Springer: Singapore, 2021; pp. 283–299.
- 41. Meraj, G.; Farooq, M.; Singh, S.K.; Islam, M.; Kanga, S. Modeling the sediment retention and ecosystem provisioning services in the Kashmir valley, India, Western Himalayas. *Modeling Earth Syst. Environ.* **2021**, 1–26. [CrossRef]
- 42. Cullingworth, J.B. Essays on Housing Policy: The British Scene; Routledge: London, UK, 2021.

- Rahman, M.M.; Szabó, G. Multi-objective urban land use optimization using spatial data: A systematic review. Sustain. Cities Soc. 2021, 74, 103214. [CrossRef]
- 44. Tapping the Revenue Potential of Property Tax in India. Available online: https://www.ideasforindia.in/topics/governance/tapping-the-revenue-potential-of-property-tax-in-india.html#:~{}:text=Property%20tax%20system%20in%20India&text=However%2C%20the%20revenue%20collections%20from,to%204%25%20of%20their%20GDP (accessed on 31 December 2021).
- 45. Lucas, G. Remote Sensing and Image Interpretation, 3rd ed.; Lillesand, T.M., Kiefer, R.W., Eds.; Wiley: Chichester, UK, 1994; ISBN 0471-305-758.
- 46. Sharma, R.K. 8 BRICS in India's Vision for Global Governance. In *India and Global Governance: A Rising Power and Its Discontents;* Routledge: London, UK, 2022.
- 47. Jiang, S.; Li, Y.; Lu, Q.; Hong, Y.; Guan, D.; Xiong, Y.; Wang, S. Policy assessments for the carbon emission flows and sustainability of Bitcoin blockchain operation in China. *Nat. Commun.* **2021**, *12*, 1938. [CrossRef] [PubMed]
- Rocha, C.; Quandt, C.; Deschamps, F.; Philbin, S.; Cruzara, G. Collaborations for Digital Transformation: Case Studies of Industry 4.0 in Brazil. *IEEE Trans. Eng. Manag.* 2021, 1–15. [CrossRef]
- Stitzlein, C.; Fielke, S.; Waldner, F.; Sanderson, T. Reputational Risk Associated with Big Data Research and Development: An Interdisciplinary Perspective. Sustainability 2021, 13, 9280. [CrossRef]
- 50. Maessen, L. Tax Avoidance by Multinational Enterprises and the Current International Tax System: An Ethical Perspectivei. Master's Thesis, Raboud University, Nijmegen, The Netherlands, 2021.
- 51. Bera, A.; Taloor, A.K.; Meraj, G.; Kanga, S.; Singh, S.K.; Đurin, B.; Anand, S. Climate vulnerability and economic determinants: Linkages and risk reduction in Sagar Island, India; A geospatial approach. *Quat. Sci. Adv.* **2021**, *4*, 100038. [CrossRef]
- 52. Meraj, G.; Singh, S.K.; Kanga, S.; Islam, M. Modeling on comparison of ecosystem services concepts, tools, methods and their ecological-economic implications: A review. *Modeling Earth Syst. Environ.* **2022**, *8*, 15–34. [CrossRef]





Article Outline and Impact of Firms Related to the Third Industrial Revolution: Investigation through Big Data

Matthieu Belarouci

Anthropo-Lab—ETHICS EA7446, Université Catholique De Lille, 59800 Lille, France; matthieu.belarouci@univ-catholille.fr

Abstract: This article is focusing on the dynamism of the Third Industrial Revolution (TIR) in the region Hauts-de-France between 2013 and 2018. Region Hauts-de-France has been a forerunner in setting up a proactive policy in favour of the TIR. It provides a relevant and suitable context for the identification of TIR activities. We assess the job dynamism of the TIR through the implementation of big data methods for the identification of the firms involved in the TIR activities and the collection of firm microlevel data. We provide evidence of the strong dynamism of the TIR activities in a context of weak regional dynamics. We show that the growth in employment arises mainly from renewable energies, positive energy buildings, circular economy and energy efficiency. Future researches are encouraged to investigate the quality of employment and to question the distinctive characteristics of the firms involved in TIR.

Keywords: Third Industrial Revolution; sustainability; growth; jobs; circular economy; function-based product-services systems; renewable energies; positive energy buildings; energy efficiency; mobility

check for updates

Citation: Belarouci, M. Outline and Impact of Firms Related to the Third Industrial Revolution: Investigation through Big Data. *Smart Cities* **2022**, *5*, 522–538. https://doi.org/10.3390/ smartcities5020028

Received: 16 February 2022 Accepted: 1 April 2022 Published: 9 April 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1. Introduction

Employment growth, or more generally value creation, derived from activities related to the TIR is a sine qua non condition for its strong sustainability. Besides, sustained economic growth and the creation of decent jobs for all make one of the key Sustainable Development Goals (SDG) set out in the United Nations 2030 Agenda for Sustainable Development. In the short term, demonstrating that activities related to the TIR contribute to creating jobs will lead to societal acceptance of the destruction involved in substituting carbon-based technologies, and will foster the investment of innovative firms in these sectors [1]. It is considered by some authors as an effective strategy to revitalise the economy of developed countries and it materialises strong expectations from public policymakers [2-7]. Insofar as the shift toward TRI inevitably comes with job destructions as activities that emit carbon disappear, the question of whether the TIR can lead to net job creation is non-trivial [8]. Despite the importance of job creation related to the TIR, there is some debate about the assessment of its positive impact on net growth in employment. Ref. [8] (Table 1, p. 759) report that empirical studies are not unanimous. Although a majority of studies show positive effects [9–13], others show either a negative [14] or absence of [15] effect on growth in employment. However, recent empirical literature has highlighted more positive effects [2,8]. It should be noted that these studies concern the green economy, which scope goes beyond that of the TIR.

Surprisingly, France has very hardly been studied. Most fall under grey literature and originate from public organisations charged with the transition [16–19] (pp. 21–22 and pp. 81–82). Their results are oftentimes extremely optimistic and based on strong assumptions, such as that of full sectorial and geographical mobility of workers [4,20]. These assumptions, even though they hold little realism, are essential to overcome the difficulty of identifying green organisations and jobs, owing to the mismatch between existing sectoral benchmarks and sustainable activities definitions. The difficulty of identification results from the technophile definition of TIR activities, which does not easily adapt to existing

sectoral codifications [21]. Furthermore, these studies are very rarely specific to the TIR activities. Our methodological approach partially solves this issue in combining big data and legal expertise with a view to gradually identified the firms involved in the TIR activities. From the data collected with businesses, we estimate job creation in a similar fashion as INSEE (Institut National de la Statistique et des Etudes Economiques is the French national office of statistics.) and use the end-sizing method to simplify the comparison of our results with local and national censuses. As [3] advocate, we gradually refined the list of firms from a broad census to those with a core business related to the TIR. From the methodology employed in our research, a precise census of firms involved in the TIR has emerged. This facilitates the way their evolution is to be monitored, the way in which their transition is better supported, and the assessment of its economic performance. It is also still important to better estimate the number of green jobs on the territory to foster the implementation of efficient local strategies—for instance, to adapt regional policies and plans regarding skills [22] (p. 1056). The originality of our paper also lies in our focus on the dynamics of employment growth in firms based on their level of involvement in the TIR fields. To our knowledge, no previous research has led to the microeconomic evaluation of the positive impact of TIR activities on the evolution of job opportunities in France.

Our results confirm that firms focused on TIR saw their dynamics increase between 2013 and 2018 with an average employment growth rate of +17.38 percent in a context of weak regional dynamism (+1.34 percent). This trend is amplified for the firms strongly focused on transitioning with an increase of +21.09 percent. Our analyses are broken down by field of activity and show that the growth is mainly supported by the transition to renewable energies, the development of positive energy buildings and circular economy. Firms focused on the fields of production of energy storage systems, energy internet and function-based product-service systems are harder to identify either because they are at an embryonic stage, or because of the lack of information to qualify their business model (e.g., function-based product-service systems). Finally, the analysis of the situation in the five districts shows that the development of TIR in the Hauts-de-France is homogenous insomuch as no regional specificities can be noticed regarding the various fields. We present the literature review and the research context in Sections 2 and 3, respectively. In Section 4, we explain the methodological approach used to identify firms involved in the TIR activities. Section 5 shows and discusses the regional and national findings, as well as findings for each field. Section 6 discusses the contribution of our method but also its limitations and avenues for future research.

2. Literature Review

The fight against climate change and the transition to a sustainable economy require actions from regions and countries, and international cooperation [6,7,23]. In this perspective, the European Commission is implementing its Green Deal policy with the help of the public authorities and local private actors to come to carbon neutrality by 2050 and to the creation of sustainable economic growth and subsequent job creation. The interest of policy makers for the TIR is enhanced by the fact that the related jobs are expected to be higher skilled [24,25] and not easily relocated [26]. As the growth and composition of employment related to the TIR is intrinsically intertwined with the policy makers' investment, innovation and normalization strategies in green economy [2,8,27,28] probing how the TIR can be a driver of growth and job creation is thus crucial to provide the public policymaker with guidance for these actions.

Numerous studies have supported the idea of positive dynamics that the TIR—and green economy more largely—has on job opportunities in several political and technological contexts. Reference [12] show that the transition towards a zero-carbon energy production in China resulted in 472,000 net jobs gains. Reference [2] assert that the Scottish policy in favour of green economy led to an increase of more than 72,000 jobs between 2011 and 2014. Following [8,21] report that policies in favour of green economy boost innovation and employment in the EU, the US and Japan. In the same geographical area [29,30] the

development of wind power technology has fostered job creation. Conversely, older studies cautiously point to some neutral or negative effects [14,15,31,32].

Even though most of the recent contributions state that the TIR related activities are prone to growth in employment, it is of importance to note that differences in metrics, methods and contexts limit the generalization of the results. First, the estimation of the growth in employment related to the TIR depends on the scope: direct, indirect or induced employment. Direct employment refers to jobs observed in organization in targeted industries, indirect employment concerns subsequent jobs resulting from changing in the targeted industries/organisations and induced employment results from the jobs generated by the activity of both direct and indirect jobs [12]. Secondly, the range of TRI jobs may differ. As in green economy, some authors adopt a purist approach considering the jobs that restoring or preserving environment, while others refer to broader definitions including wider impacts [2,33]. Thirdly, the technological context of implementation of TIR activities and its interaction with regional/national policies is inherently linked with the growth of jobs [2,34,35]. Given the context, the proportion of employment related to conversion, to new or emerging fields or greening of existing skills may differ and may have different impacts on the indirect or induced jobs.

Regarding the specificity of the French technological context—characterized by small renewable but strong nuclear attachments resulting in low carbon emission in energy production [36,37] (France reaches the highest proportion of electricity production from nuclear sources in the world with 77.63%, while it is ranked 60th for the production from the renewable energies excluding hydroelectricity account for 6.2% (Source: WorldBank).)-France provides a relevant context for the analysis of the development of TIR activities. Whereas many contributions focus on the growth associated with the transition in the field of production of energy, the importance of nuclear power plants limit growth potential and implies to consider other aspects of the TIR. Furthermore, France invested of 3.5 € billions between 2013 and 2017 for the transition in the perspective of the creation of 470,000 jobs. Surprisingly, the French dynamic of the TIR has remained relatively unexplored. Most of the analysis of the dynamism of the French job market related to the TIR fall under grey literature and are released organisations in charge of the transition [16-19] (pp. 21–22 and pp. 81–82). Then, the results must be cautiously considered. As ([4], p. 24 in [2]) note Uncritical acceptance of optimistic estimates of green job growth is especially ill-advised, given that these are typically produced by organizations and groups with a "vested interest in the outcomes". Besides, results of these studies are generally extremely optimistic and based on strong assumptions, such as that of full sectorial and geographical mobility of workers [4,20]. The PPEC ("Plan de programmation des emplois et des compétences") report on the jobs and skills programming plan (PPEC, 2018 pp. 21-22 and pp.81-82) lists 22 studies published between 2010 and 2017. With the exception of the National Treasury (Direction Générale du Trésor) (2010), they all underline the jobs creations. This lack of research arises mainly from the difficulty of identifying firms involved in the TIR activities. Contrasting with Scotland or the US, France did not adopt O*NET classification of green occupations and as the [20] confirms: "it is proven impossible to identify jobs related to energy transition by mining directly from official classifications based on PCS (jobs and socio-professional categories) or ROME (Operational Directory of Occupations and Jobs), respectively used by INSEE and Pôle Emploi, the French employment agency. The classification of "Familles Professionnelles" (FAP) (professional "families" or groups), built by DARES to bridge the gap between PCS and ROME, is equally unhelpful". Hence, the investigation of the TIR dynamism in France requires to produce *ad-hoc* identification criteria. We contribute to the literature on the impact of TIR activities on economic development by both focusing on the specific context of Hauts-de-France and proposing a general method of identification of businesses related to TIR activities.

3. Context of the Research: The Voluntarist Policy of Hauts-De-France Region

We apply the method of identification of TIR-related businesses in Hauts-de-France, that is ranked 12th among 13 regions in terms of GDP per capital, but 5th in number of

inhabitants. Hauts-de-France is the second poorest French region and TIR has been involved in development of entrepreneurship and innovation programs to promote economic growth. Hauts-de-France has been a forerunner in setting up a proactive policy in favour of the TIR. The region, inspired by [38]'s vision, was committed to revitalisation to pursue its industry conversion by favouring "technological and social biodiversity". The objective of the TIR project, launched in 2013 and renamed rev3 in 2015, is to help the transition of economic actors through its support to public, private and citizens' initiatives. The various projects aim to promote connected cities and regions, smart transport, renewable energies, circular economy, the creation of social capital and education to the challenges this societal transformation. This policy translates in the support to and promotion of 5 strategic fields, called "pillars": transitioning to renewable energies, such as wind, solar, geothermal or hydro, developing positive energy buildings, producing energy storage systems, deploying an energy internet and sustainable mobility of people and goods. In addition to these fields, business models are being strengthened: circular economy, function-based product-service systems and energy efficiency [39]. This policy relies also on financial supports to the actor of the TIR and is the most intense in France. For instance, the Region has invested 650 million euros in TIR activities in 2021 to accelerate their development.

Focusing on this regional scale is also relevant with regards to the significance of the role of the territory in the development of sustainable economy [40–45], particularly regarding sustainable mobility of people and goods [46], function-based product-service systems [47] and circular economy [48]. Indeed, their developments rely on the cooperation between a sector or a market's stakeholders and on the intervention of a third party, particularly local public authorities [45].

4. Materials and Methods

The methodological approach relies on massive data collection in order to establish distinctive characteristics of the firms involved in the TIR. The data collection is then recursive and iterative. The data generating process encompasses 3 stages. The first stage aims at delimiting the set of firms involved in the TIR activities. We combine internet webscraping and text-mining on data bases of the Regional Council Hauts-de-France in order to capture these specific firms in a large and non-discriminatory sample. The second stage focuses on the collection of legal information from the firms related to the TIR. This information is the input of the third stage. This latter stage focuses on the estimation of the yearly employment of the firms selected to assess the regional trend of the TIR. The interest of this approach relies on the traceability of the firms, since a detailed list of business units results from the data collection. By contrast with macroeconomic and sectorial simulations, our approach is more prudential because it focuses only on the direct and observable employment. Indirect and induced employment are excluded.

4.1. Stage 1: Limiting the Set of TIR Activities and Identification of the Firms

Most of the data results from internet webscraping and text mining on Regional Council data bases. We also survey professional associations and business clusters to obtain firms lists for each TIR activity. The method of webscraping consists in massive data collection from the internet. Our purpose is to collect specific lists of firms involved in TIR activities in Hauts-de-France. We collect data from 53 sources ranked and categorized given their TIR activity and their relevance (cf. Appendix A-Table A1). The sources are selected with the support of the Mission rev3, the Regional Department of TIR, the Chamber of Commerce and Industry, Ademe, *Hauts-de-France Innovation Development* and the "*Chaire des Explorateurs de la Transition*". We surveyed 40,788 observations including 5546 strongly specialized in TIR activities.

Thanks to the support of the *Centre d'Etude d'Aide à la Décision* (CEAD) *de Hauts-de-France Innovation Développement* (HDFID), we perform identification of the firms based on text-mining on the firms' descriptions in the *Applications et Services de Travail en Réseau pour l'Innovation et le Développement Economique* (ASTRIDE) data base. We select the keywords

listed in Appendix A-Table A2 in collaboration with the stakeholders involved in the selection of the sources for the webscraping data collection, *Fonds Régional d'Amplification de la Troisième Révolution Industrielle* (FRATRI) 2020 and Ademe services. As with the webscraping approach, the interest of the text-mining is to identify and propose criteria for the classification of the firms in TIR activities. We have collected 8803 firms on ASTRIDE including 4723 strongly specialized in the TIR.

4.2. Stage 2: Collection of Legal Information

Legal information is required to obtain the number of employees and the financial statements of the firms in financial data bases Diane+ Bureau Van Dijk. At the end of the first stage, 10,159 firms (20 percent of the sample) are not documented. We perform automatic requests on [49–51]. Requests are based on the name of the firms and additional information such as city, phone number and national industry sector classification. At the end of this second stage, 5709 observations over the 49,591, that is to say 11.51 percent of the sample, remain without legal information and cannot be included in the third stage. Nonetheless, it is likely that many of them already taken into account since 34.51 percent of the observations have multiple occurrences. These multiple occurrences in the selected sources are a desirable outcome insofar as it informs on the exhaustiveness of the sample and the strength of the link of the firm with the TIR activities. Since Diane+ provides information on the parent business units of the firm, the number of observations increases to 152,351, but results are only analysed for the "mother house" for which the information is the most complete.

4.3. Stage 3: Data Collection and Estimation of Employment

Two kinds of data are downloaded from the financial databases Diane+ and Astrée provided by Bureau van Dijk: data on the number of employees and standardized accounting statements from 2013 to 2018. When the number of employees is not available, we refer to OLS estimates. We predict the number of employees of the firm with the following regressors: the (i) total salaries in euros and the related (ii) benefits paid by the firm. (iii) The total value of assets reflects the size of the firm likely to modify the distribution of the wages. The (iv) industry sector and (v) period fixed effects are captured by categorical variables. Estimations are performed over 28,309 observations for which the number of employees is documented. We infer the predicted number of employees with accounting data for the rest of the sample. Following INSEE analyses, we assess the average growth in employment relative to 2013 with the end-sizing approach after controlling for modification in the structure of the sample [52]. Results are classified given the TIR activities, the department and for the samples of large and specialized firms.

5. Results

5.1. Employment Growth Related to the TIR in Hauts-De-France

Figure 1 shows the evolution of employment growth in Hauts-de-France in index with fixed base 2013 adjusted to sample attrition. The green and blue lines respectively reflect the evolution of employment in firms with a core business related to the TIR and in our overall sample. These values should be seen in the context of the general employment situation in Hauts-de-France over the same period, shown in the orange dashed line. We find that firms involved in the TIR strongly favour employment dynamics; this result is even more acute for firms with a core business related to the TIR. It is noteworthy that the slowdown observed between 2013 and 2015 in the overall sample can be attributed to the decline in the building and construction industries. As we can observe, employment raised from 2016 partly due to the change in the sample composition. Despite this end-of-period overestimation inherent to this approach, it is worth reminding that our estimation of employment growth is prudential, as only direct job creation is included. Direct employment are excluded.



Figure 1. Evolution of employment growth related to TIR in Hauts-de-France between 2013 and 2018 (in base 100 in 2013).

Table 1 shows the evolution of employment growth in firms with a core business related to the TIR and in our overall sample in volume and percent variation compared to the regional situation. We can observe a creation of +26,783 jobs in Hauts-de-France between 2013 and 2018, i.e., a growth of 1.34 percent for all sectors combined. Over the same period, organisations related to the TIR seems to have provided for up to 52,173 new jobs, of which 27,838 were created by firms strongly focused on the TIR activities.

Firms with Core Business Related to the TIR		Overall	Sample	INSEE Hauts-De-France— All Sectors ¹		
Jobs	% Var	Jobs	% Var	Jobs	% Var	
+27,838	+21.09%	+52,173	+17.38%	+26,783	+1.34%	

Table 1. Estimation of direct employment growth from 2013 to 2018.

¹ Net of job destruction. Source: [53].

5.2. Dynamics by Pillar and across Districts

In Table 2, we present the distribution of firms by pillar and districts. The distribution of pillars is illustrated in Figure 2 and Table 3. The distribution of pillars across districts is homogenous. Data collected with our census is in line with the business demographics complied by the INSEE over the same period. Nord concentrates 42 percent of businesses, Pas-de-Calais 23.8 percent, Oise 13.4 percent, Somme 11 percent and Aisne 9.9 percent. The firms listed fall for the most part into the strategic fields of the transition to renewable energies and the development of positive energy buildings that respectively account for 17.9 percent and 29.36 percent of firms listed. Table 4 shows that, on average, Pillar 1 firms are active in 2825 pillars and that only 2.59 percent of them are exclusively focused on Pillar 1 activities. It is surprising that 69 percent of Pillar 1 firms are also active in Pillar 2 activities, which indicates that these technologies are interrelated. A third of these firms are also active in Pillars 6 and 8. Pillars 1 and 5 are also better represented in the sample of specialised firms than in the overall sample.

Districts	Pillar 1: Transition- ing to Renewable Energies: Wind, Solar, Geothermal, Hydro	Pillar 2: Developing Positive Energy Buildings	Pillar 3: Getting Equipped with Energy Storage Systems	Pillar 4: Deploying an Energy Internet	Pillar 5: Sustainable Mobility for People and Goods
Nord Baa da	17.99%	28.33%	2.71%	1.67%	5.14%
Calais	19.67%	29.54%	2.05%	0.93%	3.26%
Oise	16.62%	30.12%	0.53%	0.53%	1.71%
Aisne	16.19%	31.11%	0.79%	0.73%	2.02%
Somme	16.82%	30.26%	0.91%	0.96%	2.58%
Region	17.90%	29.36%	1.86%	1.16%	3.62%
Districts	Pillar 6: Circular Economy	Pillar 7: Function- based Product- Service Systems	Pillar 8: Energy Eficiency	Total— Authors' Calculation	Distribution of Firms— INSEE
Nord	25.22%	1.44%	17.51%	42%	45%
Pas-de- Calais	25.54%	0.83%	18.19%	23.8%	21.4%
Oise	28.53%	0.42%	21.53%	13.4%	14%
Aisne	27.24%	0.64%	21.28%	9.9%	9.1%
Somme	27.04%	0.75%	20.67%	11%	10.5%
Région	26.15%	1.00%	18.96%	100%	100%

Table 2. Distribution of firms by pillar and districts.



Figure 2. Distribution of activities and employment growth rates from 2013 to 2018.

	Pillar 1: Transi- tioning to Renew- able Energies: Wind, Solar, Geother- mal, Hydro	Pillar 2: Develop- ing Positive Energy Buildings	Pillar 3: Getting Equipped with Energy Storage Systems	Pillar 4: Deploy- ing an Energy Internet	Pillar 5: Sustain- able Mobility for People and Goods	Pillar 6: Circular Economy	Pillar 7: Function- Based Product- Service Systems	Pillar 8: Energy Eficiency	Average Annual Number of Firms 2013–2018
Pillar 1	100%	96%	9%	1%	3%	36%	1%	38%	3,601
Pillar 2		100%	5%	1%	1%	64%	1%	64%	4,735
Pillar 3			100%	16%	17%	36%	11%	25%	458
Pillar 4				100%	86%	31%	75%	38%	131
Pillar 5					100%	18%	16%	10%	702
Pillar 6						100%	1%	75%	4,217
Pillar 7							100%	33%	104
Pillar 8								100%	2,618

Table 3. Table of contingency of number of firms by pillar.

Table 4. Pillar distribution of direct job creation from 2013 to 2018.

Pillars	Firms with C Related t	Core Business o the TIR	Overall Sample		
_	Jobs	% Var.	Jobs	% Var.	
Pillar 1	+15,894	25.65%	+21,944	16.48%	
Pillar 2	+8357	14.78%	+10,362	7.29%	
Pillar 3	+2618	8.97%	+2556	8.66%	
Pillar 4	-243	-1.43%	+88	0.51%	
Pillar 5	+7167	8.55%	+7277	8.68%	
Pillar 6	+17,208	21.17%	+25,048	11.30%	
Pillar 7	-1376	-8.97%	-1073	-7.04%	
Pillar 8	+6179	16.39%	+1579	2.86%	
Total without duplicates	+27,838	+21.09%	+52,173	+17.38%	

The cross-sectoral fields that are defined as economic models in rev3 benchmark rate similarly well. Businesses involved in circular economy account for 26.15 percent of the companies listed and the field of energy efficiency accounts for 18.96 percent of the sample. The other fields are conversely largely under-represented, either because they are at an embryonic stage, as for the deployment of an energy internet, or because of the difficulty of identifying which businesses develop operations in some branches of activity, particularly function-based product-service systems.

5.2.1. Analysis of the Dynamics for Each Pillar

For each pillar, employment growth is derived in terms of volume over 2013 to 2018 and in terms of initial job quantity in 2013. Table 4 shows that employment dynamics differ from one pillar to the next. Activities related to transitioning to renewable energies (Pillar 1) and to developing positive energy buildings (Pillar 2) are largely represented in the sample. They generated a creation of +16.48 percent and +7.29 percent jobs respectively. Fields that are under-represented, and particularly "getting equipped with energy storage systems" (Pillar 3), are also growing strongly (+8.66 percent). Pillar 5 (sustainable mobility for people and goods) shows a similar growth rate but is abnormally under-represented with regards to the importance of the transportation industry in Hauts-de-France [54]. This under-representation results from the difficulty of identifying sustainable businesses in the transportation industry. Among the cross-sectoral pillars, circular economy (Pillar 6) and energy efficiency (Pillar 8) are particularly dynamic with growth rates of 11.30 percent and 2.86 percent observed in the overall sample. However, these dynamics are stronger for companies with a core business directly related to the TIR. In that regard, energy efficiency (Pillar 8) is marked by a growth rate of 16.38 percent. The trend showed for companies with a core business related to the TIR is twice as strong as for the overall sample in the fields linked to Pillars 2, 6 and 8.

5.2.2. Analysis of the Dynamics across Districts

In Table 5, we reported employment dynamics related to the TIR across districts. Regarding job creation, Nord and Pas-de-Calais display stronger dynamics than the South of the region, because of their initial job quantity in 2013 and of the scale of their growth rates.

Districts	Firms with C Related t	Core Business to the TIR	Overall Sample		
_	Jobs	% Var.	Jobs	% Var.	
Nord	40,196	24.57%	54,864	8.27%	
Pas-de-Calais	31,935	24.18%	45,637	11.01%	
Oise	7900	10.89%	-589	-0.25%	
Aisne	6542	13.95%	12,014	9.23%	
Somme	-	-	-	-	

Table 5. Direct job creation by district in Hauts-de-France from 2013 to 2018¹.

¹ Depending on the number and distribution of firms on the territory, the same business can be listed in several districts. It is impossible to derive regional employment from the sum of employment in each of the region's districts.

We can observe that information related to firms in Somme lacks sufficient data to provide reliable identification of job opportunities in this district. This lack results from voluntary failure from firms to disclose information. The ASTRIDE data base of the Regional Council Hauts-de-France, devoted only to firms located in the former Nord-Pas de Calais territory until 2018, also reports a small number of firms for this district.

6. Discussion

The purpose of this article is to report on job creation generated by the TIR in Hauts-de-France. Our methodological approach enables the listing of firms related to the TIR in view of their analysis. To our knowledge, no previous research has examined the employment dynamics linked to the TIR through microeconomic modelling. We estimate net first job creation linked to the TIR from 2013 to 2018 at +52,173 for a growth of +17.38 percent, while the region counted a mere +26,783 jobs created for a growth of +1.34 percent. The proportion of TIR-related employment in Hauts-de-France would amount to 6.7 percent, i.e., 159,000 jobs. We can observe that this positive trend is stronger for firms classified as "specialised firms", i.e., as having a core business related to the TIR, with a growth rate of +21.09 percent or 0.805 job created per business per year. The ANOVA test confirms the difference between the two groups (F-value = 2.295 **; p-value= 0.0076). This result confirms the link between the development of the TIR and job creation established by academic [8,12,21,29,30] and grey literature [16,17,55].

The study of the dynamics by pillar shows that this strong growth is essentially supported by the transition to renewable energies, the development of positive energy buildings, circular economy and energy efficiency. The dynamism of the field of renewable energy is materialized also by the change in energetic mix. The production of renewable energy was 21.2% in 2019 whereas it was 4.2% in 2013. It substituted partly nuclear production which drop from 80% to 60% in the same period (See [56]). This suggests that these job concerns the emergence of new skills [33]. In the field of circular economy, there is a particularly significant difference between specialized firms (+16.39 percent) and the

overall sample (+2.86 percent). Only firms focused on the fields of energy internet and function-based product-service systems register job destruction. However, these fields are ill-represented in our sample. Surprisingly, the field of sustainable mobility is also under-represented, as Hauts-de-France is considered the third leading region for logistics in France [54]. The main difficulty lies in the absence of criteria differentiating sustainable logistics companies from traditional ones. Examining the distinguishing features of firms involved in sustainable mobility is required to better target censuses, as, following [57], representations of sustainability remain absent or fragmentary among the actors of the sector, all the more so as they cover different realities, depending on the nature of goods and actors concerned [40].

Results also reveal the importance of the local context [40,41,58]. This growth in employment is intrinsically related to the voluntarism of the regional authorities in supporting TIR. Remaining that the Hauts-de-France owns the second biggest nuclear power plant in Europe, the development of renewable energies especially wind energy has only been possible by the public investment and active cooperation with the firm in charge of the production and distribution of electricity. Another evidence of the importance of the local context is observed in the territorial differences. The dynamics by district shows that job creation is stronger in Nord and Pas-de-Calais, which constitute the historical region until 2015. They have benefited to the TIR policy in the early 2010, while the other districts Somme, Oise and Aisne were involved during 2015. These latter districts less exposed to TIR policy exhibit lower rate of growth. The dynamics is uneven in Oise: specialised firms increased employment by +10.89 percent while the overall sample shows sluggish growth (-0.25 percent). We advocate for more contextual analysis of the growth triggered by the TIR. Political and technological contexts set the economic growth potential of the TIR and the sources and the composition of new jobs (conversion, entirely new skills or expansion of existing jobs).

This research provides also a general method of identification of firms related to TIR activities. As many countries, France do not provide green or transition related classification of jobs contrasting with US or Scotland [2]. The bigdata based approach permit to both identify a large number of organizations involved in the TIR, but also to focus on specific subset of more dynamic business unit. It fosters the analysis of alternative metrics of the development of the TIR as well as the growth in fundraising, economic value added or turnover. It is also appropriate to understand the determinants of the success of these organization. Nonetheless, at this stage the method cannot capture precisely the nature and quality of the jobs without combining complementary data collection.

If employment growth is a sign of economic performance, it cannot account for the employment dynamics of the TIR on its own. Further research to investigate the quality of employment [52,59] is needed in order to foster a fair shift to a greener economy [60]. Some questions, such as "Are these jobs new occupations, changing occupations or occupations developing from pre-existing jobs already linked to the TIR? Is the dynamics inclusive?" require consideration [2,33,41] to promote understanding and support of the TIR activities. Although there is no unanimously accepted definition of green jobs or TIR today, the identification of skills required by the firms and the quality of jobs created is a question to address to better direct support and to improve the characterisation of the dynamics [33]. A more detailed description of the employment dynamics related to the TIR would require a comparison of the dynamics of shifting actors to that of their competitors who are not involved in sustainable activities. The outcomes of this detailed comparison would contribute to singling out the dynamics specifically related to the TIR in Hauts-de-France, to understanding which business models are attached to these organisations, and to providing local public authorities with decision-making tools. There is a need to mobilize multiple research designs, in particular those based on the collection of primary data, in order to reference this dimension of TIR-related employment. In addition, job creation cannot account for all economic dimensions of the TIR. Other dimensions would require further analysis to measure the success of these firms. Their survival rate, their profitability, the

amount of funds raised, or their innovation potential would provide a more acute view of the economic development of the TIR [44].

7. Conclusions

The purpose of this research is to report on job creation related to the TIR in Hautsde-France from 2013 to 2018. The proactive policy of Hauts-de-France Regional Council in favour of the TIR since 2013 has fostered a unique context where networks and organisations in charge of the shift are prominent, facilitating the identification of the firms focused on the TIR activities. The originality of our approach is its use of the firms' individual data to facilitate monitoring of the dynamics over time and help targeted interventions from researchers wishing to further explore the profile of these organisations as well as from public authorities involved in supporting and guiding firms in shifting.

We collected a sample of 49,591 entities with their headquarters in Hauts-de-France and established distinctive characteristics of the firms involved in the TIR to create two groups: a large sample of 26,771 firms and a sub-sample of 6403 firms with a core business related to the TIR. Our results show the strong and positive employment dynamics in firms focused on the TIR with an average employment growth rate of +17.38 percent in a context of weak regional growth (+1.34 percent). This trend is amplified for the specialised firms with an increase of +21.09 percent. This growth in employment arises mainly from the fields of the production of renewable energies, positive energy buildings, circular economy and energy efficiency. These results support recent literature on the ability of the TIR at generating economic growth and confirm the importance of the political context in its development even we do not find differences in the composition TIR-related activities among districts.

While it is impossible to study the dimensions of spillover of the TIR entirely at this stage, our research forms a basis for broader and more critical research, particularly regarding the quality of employment created [2,33,59], in oder to foster a fair shift to a greener economy [59,60]. Creating new criteria of recognition of firms focused on the TIR and accounting for other dimensions than created value, survival of firms and their degree of innovation in comparison to firms in more traditional sectors are but a few possible avenues for future research. Theses studies would contribute to better understand the factors fostering economic success of firms focused on the TIR in view of providing better support to their development without supplanting the aim of sustainability.

Funding: This research was funded by the Regional Council Hauts-de-France and the Mission rev3 of the Chamber of Commerce and Industry of Lille (France).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: We gratefully thank the Mission rev3, the Chamber of Commerce and Industry Hauts-de-France, the Regional Council of Hauts-de-France TIR Department, Ademe, the cluster FEDEREC, the cluster Polénergie, the cluster FIBOIS, C2DE, CERD, and HDFID for their advice, administrative and technical supports.

Conflicts of Interest: The funder has been involved in the collection of data only.

Appendix A

 Table A1. Breakdown of sources by Pillar.

Pillar 1: Transition- ing to Renewable Energies: Wind, Solar, Geothermal, Hydro	Pillar 2: Developing Positive Energy Buildings	Pillar 3: Getting Equipped with Energy Storage Systems	Pillar 4: Deploying an Energy Internet	Pillar 5: Sustainable Mobility for People and Goods	Pillar 6: Circular Economy	Pillar 7: Function- Based Product- Service Systems	Pillar 8: Energy Eficiency
Ademe Club International	Ademe Club International	Ademe Club International	Ademe Club International	Ademe Club International	Ademe Club International	Ademe Club International	Ademe Club International
Ademe- Projets Verts	Ademe- Projets Verts	Ademe- Projets Verts	Ademe- Projets Verts	Ademe- Projets Verts	Ademe- Projets Verts	Ademe- Projets Verts	Ademe- Projets Verts
					Ademe- Repair Café	Ademe- Repair Café	
Allice Alliance	Allice Alliance	Allice Alliance	Allice Alliance		Aquavaley	Club Noé	Allice Alliance
Annuaire Des Diagnos- tiqueurs Immobiliers	Cd2e	Aphypac	Cncres	Cncres	Eco- Circulaire.Org– Organisations Adhérentes	Cncres	Aquavaley
Aphypac	Enedis	Biogaz Valley	Enedis projets	Pexe	Fibois	Pexe	Biogaz Valley
Aquavaley	Fibois	Cd2e	Gimelec	Projets Rev3	Hydreos	Projets Rev3	Enedis
Biogaz Valley	Afcobois	Enedis projets	Greentech	Reseau Scic	Pexe	Reseau Scic	Fibois
Cd2e	Fédération Nationale Du Bois	Gimelec	Pexe		Projets Rev3		Gimelec
Cerrd	Greentech	Greentech	Pôle Energie Hdf		Rep-Gf— Opérateur	Liste Des s Attestés	Greentech
Enedis	Pexe	Pexe	Pole Medee		Reseau Scic		Pôle Energie Hdf
Fimea	Pôle Energie Hdf	Pôle Energie Hdf	Projets Rev3		Sinoe		Pole Medee
Fédération Nationale Du Bois	Projets Rev3	Pole Medee	Reseau Scic		Syderep— Dbps		Projets Rev3
Gimelec	Qualibat	Projets Rev3	Pôlenergie- Adherents		Syderep— Dea		Qualibat
Greentech	Rge	Rs2e			Syderep— Emb		Unités De Méthanisation
Pexe	Reseau Scic	Reseau Scic			Syderep Et Accum	—Piles nulateurs	Atee Club Biogaz
Pole Dream	Team2	Ufe			Syderep—Pu		Energie2020
Pôle Energie Hdf	Cd2e Acteurs Eco- Construction	Unités De Méthanisation			Syderep— Rep-Deee		Pôlenergie- Methanisation

Smart Cities 2022, 5

Table A1. Cont.

Pillar 1: Transition- ing to Renewable Energies: Wind, Solar, Geothermal, Hydro	Pillar 2: Developing Positive Energy Buildings	Pillar 3: Getting Equipped with Energy Storage Systems	Pillar 4: Deploying an Energy Internet	Pillar 5: Sustainable Mobility for People and Goods	Pillar 6: Circular Economy	Pillar 7: Function- Based Product- Service Systems	Pillar 8: Energy Eficiency
Pole Medee	Cd2e Entreprises	Cd2e A Eco-Cons	cteurs truction		Syderep— Vhu		Pôlenergie- Adherents
Projets Rev3	Cd2e Laboratoires	Cd2e Entreprises			Team2		Pefc
Rge	Pôlenergie- Adherents	Cd2e Laboratoires			Ufe		
Rs2e	Pefc	Energie2020			Upds		
Reseau Scic		Pôlenergie- Methanisation			Atee Club Biogaz		
Syndicat Renour	t Energie velable	Pôlenergie- Adherents			Pefc		
Transition Territoire	Ecologique s Emplois						
Ufe							
Unités De Méthanisation							
Atee Club Biogaz							
Cd2e A Eco-Con	Acteurs struction						
Cd2e Entreprises							
Cd2e Laboratoires							
Energie2020							
Pôlenergie-N	lethanisation						
Pôlenergie- Adherents							

 Table A2. Key words used in the search on ASTRIDE—Hauts-de-France Innovation Développement.

Pillar 1: Transition- ing to Renewable Energies: Wind, Solar, Geothermal, Hydro	Pillar 2: Developing Positive Energy Buildings	Pillar 3: Getting Equipped with Energy Storage Systems	Pillar 4: Deploying an Energy Internet	Pillar 5: Sustainable Mobility for People and Goods	Pillar 6: Circular Economy	Pillar 7: Function- Based Product- Service Systems	Pillar 8: Energy Eficiency
Renewal energies	Positive energy buildings	Energy storage	Energy Internet	Mobility of people and goods	Circular economy	Product- service system	Energy efficiency

534

Smart Cities 2022, 5

		Table A2. Cont.					
Pillar 1: Transition- ing to Renewable Energies: Wind, Solar, Geothermal, Hydro	Pillar 2: Developing Positive Energy Buildings	Pillar 3: Getting Equipped with Energy Storage Systems	Pillar 4: Deploying an Energy Internet	Pillar 5: Sustainable Mobility for People and Goods	Pillar 6: Circular Economy	Pillar 7: Function- Based Product- Service Systems	Pillar 8: Energy Eficiency
Energy efficiency	Energy consumption	excessive spare capacity	SMART NETWORK	Multimodal transport	"zero emissions"	Energy transition	Direct energy efficiency
Hydroelectricity	Thermal renovation	Production irregularity	Smart grids	"zero emissions"	"Adaptating to climate change"	Service Economy	Indirect energy efficiency
Energy system	Energy and enivronmen- tal quality	"zero emissions"	Smart cities	"Adaptating to climate change"	"Alleviatir chan	ng climate ge″	Ecodesign
Decarbonated	Passive house	"Adaptating to climate change"	Smart grid	"Alleviating climate change"	recycling		Rebound effect
Wind energy	Energy producing buildings	"Alleviating climate change"	Smart city	Carpooling	Ecodesign		Heating
Photovoltaic	Own energy consumption	Energy transition	Load management	Swift mobility	Energy transition		Lighting
Geothermal	Revegetation	Energy control	"zero emissions"	Intermodality	Waste		Compressed air
Hydraulic	Brownfield redevelop- ment	Green energies	"Adaptating to climate change"	E-mobility	Sustainable consumption		Energy transition
Renewable marine energy	Innovati manag	ve waste gement	"Atténuation du changement climatique"	"Alleviating climate change"	Sustainable production		Green activities
Biomass	Ecodistrict		Energy transition	Bike	Salvage		
Methanation	Bio-based materials		Green activities	Electric vehicles	Green activities		
Geothermal energy	EV charging i	infrastructure		EV charging infrastruc- ture	Rejects		
Hydrogen	EVSE			EVSE	Sustainable supply		
Energy conservancy	Hydraulic storage			Connector terminals	life cycle assessment		
Green energies "zero	Compressed air storage méthanation			Energy transition Air pollution	LCA Bio-based		
"Adaptating to climate change"	"zero emissions"			Noise	Industrial ecology		

Smart Cities 2022, 5

Pillar 1: Transition- ing to Renewable Energies Wind, Solar, Ceothermal, Hydro Pillar 3: Equipped with Energy Storage Systems Pillar 4: Deploying with Energy Storage Systems Pillar 4: Deploying Mobility for People and Green activities Pillar 6: Economy Pillar 5: Based Economy "Alleviating Cimate change" "Adaptating to climate change" Green activities Bioeconomy Service Systems Fillar 7: Product- Service Service Service Service Service Service Service Service Service Service Service Service Fillar 6: Based Circular Mobility for Alleviating climate change" Bioeconomy "Alleviating Cimate Change" Bioeconomy Image People Service S								
"Alleviating climate change" Green activities Bioeconomy Heating "Alleviating climate change" biocarburants biocarburants biocarburants Bioeconomy Heating "Alleviating climate change" biocarburants Bioeconomy Lighting Heating "Alleviating climate change" biocarburants Lighting Heating biocarburants biocarburants Lighting Lighting Mobility management mobility management Energy transition air Energy compressed Energy control Energy Solar Green activities Energy control Energy Waste heat Saving retrofit Energy Energy Heat new production green activities Saving retrofit Energy Heat server green gre	Pillar 1: Transition- ing to Renewable Energies: Wind, Solar, Geothermal, Hydro	Pillar 2: Developing Positive Energy Buildings	Pillar 3: Getting Equipped with Energy Storage Systems	Pillar 4: Deploying an Energy Internet	Pillar 5: Sustainable Mobility for People and Goods	Pillar 6: Circular Economy	Pillar 7: Function- Based Product- Service Systems	Pillar 8: Energy Eficiency
Heating Heating "Alleviating climate change" biocarburants biocarburants Lighting Heating Compressed air Lighting Energy transition Compressed air Green activities Energy transition Heat pump Control Solar Green activities Energy saving retrofit Energy saving retrofit Heat recycling Wood Firedamp Wood	"Alleviating climate change"	"Adapt climate	tating to change″		Green activities	Bioeconomy		
Lighting Heating bioéthanol Compressed air Lighting Mobility management Energy transition air Green Energy control Heat pump Energy control Solar Green activities Biogas Energy saving retrofit Firedamp Wood Steel production gas Lighting	Heating Heating	"Alleviating c "Alleviating c	limate change" limate change"		biocarburants biocarburants			
Compressed air Lighting Mobility management Energy transition air Green activities transition Heat pump Energy control Solar Green activities Waste heat Energy saving retrofit Heat recycling Wood Biogas Vood Steel production gas Lighting	Lighting	Heating			bioéthanol			
Energy Compressed transition air Green Energy activities transition Heat pump Energy control Solar Green activities Waste heat Energy saving retrofit Heat recycling Wood Firedamp Vood Steel production gas Law carbon	Compressed air	Lighting			Mobility m	anagement		
Heat pump Energy control Solar Green activities Waste heat Energy saving retrofit Heat recycling Wood Biogas Vood Firedamp Vood Steel production gas Low carbon	Energy transition Green activities	Compressed air Energy transition						
Solar Green activities Waste heat Energy saving retrofit Heat recycling Wood Biogas	Heat pump	Energy control						
Waste heat Energy saving retrofit Heat recycling Wood Biogas - Firedamp - Wood - Steel production gas - Low carbon -	Solar	Green activities						
Heat recycling Wood Biogas	Waste heat	Energy saving retrofit						
Biogas Firedamp Wood Steel production gas Low carbon	Heat recycling	Wood						
Firedamp Wood Steel production gas	Biogas							
Wood Steel production gas	Firedamp							
Steel production gas	Wood							
Low carbon	Steel production gas							
	Low carbon							

Table A2. Cont.

References

- 1. Zaoual, A.-R. Pour une contribution du management stratégique à l'écologie industrielle. *Développement Durable Territ*. **2014**, 5. [CrossRef]
- 2. Sofroniou, N.; Anderson, P. The green factor: Unpacking green job growth. Int. Labour Rev. 2021, 160, 21–41. [CrossRef]
- 3. Poschen, P.; Renner, M. Green Jobs: Protecting the Environment Can Go Hand in Hand with Economic Prosperity and Job Opportunities. *Financ. Dev.* 2015, 52, 14–17.
- Morriss, A.P.; Bogart, W.T.; Dorchak, A.; Meiners, R.E. *Green Jobs Myths*; University of Illinois Law and Economics Research Paper Series; n° LE09-001; University of Illinois College of Law: Champaign, IL, USA, 2009.
- 5. Tantau, A.; Santa, A.-M.I. New Energy Policy Directions in the European Union Developing the Concept of Smart Cities. *Smart Cities* 2021, 4, 241–252. [CrossRef]
- 6. International Labour Organization. *World Employment and Social Outlook 2015: The Changing Nature of Jobs;* ILO Research Department: Geneva, Switzerland, 2015.
- European Commission. Communication De La Commission Au Parlement Européen, Au Conseil, Au Comité Economique Et Social Européen Et Au Comité Des Régions Responsabilité Sociale Des Entreprises; Une Nouvelle Stratégie De L'UE Pour La Période 2011–2014; European Commission: Brussels, Belgium, 2011.

- 8. Aldieri, L.; Carlucci, F.; Cirà, A.; Ioppolo, G.; Vinci, C.P. Is green innovation an opportunity or a threat to employment? An empirical analysis of three main industrialized areas: The USA, Japan and Europe. J. Clean. Prod. 2019, 214, 758–766. [CrossRef]
- 9. Gagliardi, L.; Marin, G.; Miriello, C. The greener the better? Job creation effects of environmentally-friendly technological change. *Ind. Corp. Chang.* 2016, 25, 779–807. [CrossRef]
- 10. Horbach, J.; Rennings, K. Environmental innovation and employment dynamics in different technology fields—An analysis based on the German Community Innovation Survey. J. Clean. Prod. 2013, 57, 158–165. [CrossRef]
- 11. Licht, G.; Peters, B. The Impact of Green Innovation on Employment Growth in Europe; European Commision: Brussels, Belgium, 2013.
- 12. Cai, W.; Wang, C.; Chen, J.; Wang, S. Green economy and green jobs: Myth or reality? The case of China's power generation sector. *Energy* **2011**, *36*, 5994–6003. [CrossRef]
- 13. Pfeiffer, F.; Rennings, K. Employment Impacts of Cleaner Production—Evidence from a German Study Using Case Studies and Surveys. *Bus. Strategy Environ.* 2001, *10*, 161–175. [CrossRef]
- 14. Cainelli, G.; Mazzanti, M.; Zoboli, R. Environmentally oriented innovative strategies and firm performance in services. Microevidence from Italy. *Int. Rev. Appl. Econonm.* **2011**, 25, 61–85. [CrossRef]
- 15. Rennings, K.; Zwick, T. Employment Impact Of Cleaner Production on The Firm Level: Empirical Evidence From A Survey In Five European Countries. *Int. J. Innov. Manag.* **2003**, *6*, 319–342. [CrossRef]
- 16. ADEME. Ministère de la Transition Ecologique et Solidaire, rev3. In *Enjeux Energétiques et Emplois en Hauts-de-France: Scénarios Pour la Troisième Révolution Industrielle/rev3;* ADEME: Angers, France, 2018.
- ADEME. Économie Circulaire et Emplois en Hauts-de-France, Synthèse de L'étude Prospective; Réalisée Pour l'ADEME par le Groupement I-Care; Amnyos, Technopolis: Paris, France, 2020.
- Observatoire National Des Emplois Et Métiers De L'économie Verte. Les Eco-Activités et L'emploi Environnemental en 2018; Nauroy, F., Ed.; Commissariat Général au Développement Durable, Sous-Direction de L'information Environnementale: Paris, France, 2020.
- 19. Commissariat Général au Développement Durable. *Les Eco-Activités et L'emploi Environnemental en 2017: Premiers Résultats;* Nauroy, F., Ed.; Commissariat Général au Développement Durable: Paris, France, 2019.
- 20. Ministère du Travail; de L'emploi et de L'insertion. *Plan de Programmation des Emplois et des Compétences*; Ministère du Travail: Paris, France, 2019.
- 21. Shapira, P.; Gok, A.; Sensier, M. Probing "green" industry enterprises in the UK: A new identification approach. *Technol. Forecast. Soc. Chang.* **2014**, *85*, 93–104. [CrossRef]
- Consoli, D.; Marin, G.; Marzucchi, A.; Vona, F. Do green jobs differ from non-green jobs in terms of skills and human capital? *Res. Policy* 2016, 45, 1046–1060. [CrossRef]
- 23. United Nations Environment Programme. Green Jobs: Towards decent work in a sustainable, low-carbon world. In *Report Is Commissioned and Funded by UNEP, ILO, IOE and ITUC;* United Nations Environment Programme: Nairobi, Kenya, 2008.
- 24. Muro, M.; Rothwell, J.; Saha, D. Sizing the Clean Economy: A National and Regional Green Jobs Assessment, Metropolitan Policy Program, 1775 Massachusetts Avenue, NW Washington, DC United States 20036-2188; UNEP: Nairobi, Kenya, 2011.
- Acemoglu, D.; Autor, D. Skills, Tasks and Technologies: Implications for Employment and Earnings. In Handbook of Labor Economics; Elsevier: Amsterdam, The Netherlands, 2011; Volume 4, pp. 1043–1171.
- 26. Bowen, A.; Duffy, C.; Fankhauser, S. 'Green Growth' and the New Industrial Revolution. Grantham Research Institute on Climate Change and the Environment at London School of Economics and Political Science. Policy Brief January; Grantham Research Institute on Climate Change and the Environment and Global Green Growth Institute: London, UK, 2016.
- 27. AIEA. Impact on Employment and Activity-Industry Approaches of the Update Scenarios of the Third Industrial Revolution/Rev3 Master Plan in Hauts-de-France Region: Final Report; 50 Refs.; Available from the INIS Liaison Officer for France, Report Number: INIS-FR-18-1340; INIS Liaison Officer for France: Paris, France, 2018; 129p.
- Bowen, A.; Kuralbayeva, K.; Tipoe, E.L. Characterising green employment: The impacts of 'greening' on workforce composition. Energy Econ. 2018, 72, 263–275. [CrossRef]
- Aldieri, L.; Garofalo, A.; Vinci, C.P. R&D Spillovers and Employment: A Micro-Econometric Analysis; MPRA Paper 67269; University Library of Munich: Munich, Germany, 2015.
- Aldieri, L.; Grafström, J.; Vinci, C.P. Job Creation in the Wind Power Sector through Marshallian and Jacobian Knowledge Spillovers; Ratio Working Papers 340; The Ratio Institute: Stockholm, Sweden, 2020.
- Böhringer, C.; Keller, A.; van der Werf, E. Are Green Hopes too Rosy? *Employ. Welf. Impacts Renew. Energy Promot. Energy Econ.* 2013, 36, 277–285.
- Ge, Y.; Zhi, Q. Literature Review: The Green Economy, Clean Energy Policy and Employment. *Energy Procedia* 2016, 88, 257–264. [CrossRef]
- Dierdorff, E.C.; Norton, J.J.; Drewes, D.W.; Kroustalis, C.M.; Rivkin, D.; Lewis, P. Greening of the World of Work: Implications for ONET-SOC and New and Emerging Occupations; Report Prepared for U.S. Department of Labor, Employment and Training Administration; Office of Workforce Investment, Division of Workforce System Support: Washington, DC, USA, 2009.
- Marin, G.; Vona, F. Climate policies and skill-biased employment dynamics: Evidence from EU countries. J. Environ. Econ. Manag. 2019, 98, 102253. [CrossRef]
- 35. Rodrik, D. Green industrial policy. Oxf. Rev. Econ. Policy 2014, 30, 469–491. [CrossRef]

- Sovacool, K.B.; Schmid, O.; Stirling, A.; Walter, G.; MacKerron, G. Differences in carbon emissions reduction between countries pursuing renewable electricity versus nuclear power. *Nat. Energy* 2020, *5*, 928–935. [CrossRef]
- 37. Poumadère, M.; Bertoldo, R.; Samadi, J. Public perceptions and governance of controversial technologies to tackle climate change: Nuclear power, carbon capture and storage, wind, and geoengineering. *WIREs Clim. Chang.* **2011**, *2*, 712–727. [CrossRef]
- 38. Rifkin, J. *The Third Industrial Revolution: How Lateral Power Is Transforming Energy, the Economy, and the World;* Palgrave Macmillan: London, UK, 2011.
- 39. CCI Hauts-de-France; Conseil Régional Hauts-de-France. *Mission Troisième Révolution Industrielle*; Référentiel de la Troisième Révolution Industrielle en Hauts-de-France: Paris, France, 2017.
- Maillefert, M.; Robert, I. Écologie industrielle, économie de la fonctionnalité, entreprises et territoires: Vers de nouveaux modèles productifs et organisationnels? Développement Durable Territ. 2020, 11. [CrossRef]
- 41. Barbieri, N.; Consoli, D. Regional diversification and green employment in US metropolitan areas. *Res. Policy* **2019**, *48*, 693–705. [CrossRef]
- 42. Bouleau, G.; Carter, C.; Sergent, A.; Fournis, Y. Quels territoires pertinents pour écologiser les industries qui misent sur le renouvelable? *Développement Durable Territ.* 2020, 11. [CrossRef]
- Battaglia, M.; Cerrini, E.; Annesi, N. Can environmental agreements represent an opportunity for green jobs? Evidence from two Italian experiences. J. Clean. Prod. 2018, 175, 257–266. [CrossRef]
- 44. Li, W.; Jue, W.; Rongxiao, C.; Yongqin, X.; Shi, Q.L.; Feimei, W.; Mahmoud, M.; Xueping, W. Innovation-driven industrial green development: The moderating role of regional factors. *J. Clean. Prod.* **2019**, *222*, 344–354. [CrossRef]
- 45. Nahrath, S.; Gerber, J.-D. Pour une approche ressourcielle du développement durable. *Développement Durable Territ.* 2014, 5. [CrossRef]
- 46. Raimbert, C.; Raton, G. Collectifs logistiques et territoires dans les circuits courts alimentaires de proximité: La robustesse de la coopération analysée au prisme des communs. *Développement Durable Territ.* **2021**, 12. [CrossRef]
- 47. Roman, P.; Muylaert, C.; Ruwet, C.; Thiry, G.; Maréchal, K. Intégrer la territorialité pour une économie de la fonctionnalité plus soutenable. *Développement Durable Territ*. **2020**, *11*. [CrossRef]
- Doré, G. Économie circulaire et écologie industrielle. Approche empirique à partir d'expériences de clusters et de territoires. Développement Durable Territ. 2021, 12. [CrossRef]
- 49. Bulletin Officiel des Annonces Civiles et Commerciales (Bodacc). Available online: https://www.bodacc.fr/ (accessed on 8 February 2022).
- 50. Société.com. Available online: https://www.societe.com/ (accessed on 8 February 2022).
- 51. Infogreffe—Registre du Commerce et des Sociétés. Available online: https://www.infogreffe.fr/ (accessed on 8 February 2022).
- Bacheré, H.; Mirouse, B. Décomposition de L'évolution de L'emploi par Catégorie D'entreprise: Comparaison des Méthodes et Application à L'économie Française; Document de Travail Insee, H2021-02; Institut National de la Statistique et des Etudes Economiques: Paris, France, 2021.
- Institut National de la Statistique et des Etudes Economiques (INSEE). Available online: https://www.insee.fr/fr/statistiques/ serie/010567992 (accessed on 8 February 2022).
- 54. Lejeune, F.; Terra, S. La troisième région logistique française. Atouts et défis des Hauts-de-France; Insee: Paris, France, 2017.
- ADEME. Marchés et Emplois Dans le Domaine des Energies Renouvelables, Situation 2013–2015; ADEME: Angers, France, 2017.
 Regional Climate Obervatoire. Available online: http://www.observatoireclimat-hautsdefrance.org/Les-grandes-questions/
- Quelle-est-la-production-d-energie-renouvelable-en-Nord-Pas-de-Calais (accessed on 8 February 2022).
- 57. Schiebel, J. Quelle place pour la durabilité dans les représentations sociales de la mobilité ? Vers une meilleure compréhension des actions mises en œuvre. *Développement Durable Territ.* **2018**, *9*. [CrossRef]
- 58. Gonçalves, A. Les organisations logistiques à promouvoir dans le cadre des Politiques de Transport Durable de Marchandises: La question des circuits de proximité. *Développement Durable Territ.* **2013**, *4*. [CrossRef]
- 59. Masterman-Smith, H. Green Collaring a Capital Crisis? Labour Ind. 2010, 20, 317–330. [CrossRef]
- 60. Pettinger, L. Green Collar Work: Conceptualizing and Exploring an Emerging Field of Work. *Sociol. Compass* **2017**, *11*, e12443. [CrossRef]